

# Australian Government Department of Defence Defence Science and Technology Organisation Army Sustainability Modelling Analysis and Reporting Tool Phase 1: User Manual and Results Interpretation Guide

Matthew K. Richmond, John Stewien, Michael Zucchi, Andrew Roth, Shane Miller and Soi Do

Land Operations Division Defence Science and Technology Organisation

DSTO-GD-0595

## ABSTRACT

This report is designed to assist users of the Army Sustainability Modelling Analysis and Reporting Tool (A-SMART) in setting up input parameters and scenarios, running the models and interpreting the model outputs. A-SMART is a software tool under contracted development, sponsored by Army Headquarters and managed by the DSTO, which has reached the end of its first phase of development. A-SMART is a strategic-level force structure liability analysis tool, designed to forecast the dynamics of personnel and equipment populations over time for a particular scenario; it is intended that it will support analysis of the sustainability of planned Army force structures against a range of possible scenarios, as well as the force options testing process. A-SMART Phase 1 has been limited to the development of personnel, major equipment and supplies/strategic lift modules; it is envisioned that a second phase of development would extend the capability to the other Fundamental Inputs to Capability including collective training and facilities, and also include a costing module.

#### **RELEASE LIMITATION**

Approved for public release

Published by

Land Operations Division DSTO Defence Science and Technology Organisation PO Box 1500 Edinburgh South Australia 5111 Australia

*Telephone: (08) 8259 5555 Fax: (08) 8259 6567* 

© Commonwealth of Australia 2009 AR-014-654 November 2009

### APPROVED FOR PUBLIC RELEASE

# Army Sustainability Modelling Analysis and Reporting Tool Phase 1: User Manual and Results Interpretation Guide

# **Executive Summary**

This report constitutes a guide for users of the Army Sustainability Modelling Analysis and Reporting Tool (A-SMART) Phase 1 in setting up and running scenarios and to assist in the interpretation of model results. This manual leads the user through the set-up procedure and documents, creating/loading experiments and input data, viewing/editing input data, setting up operational scenarios, running the models, and viewing/interpreting results. Other reports are under development that will provide a description of the algorithms and architecture that underpin the models, in addition to some historical studies of the Australian Army as a part of the model verification.

A-SMART is a strategic-level force structure liability analysis tool, designed to forecast the dynamics of personnel and equipment populations over time for a particular scenario (a scenario is defined here as a plan of the operational commitments for the Army over a certain time horizon including the allocation of units to task groups and operational phases); it is intended that it will support analysis of the sustainability of planned Army force structures against a range of possible scenarios, as well as the force options testing process. A-SMART Phase 1 has been limited to the development of personnel, major equipment and supplies/strategic lift modules; it is envisioned that a second phase of development would extend the capability to other Fundamental Inputs to Capability (FIC) including collective training and facilities, and would also include a costing module. In its mature form A-SMART would support sustainability and costing analysis, across all FIC, of Army migration (including force expansion and changes to readiness posture) and allow testing of current/proposed force structures against a range of operational scenarios and policy initiatives.

Inputs into A-SMART include: unit establishment data for personnel and equipment (which provides the target populations), recruitment/procurement rates, separation/loss rates, career profiles (including rank progression and course details), maintenance frequency and periods, and current asset levels. The primary outputs from A-SMART are forecast population levels which are presented with the target levels; any gaps represents a risk to the sustainability of the force for the particular scenario.

# Contents

1.	INT	RODUCTI	[ON	1
	1.1	Overviev	v of A-SMART	1
	1.2	System R	Lequirements	2
		1.2.1	Minimum Hardware Requirements	2
		1.2.2	Recommended Hardware Setup	2
		1.2.3	Prospective Users of A-SMART	3
2.	SET	TING UP S	SCENARIOS AND MODEL INPUTS	3
	2.1	Managin	g ORBATs and Scenarios	4
		2.1.1	Creating/Loading an ORBAT or Scenario	4
		2.1.2	Readiness Levels and Jobcode to ECN mappings	5
	2.2	Viewing	and Editing the Force Structure	6
		2.2.1	Viewing and Editing the Force Structure	6
		2.2.2	Linking Units and Subunits	7
		2.2.3	Assigning Personnel to a Unit/Subunit	8
		2.2.4	Viewing and Editing Model Personnel Levels	9
		2.2.5	Viewing and Adjusting Major Systems Levels	13
		2.2.6	Creating Icons	13
		2.2.7	Force Structure Migration Plan Output	14
	2.3	Defining	Task Groups	15
		2.3.1	Viewing, Adding and Removing Task Groups	15
		2.3.2	Adding Force Elements to a Task Group	16
		2.3.3	Viewing personnel and Systems Assigned to a Task Group	17
		2.3.4	Creating Icons	17
	2.4	Setting-u	p Operations	17
		2.4.1	Creating Operations and Phases	18
		2.4.2	Assigning Task Groups to an Operation	19
		2.4.3	Setting-up Rotations	20
		2.4.4	Viewing Personnel and System Levels by Task Group	20
		2.4.5	Viewing Deployment Targets Over the Time Horizon	21
		2.4.6	Defining Operation Settings	22
		2.4.7	Setting Phase Parameters	24
		2.4.8	Reviewing Unit Allocations to Deployments	25
	2.5	Setting P	ersonnel Recruitment and Separation Rates	26
		2.5.1	Rate Adjustment	27
	2.6	Setting N	Iajor System Rates	29
		2.6.1	Base Rates	29
	• -	2.6.2	Varying Rates	30
	2.7	Viewing	and Editing Training Information	31
		2.7.1	The Training Courses Screen	31
		2.7.2	The Training Streams screen	33

3.	SUP	PLIES & S	TRATEGIC LIFT MODULE	35
	3.1	The Logi	stics Setup Screen	35
		3.1.1	Setting Days of Supply levels	38
		3.1.2	Route Planner	38
		3.1.3	Setting Strategic Lift Options	39
		3.1.4	Aircraft and Ship Transport Availability	40
	3.2	The Logi	stics Output Screen	41
	3.3	The Logi	stics Editor	43
4.	RUN	NNING TH	IE MODELS AND VIEWING RESULTS	47
	4.1	Personne	l Module	47
		4.1.1	Running the Personnel Module	47
		4.1.2	Viewing Personnel Results	49
		4.1.2.1	Browsing Personnel Level Results	50
		4.1.2.2	Using Traffic Lights to Identify Failure Points	52
		4.1.2.3	Browsing Personnel Movement Results	53
		4.1.2.4	Browsing Time In Rank Results	53
		4.1.2.5	Browsing Staff Years Results	53
		4.1.2.6	Browsing Corp/Rank Matrix	54
		4.1.3	Training Results	55
	4.2	Major Sy	stems Module	57
		4.2.1	Running the Major Systems Module	57
		4.2.2	Viewing Major System Results	57
~	TNIT	EDDDETA		50
э.		Democran	NG MODEL OUTPUIS	39
	<b>J.I</b>	Fersonne	Interpreting the Desults	<b>39</b>
		5.1.1 5.1.2	A chieving a Sustainable Force	39
		0.1.2 E 1 0 1	A diverting a Sustainable Force	62
		5.1.2.1 E 1 2	Adjusting Personnel Model Rates	62
		5.1.5 E 1 4	Eliminating Training Dottlenecks	65
	59	0.1.4 Major Su	Interpreting Scenarios involving Deployments	70 79
	5.2	F 2 1	Insights into the Seture and Analysis of Major System Module	73
		3.2.1	Baselta	70
		5.2.1.1	Basic Example	73 74
6.	A-S	MART DA	TA FORMATS AND LOADING PROCEDURES	78
	6.1	Introduct	10 <b>n</b>	78
		6.1.1	lypographic conventions	78
	6.Z	UKBAT		79
		6.2.1	Quick Start	79
		6.2.1.1	Major Systems	80
		0.2.2	Data Sources	81
		6.2.2.1	Personnel Entitlements	81
		<i>b.Z.Z.1.1</i>	LMK6A22	81
		0.2.2.1.2	Drians-A	ð1

	6.2.2.1.3	SED Review Team	. 81
	6.2.2.2	Major Systems Entitlements	. 82
	6.2.2.2.1	Defence Entitlements System	. 82
	6.2.2.2.2	SED Review Team	. 82
	6.2.3	Data Files	. 82
	6.2.3.1	Structure File	. 82
	6.2.3.1.1	Structure File Format	. 82
	6.2.3.2	Personnel Entitlement Data	. 83
	6.2.3.2.1	Control file Format	. 83
	6.2.3.2.2	Entitlement File Format	. 84
	6.2.3.3	System Entitlement Data	. 85
	6.2.3.3.1	Control File Format	. 85
	6.2.3.3.2	Entitlement File Format	. 86
6.3	Model D	ata	. 87
	6.3.1	Personnel Asset	. 87
	6.3.1.1	Quick Start	. 87
	6.3.1.2	Personnel Asset Data Format	. 87
	6.3.2	Jobcode Map	. 89
	6.3.2.1	Quick Start	. 89
	6.3.2.1.1	Export Jobcode Map	. 90
	6.3.2.1.2	Import Jobcode Map	. 90
	6.3.2.2	Jobcode Map Format	. 90
	6.3.3	Model Rates	. 91
	6.3.3.1	Quick Start	. 91
	6.3.3.1.1	Export Model Rates	. 92
	6.3.3.1.2	Import Model Rates	. 92
	6.3.3.2	Model Rates Format	. 92
	6.3.3.2.1	Variable Rates	. 93
6.4	Backup a	nd Restore	. 94
	6.4.1	Database Backup	. 94
	6.4.1.1	Database Backup	. 94
	6.4.1.2	Database Restore	. 95
	6.4.1.3	File Format	. 96
	6.4.2	ORBAT Backup	. 96
	6.4.2.1	ORBAT Backup	. 97
	6.4.2.2	ORBAT Restore	. 97
	6.4.2.3	File Format	. 97
6.5	Tree Stru	cture Sample	. 98
6.6	Personne	I Entitlement Sample	. 98
6.7	Major Sy	stems Entitlement Sample	103
6.8	Major Sy	stems Data Load	104
	6.8.1	Custom Loaders	104
	6.8.2	LINKING Department ID	105

	6.9	Training	g Streams	
		6.9.1	Data Capture	
		6.9.1.1	Courses	
		6.9.1.2	Streams	
		6.9.2	Data Processing	
		6.9.2.1	Streams	
		6.9.3	Recommendation	110
7.	SUN	/MARY		110

# Figures

Figure 1:	The Scenarios Screen	3
Figure 2:	The Jobcode to ECN Mapping Tab	6
Figure 3:	The Force Screen	7
Figure 4:	Linked Subunits	
Figure 5:	Viewing Assigned Personnel	9
Figure 6:	Model Personnel Levels Derived from User-assigned Levels	
Figure 7:	Setting Personnel Levels and Training Availability	
Figure 8:	Monthly Personnel Target Numbers	
Figure 9:	The Systems SED Tab	
Figure 10:	The Icon Editor	14
Figure 11:	Force Personnel Entitlement Migration Plan	
Figure 12:	The Task Groups Screen	
Figure 13:	Viewing Assignments to Task Groups	17
Figure 14:	The Operations Screen	
Figure 15:	Viewing Entitlement Levels by Task Group	
Figure 16:	Viewing Deployments Geographically	
Figure 17:	Setting Casualty Rate Parameters at the Operation Level	
Figure 18:	Setting Return to Duty Rates	
Figure 19:	Setting Equipment Operation Parameters	
Figure 20:	Setting Casualty Rates for a Conventional War Type Operation	
Figure 21:	The Deployment Review tab	
Figure 22:	Editing Recruitment Rates	
Figure 23:	Varying Rates Over Time as a Percentage (Y-Axis) of the Base Rates	
Figure 24:	System Base Rates	
Figure 25:	System Varying Rates	
Figure 26:	The Training Courses Screen	
Figure 27:	The Training Module Courses Screen	
Figure 28:	Viewing Training Stream Information	
Figure 29:	Amending On-the-Job-Training and Time-In-Rank Periods	
Figure 30:	Specifying Class 1 Inputs	
Figure 31:	Logistics Copy From button	
Figure 32:	Setting Days of Supply Levels	
Figure 33:	Setting an Air Route	
Figure 34:	Strategic Lift	
Figure 35:	Setting Transport Availability	41
Figure 36:	Logistics Output Screen	
Figure 37:	Viewing Strategic Lift Results	
Figure 38:	Viewing Logistics Requirements Plots	
Figure 39:	Logistics Editor Screen	
Figure 40:	Editing Ammunition Settings	
Figure 41:	Editing Locations	
Figure 42:	System Types Input Screen	
Figure 43:	System Classes Input Screen	
Figure 44:	Running the Model	
Figure 45:	The Personnel Results Screen	
Figure 46:	Filtering Results by Corps, Rank and Operation	51
Figure 47:	The Personnel Results Options Tab	
Figure 48:	Viewing Promotions In and Out over the Time Horizon	
~	-	

Figure 49:	Staff Years Results	. 54
Figure 50:	Key to Sustainability Matrix	. 55
Figure 51:	Personnel Sustainability Matrix	. 55
Figure 52:	Viewing Training Results	. 56
Figure 53:	Viewing Major Systems Results	. 58
Figure 54:	Setting Options for Major Systems Results	. 58
Figure 55:	Unsustained RAA, RAINF Populations	. 60
Figure 56:	Traffic Light System for Personnel Results	. 61
Figure 57:	Unsustained RAA Population	. 62
Figure 58:	High Separation Rate Causing Unsustained Population Levels	. 63
Figure 59:	RAA Ground Based Air Defence Recruitment Rate	. 63
Figure 60:	Increasing the RAA Ground Based Air Defence Recruitment Rate	. 64
Figure 61:	RAA Ground Based Air Defence Separation Rate	. 64
Figure 62:	Decreasing the RAA Ground Based Air Defence Separation Rate	. 64
Figure 63:	Almost sustained RAA Population Levels	. 65
Figure 64:	Unsustained RAA Population Level	. 66
Figure 65:	Sustained PTE Population Level with Excess	. 66
Figure 66:	Critically Failing LCPL Population Level	. 67
Figure 67:	Training Results for Ground Based Air Defence, PTE, SUBJ 1 CPL	. 68
Figure 68:	Personnel Requirements for SUBJ 1 CPL Course	. 68
Figure 69:	Sustained Ground Based Air Defence Population Level	. 69
Figure 70:	Sustained LCPL Population Level	. 69
Figure 71:	Promotions into LCPL	.70
Figure 72:	Personnel Results Involving an Operation	.71
Figure 73:	RAA Personnel Results	.72
Figure 74:	Major Systems Sample Force Structure	. 74
Figure 75:	Major Systems Sample Results - Bushmaster IMTV Population Levels	. 75
Figure 76:	Operation Set-up	.76
Figure 77:	Major Systems Sample Results - Bushmaster IMTV Population Levels with Operatio	n
C		. 77
Figure 78:	Major Systems Sample Results - Bushmaster IMTV Population Levels with Operatio	n
C	and Increased Deep Maintenance Capacity	. 77
Figure 79:	Major Systems Sample Results - Bushmaster IMTV Population Levels with Operatio	n
C	and Increased Repair Pool	. 78
Figure 80:	A-SMART Data Loading Tools	. 79
Figure 81:	Entitlement Data Load Screen Personnel	. 80
Figure 82:	Output Shown When Importing Personnel	. 80
Figure 83:	Jobcode Editing	. 89
Figure 84:	Model Rates Page	. 91
Figure 85:	Backing Up the asm Database	. 94
Figure 86:	Restoring the asm Database	. 95
Figure 87:	Creating a New asm Database	. 96
Figure 88:	Tree Structure Hierarchy	. 98
Figure 89:	Example Course Spreadsheet	107
Figure 90:	Career Profile - Rifleman	108
Figure 91:	Structure of Training Data	109

# Glossary

ARA	Australian Regular Army
ASM	Army Sustainment Model
A-SMART	Army Sustainability Modelling Analysis and Reporting Tool
CFSM	Combat Force Sustainment Model
DES	Defence Entitlements System
DGPP-A	Director General Preparedness and Plans - Army
ECN	Employment Category Number
FIC	Fundamental Inputs to Capability
FSMT-P	Force Sustainability Modelling Tool Prototype
GB	Gigabyte
GRES	General Reserve
HQ	Headquarters
HTA	Hardening the Army
JOLTS	Joint Operational Logistics Tool Suite
LHQ	Land Headquarters
MAE	Manual of Army Employment
OJT	On-the-Job-Training
ORBAT	Order of Battle
RAM	random access memory
SED	Single Entitlement Document
SIGC	Stock Item Group Code
SO	Special Operations
TC-A	Training Command – Army
TFSM	Task Force Sustainment Model
TIR	Time-In-Rank
TMP	Training Management Package
TPMSS	Training Management Pack Support System

# 1. Introduction

## 1.1 Overview of A-SMART

The original driver for developing a sustainability analysis tool for Army was to examine the viability of the Combat Force Sustainment Model (CFSM), with various policy options under varying scenarios. The CFSM was proposed by Land Headquarters (LHQ) in 2002 as the plan for Army to meet the 2000 White Paper guidance. In 2002 DSTO was tasked to analyse this proposal and assess the level of risk to its viability. One of the major risks identified was the perceived lack of quantitative understanding of how the proposed force would evolve over time, i.e. how sustainable it was. This prompted the effort to develop a dynamic modelling tool that could be used to analyse sustainability aspects of the CFSM. The Force Sustainability Modelling Tool Prototype (FSMT-P) was delivered for evaluation by LHQ in mid 2003. The tool was then used to assess the sustainability of the Hardening the Army (HTA) proposed force structure.<sup>1</sup>

Following this work, the Director General Preparedness and Plans – Army (DGPP-A) approached DSTO to assist in developing the Army Sustainment Model (ASM). It was proposed that the ASM be developed by extending the capability of the FSMT-P to model the whole of the Army and not the combat elements only. In November 2005 a MINCS(L) application was approved to fund the development of a prototype version of the ASM; it was intended that in this phase of development, the level of funding would support the provision of limited function sustainability model only, which would support ongoing experimentation and modelling within Army. Whereas the Phase 1 MINCS(L) would fund the development of a software tool to model personnel, individual training, supplies and major systems, further support in the form of a Phase 2 MINCS(L) would be required to extend the functionality to other Fundamental Inputs to Capability (FIC), including facilities and collective training, as well as costing analysis and optimisation. The software has been renamed the Army Sustainability Modelling Analysis and Reporting Tool (A-SMART) to better describe its functionality and to distinguish it from its predecessor.

A-SMART is a strategic-level force structure analysis tool, designed to support high-level force structure decision-making and capability analysis, as well as providing inputs for mobilisation and logistical planning. A-SMART will forecast the sustainability of a particular force structure as it attempts to meet expected operational requirements and will enable Army to better answer high-level questions, such as:

- What deployable capabilities can Army provide?
- How long can they be deployed for?
- How much will a deployment cost?
- What resources are required?
- If an operation cannot be sustained, why? That is, information on problem corps, ranks or trades, major systems types/variants.
- How do operational levels affect the required training schools force structure?

<sup>&</sup>lt;sup>1</sup> Richmond, M.K., Filinkov, A. and Nicholson, R.J.F. (2007). Sustainability of the Interim Objective Force: Personnel (U) DSTO-TR-2039.

Further, A-SMART will allow for the better design of the Army force structure in accordance with government guidance and for high-level decision makers to make informed planning decisions readily incorporating analysis of all of the FICs.

Overall the purpose of A-SMART is to:<sup>2</sup>

- Provide a transparent and predictable basis for force structure decision making;
- Analyse the extent to which Army force structure meets Government directed capability requirements (the Army Sustainment Diagram);
- Assist with resource planning;
- Assist with personnel policy planning;
- Justify any requests for additional resources/capabilities; and
- Provide an overarching sustainment model to encompass the CFSM, TFSM and SOCFSM.

## **1.2 System Requirements**

A-SMART was developed using the Microsoft .NET 3.0 Framework. To run the program, this Framework needs to be installed. The installer for this software is available from the Microsoft web site.

Storage of scenario information in A-SMART is handled by a relational database which was created with PostgreSQL version 8.3. PostgreSQL is a relational database management service and must also be installed and properly configured in order to run A-SMART.

#### 1.2.1 Minimum Hardware Requirements

2 GB RAM Pentium 4 processor or equivalent Windows XP operating system or higher 1280x1024 16-bit colour display

1.2.2 Recommended Hardware Setup

3 GB RAM3.0 GHz processor or higher64-bit Windows operating system1600x1200 32-bit colour display

<sup>&</sup>lt;sup>2</sup> Minor Capability Submission (Land) 32.07 Army Sustainment Model (ASM).

1.2.3 Prospective Users of A-SMART

A-SMART models preparedness and mobilisation of the land force. Consequently, a significant background knowledge of the Australian Army doctrine in this area<sup>3</sup> is required (or advice from someone who does) to setup scenarios that are appropriate and to interpret results. Furthermore, the software tool is detailed and (although efforts have been made to simplify it) consequently so is the user interface. Therefore, some training on the software is necessary and this User Manual & Results Interpretation Guide should be used as an aid during training and to assist with ongoing use of the tool.

# 2. Setting up Scenarios and Model Inputs

To create and run a scenario, you need to open the A-SMART application. Double click on the *SMARTAAS.exe* icon which is located in the *Release* folder. When the program has loaded, the *Scenarios* page will be displayed (Figure 1).

The process of creating and running a scenario, as well as viewing the results, has been broken down into a series of steps. Users can navigate through each step using the tabs located on the left-hand side of the screen. Some of these tabs display the results after a module is run and consequently, if selected, these tabs will not display any meaningful information until after a scenario has been set up and the model run.



Figure 1: The Scenarios Screen

<sup>&</sup>lt;sup>3</sup> Preparedness and Mobilisation (Provisional), Australian Defence Doctrine Publication 00.2, Commonwealth of Australia 2004.

# 2.1 Managing ORBATs and Scenarios

#### 2.1.1 Creating/Loading an ORBAT or Scenario

Within A-SMART, an ORBAT contains all of the information about a force structure including the unit hierarchical structure, personnel and equipment entitlement levels, jobcodes, ranks and locations. A-SMART has been designed to allow an ORBAT to be used across a range of scenarios, where a scenario is defined here by the setup of operations and changes to recruitment, separation and attrition rates. Separating the ORBAT from the scenarios in the databases means the same force structure does not need to be saved multiple times when model input changes only are being investigated, significantly reducing the memory requirements.

Once set-up, an ORBAT and scenario contains all of the information necessary to run the model. Note, however, that some information does not save to a specific ORBAT or scenario and, when altered, such settings are effectively changed for all future model runs of the particular database; non-scenario specific inputs include Readiness Levels, Jobcode to Employment Category Number (ECN) mappings, Training Courses, Training Streams and Logistics Settings.<sup>4</sup>

To create a completely new ORBAT, click the *New ORBAT* button at the top left of the screen; the ORBAT can then be built up within the tool (see Section 2.2) or loaded from external data (see Section 6.2). Alternatively, users can choose an existing scenario by selecting it from the *Current ORBAT* drop-down box. Users can also clone an ORBAT (to use as a template or start point for a new ORBAT) by selecting the ORBAT to be cloned and clicking the *Clone ORBAT* button. This may be useful if, for example, a user wants to amend the force set of an existing ORBAT; say by duplicating a battalion.

To create a new scenario for a particular ORBAT, first select the ORBAT and then click the *New Scenario* button at the top right of the screen. Alternatively, users can choose an existing scenario by selecting it from the *Current Scenario* drop-down box. Users can also clone a scenario (to use as a template or start point for a new scenario) by selecting the scenario to be cloned and clicking the *Clone Scenario* button. This may be useful if, for example, a user wants to use the force set and task groups defined in an existing scenario, but wishes to vary, e.g., the recruitment or separation rates over the time horizon<sup>5</sup> to explore what effect this would have on sustainability.

The *Scenarios* screen also allows users to change the name of the ORBAT or scenario, as well as the initial year, start month and end month of the time horizon. For the purposes of marrying model inputs to real-world dates, month zero corresponds to January in the specified initial year; so a scenario defined with an initial year of, say, 2009 and a time horizon start month of 6 and end month of 90 would cover the period June 2009 to December 2017.

<sup>&</sup>lt;sup>4</sup> Note that the individual training and logistics settings referred to here can be viewed and altered in the *Training Courses, Training Streams* and *Logistics Editor* tabs (discussed further in Sections 2.7 and 3.3, respectively).

<sup>&</sup>lt;sup>5</sup> Time horizon is defined to be the period of interest for the model run scenario; note this period is arbitrary and can be set by the user.

There is functionality within the *Scenarios* screen that allows the backup and restoration of particular ORBATs, including the details of any related scenarios; this facility is discussed further in Section 6.4.2. To backup the entire database it is best to make use of the database administration tools available within the open source database, pgadminIII (which is installed during the A-SMART installation process); see Section 6.4.1.

It should be noted that there is no need to explicitly save any changes made in the program. Because A-SMART writes to a database to store scenario information, changes are saved as soon as they are made. Some parts of the program incorporate 'Undo' functions, notably the Force Structure editor, but users are warned to take care when modifying settings and data.

#### 2.1.2 Readiness Levels and Jobcode to ECN mappings

The *Scenarios* screen also contains some options which are global defaults – that is, changing these values will change them for all scenarios.

The first of these global defaults are the readiness levels. Each unit in a task force needs to pass through a number of readiness levels as it mobilises before a deployment. Readiness levels in A-SMART are equivalent to the concept of readiness notices. By changing the number and length of the readiness levels, the user can alter the duration for a unit to build-up towards deployment and different units can be assigned to the different initial readiness levels.

In Figure 1, five levels are defined (arbitrarily without loss generality); *Base, Low, Medium, High* and *On Call*; the *Time at Level* values for the six levels are 24, 12, 6, 3 and 3 months, respectively. This means that a unit initially at the *Base* readiness level would progress after 24 months to *Low,* as well as 12 months each at *Low,* 6 at *Medium,* 3 at *High* and 3 at *On Call,* to progress to a deployment; a total build-up time of 48 months. Note that every element of the force structure is currently required to have a readiness level defined; this includes enabling components that presumably would never deploy (this may be changed in future versions).

Clicking on a readiness level allows the user to change the level's title, the time a unit must spend at that level before progressing to the next, and the colour in which the level will be displayed in the personnel plots once the model has been run. Clicking the *[+] Level* and *[-] Level* buttons allows extra levels to be added and deleted.

The personnel module imports data from Single Entitlement Documents (SED), where position requirements are specified by jobcode; however, personnel are defined in terms of their trade specialisations and career profile by ECNs and consequently ECNs are used within the modelling tool to define personnel classes. As a result, jobcodes are mapped to ECNs using weightings to spread the population counts (which are sourced from the SEDs). The mappings of jobcodes to ECNs are set globally (see Figure 2). The *Jobcode / ECN* tab can be used to modify jobcode to ECN mappings, as well as import and export jobcode to ECN information to or from a specially formatted Excel spreadsheet. When changing the ECN mappings it is important that the combined Weight values for all ECNs in a jobcode add up to 1; the *Balance Weights* button can be used to set an even weighting for ECNs, for instance if there are 2 ECNs for a jobcode it will set both ECNs weight value to 0.5.



*Figure 2:* The Jobcode to ECN Mapping Tab

# 2.2 Viewing and Editing the Force Structure

One of the most important model inputs is the definition of the in-barracks force structure. The *Force* screen (located at the top left, see Figure 3) allows users to view and edit the current state of the force and how it will change going forward. Most users will start with a default ORBAT/scenario with the current force structure data loaded; however, some users may wish to amend this data and the procedures to do so are described below.

#### 2.2.1 Viewing and Editing the Force Structure

Under the *Setup* tab, the force is represented as a tree with elements that can be expanded to show the sub elements. Users can click the [+] and [-] boxes next to elements to expand and collapse them in the tree. The *Expand All* and *Collapse All* buttons expand and collapse all elements in the tree. Note, however, that performing an *Expand All* on large and complex force structures (e.g. the complete Australian Army force structure) may take upwards of a minute.

New formations or units can be added to the force by clicking the *New Formation* or *New Unit* button, respectively. To add subunits, select a unit by left-clicking on it and then press the *New Brick* button. Structures containing multiple levels of depth can be created by selecting a subunit and clicking on *New Bricks*; note that subordinate units can be added to any level in the organisational tree and effectively create any number of levels.

Subunits and even whole units can also be cut, copied and pasted into new locations within the tree. Right clicking on an element will bring up these options. Subunits can be cut from one unit and pasted into another by cutting the subunit, then right-clicking the destination unit and selecting *Paste*. Note that a unit's subordinate subunits and entitlement data are transferred (if loaded). *Ctrl Z* will initially undo any of these changes if an error is made.

Current Scenari	io: User Manual Sc	enario													
Scenarios	Setup Migr	ation Plan													
Force	IRO New Roomal	tool Contract Cont	Indiated Income	(Durned at)	Trends All	Execution and									
Task groups	in new rolling	New Brick Line	CHARGE College	Expand we Co	subsc wi	EXPLANT POINT									
Concretions	Icor	Unit/Brick		Timeline		2	Selected	Sub	unit 0 / 9				1		
openace a	C the	a famating	08 09	10 11 12	13 14	15 1	Active Rand	te 0 to 3	2000						
diance Setup	in Unional	wn formadon 16	_				1.0.000.00	(verse	00105			1.	1 1 1 1 1		
gistics Output	E Tes	t Unit 0					Location	TOM	PRICE			1	Det Ma		
distics Editor	8	Subunit 0 / 0	Address and the				Readiness	On C	all			~	Set All		
Indel Dates	1	Subunit 0 / 0 / 0A	Charles in succession	-			Turne	Celu					Set All		
HODEL Kates		Subunit 0 / 0 / 08		and the second	a second second		1.325m	Course		-			and see		
In The Model		Subunit 0 / 1			a second second		Personnei	SED	Systems SED	Personnel	Levels	System	n Levels	Personnel	Icon
sonnel Results		Subunit 0 / 2			· · · · · ·									F. 4.	
aining Reads		Suburit 0 / 3					Personnel 3	in Group			-	Perm	doc sider	Codes	
		Suburit 0 / 4	-				Jobcode		Title	Rank C	LC )	ldd F	ter		
stem itesuits		Suburit 0 / 6	Transfer of the local division of the				37042	RES-A E	LECTRICIAN LCPL	LCPL 1		Se Se	sarch		
ining Courses		Subunit 0 / 7	e himse a subject of				35647	RES-A 4	ARTGND None WO1 WO1 3 4ANARM None LCPL LCPL 2	- Notes	ove				
ining Duanau		Subunit 0 / 8					31089	REG CM		LOPL 2		3ob	Code	Tr	ade
any steams	F.1	Subunit 0 / 9			a second second		35895	DEC AC	ET ADMINI AARC CCT	SGI 3		11	56 9	A Chief of Arms	/
Tools	12	Suburit 0 / 10	and the second se		a second second		30900	DDG-A C	ST MED ATTENDAN	DTE 1		110	50 8	A Executive	
	1	Subunit 0 / 11	2		a second second		32118	REG MU	L'TIMEDIA TECH PT	PTE 3		110	51 8	A Ops Comman	d
		Subunit 0 / 12	14				33508	REG PH	OTOPR None OPL	CPL 2	É.	342	75 7	A Executive	
		Subunit 0 / 13	100 C		a second second		30471	REG AS	STDENT ASSTDENT:	W02 3	É.	11	76 7	A Logistics	1
		Subunit 0 / 14	6. million (1997)		a second second		41551	RES-A T	HERA OFFR None L'	LTCC 3	8	11	77 7	A Attache/Intel	ligence
		Subunit 0 / 15	and the second se	and the second second	a second second		40373	REG EN	G OFFR None MAJ	MAJ 3		11	78 7	A HR/Training	
		Subunit 0 / 16	Colonia (march)		A DESCRIPTION OF THE OWNER OF						1.0	11	79 7	A Ops Comman	a
		Suburit 0 / 17	Access in the		a second second							30	100 H	EG AUN CPL	
		Subunit 0 / 18										200	XX1 P	EC ACA DIE	
		Subunit 0 / 19	<u>.</u>									30	304 8	EG ACA SGT	
		Subunit 0 / 20	and the second s									30	06 R	EG ACA WOL	
	1 1 C	E UNE 1	_									30	007 R	EG ACA WO2	
	1 I I I I I	e unic 2	-		· · · · · ·							30	306 F	EG AIRCMAN L	OADMASTER CP
	0.10	E UNE 3			_							300	309 R	EG AIRCMAN L	OADMASTER LC
	in tes	A MIRE T	-									30	010 R	EG AIRCMAN L	OADMASTER PT
	(B) Tes	t the G										300	012 F	EG ABRICMAN L	OADMASTER SG
	III Tes	FILME 2	-									30	215 R	EG AIRCMAN L	DADMASTER W
	ill Tes	t Linit A										30	240 F	LG ADVENGRO	VK NORE CPL
	F Tes	t Link 9	-									300	/36 F	EG ADVCARGO	SPECIAL
												300	158 0	EG ADVCARGO	CDEC DUPL
A. Bell												30	190 0	EG ATP DISPAT	CHER CR
and and												300	001 B	EG AIR DISPAT	CHER LCPL
-												200	145	DO AND DECRAT	An and a start to

*Figure 3:* The Force Screen

A unit or subunit's name can be changed by selecting it from the tree and editing the text in the *Selected* box in the top right-hand corner of the screen. In addition, each subunit has a *Location*, a *Readiness* level and a *Type*. All of these properties can be changed by clicking on a subunit and selecting options from the relevant dropdown boxes in the top right-hand corner of the screen.

A unit's *Location* reflects where it is usually based, and may be relevant for logistics and facilities purposes in the future (but is not currently used in the model runs). The *Readiness* level impacts on the time required for a unit to mobilise for a deployment, as explained in Section 2.1.2 above. A unit's *Type* refers to its force type or function (a property distinct from the corps of its personnel), and is used in determining casualty rates during deployments; examples of unit type include, infantry, armour, air defence, medical etc.

*Location, Readiness* level and *Type* can be set at any level in the force structure tree. Caution should be used in changing these settings at the higher levels in the tree (e.g. formation level), since this will assign any change to all subordinate units overwriting any existing subunit settings.

#### 2.2.2 Linking Units and Subunits

Elements in the force tree can be linked together to provide for force migration; as the linked units will have different personnel entitlements, this functionality allows the personnel population target to be set to change over time. Linked elements must reside at the same level of depth in the tree (i.e. a unit cannot be linked with a subunit). To link two elements, select the first element then ctrl-click the second element and click the *Link* button at the top of the screen (see Figure 4).

#### DSTO-GD-0595



Figure 4: Linked Subunits

There should now be a point in the timeline<sup>6</sup> where the first element ends and an arrow points to the start of the second element. This point represents the time at which the first element will transform into the second. The timing of this migration point can be changed by clicking near to the arrow and dragging the ends of the two bars in the timeline. The numbers displayed on the timeline represent the calendar years, thus 08 equals 2008 and so on. If a unit is planned to be created during the timeline of the model run, the start date can be changed by clicking near to the start of the relevant Gantt chart and dragging across to the right to the desired date. Similarly, if a unit is to be shut down prior to the end of the time horizon the end of the Gantt chart can be dragged to the left.

#### 2.2.3 Assigning Personnel to a Unit/Subunit

If a new unit/subunit hierarchy has been created (and consequently does not have any entitlement data associated with it) or you wish to amend current entitlement data, the next step is to allocate personnel and equipment to each unit/subunit. Users can view the personnel entitlement levels for any unit or subunit by left-clicking on the element and selecting the *Personnel SED* tab on the right hand side of the screen. Current personnel entitlement levels are displayed in the *Personnel In Group* box in the middle of the screen. Figure 5 shows an example of personnel establishment; 12 different job types have been assigned to the selected subunit, the last of which is jobcode 40897 – *REG INF OFFR None LT*. Note that when a higher level unit is selected only the personnel assigned at that level (e.g. into a HQ) will be displayed and not allocations to subunits that fall under the hierarchy of that unit.

Personnel can be added to force elements at any level in the tree. To add additional personnel to a subunit, left click on the subunit to select it. When the *Personnel SED* tab is selected, a list of positions (as defined in the entitlement database) appears. These jobcodes can be filtered by *Jobcode* and/or *Title*. Once the desired position has been located, it can be added to the subunit by selecting it and then clicking the *<- Add* button.

<sup>&</sup>lt;sup>6</sup> We define timeline to be the period during which the force is planned to exist including start and end dates.

Personn	el SED	Systems SED	Perso	onnel Le	evels S	ystem Level	s Personnel	Icon		
Personnel	In Grou	IP				Available Jo	ob Codes			
Jobcode		Title	Rank	OLOC	<- Add	Filter —				_
37939	RES-A	MGROPSOS NONE V	WO:	1	Remain	Search				
60195	REG RS	SM AACC WO1	WO:	1	Remove					
38885	RES-A	PHOTOPR NONE CP	CPL	1		Job Code	Tra	de		^
30056	REG AL	OVCARGOSPEC CPL	CPL	3		1156	9A CHIEF OF ARM	IY		
30376	REG AS	SPVLNGISD NONE C	CPL	1		1160	8A EXECUTIVE	n		
35131	REG TE	ECH LABORATORY P	PTE	2		1175	7A EXECUTIVE			
34413	REG SF	VCMDSPTS NONE 5	SGT	3		1176	7A LOGISTICS			
31664	REG EL	ECTRICN CPL	CPL	1		1177	7A ATTACHE/INTE	ELLIGENCE		
40897	REG IN	CMBT CLK NONE W	IT.	3		1179	7A OPS COMMAND	D		
10057	1120 11			2		30000	REG ACA CPL			
<				>		30001	REG ACA LCPL			
Job Desc	ription	EG INE OFFR NONE	IT			30004	REG ACA SGT			
Job Code		0907				30006	REG ACA WO1			
01.00		0057				30007	REG ACA WO2	ADMASTER (	DI	
0200	2			$\sim$		30009	REG AIRCMAN LO	ADMASTER L	.CPL	
MLOC	3	1				30010	REG AIRCMAN LO	ADMASTER P	TΕ	
GRES	C	)		-		30012	REG AIRCMAN LO	ADMASTER S	GT	~
				$\checkmark$		<			>	

*Figure 5: Viewing Assigned Personnel* 

Once a jobcode has been added to a subunit, the number of personnel allocated to that jobcode will default as 1 but can be amended; left-click to select a jobcode from the *Personnel In Group* box (which displays the jobcodes currently assigned to the subunit). Below, three text fields will display the number of personnel assigned to that job in the subunit by OLOC, MLOC and General Reserve (GRES). The default values after adding a jobcode to a subunit are 1 at OLOC, 1 at MLOC and 0 GRES. The MLOC and OLOC values refer to Australian Regular Army (ARA) positions, and should in almost all cases be set to the same number (note: the concept of MLOC and OLOC may be removed from future versions of the tool as it is now an outdated Army concept). GRES refers to reservists, who are included to allow the user to view the unit entitlement; however, reserves are not currently modelled. In Figure 5, the position 'REG INF OFFR None LT' has been selected and the number of personnel assigned to the jobcode is 3 at OLOC, 3 at MLOC, and 0 at GRES.

#### 2.2.4 Viewing and Editing Model Personnel Levels

Although users assign personnel to subunits using the jobcode types defined in the PMKeyS database, the model uses a different categorisation system for defining personnel levels. This system is based on ECNs, and is used because it allows the incorporation of training data into the modelling process. The model automatically converts the personnel assigned to a subunit by the user (in jobcodes) into personnel levels that the tool can use to model aspects of individual training (ECNs or trade stream); the mapping from jobcode to ECN has been discussed previously in Section 2.1.2.

To view the personnel levels that the model will make use of, click on the *Personnel Levels* tab to show a list of positions (by trade stream), as well as the ARA and GRES personnel entitlement levels (noting that at present we are not modelling reservists). The levels shown in this tab may be different to those set in the *Personnel SED* tab, because a single PMKeyS position may be able to be filled by (and is thus spread between) personnel from a number of different corps and trade streams and, furthermore, if a jobcode is not mapped to a trade stream (which is modelled) it will not be represented at all (for example, we do not currently model personnel from the AABC, Australian Army Band Corps).

In Figure 6, two jobcodes have been assigned to the subunit in the Personnel SED tab; however, each of these jobcodes can be filled by personnel from a number of different trade streams. For example, the REG ACA RAAC CPL position is apportioned over five different trade streams.

Personner SED	Systems SED	Person	nel Levels	System Levels				
Personnel In Group								
Jobcode	Title	Rank	OLOC MLO	C GRES	<- Ad			
30004 REG ACA	SGT	SGT	1 1	0				
Personnel SED	Systems SED	Perso	nnel Levels	System	Leve			
Personnel SED	Systems SED	Perso	nnel Levels	System	Leve			
Personnel SED Strea	Systems SED	Perso Rank	nnel Levels	System	Leve			
Personnel SED Strea All selected items	Systems SED	Perso Rank All rank CPI	nnel Levels OLOC 0.167	System ARA 2 0.167	Leve			
Personnel SED Strea All selected items ASLAV - Crewma ASLAV - Crewma	Systems SED	Perso Rank All rank CPL	nnel Levels OLOC s 2 0,167 0 167	System ARA 2 0.167 0.167	Leve			
Personnel SED Strea All selected items ASLAV - Crewma ASLAV - Crewma Carroo Specialist	Systems SED	Perso Rank All rank CPL CPL SGT	nnel Levels OLOC s 2 0.167 0.25	System ARA 2 0.167 0.167 0.25	Leve			
Personnel SED Streat All selected items ASLAV - Crewma ASLAV - Crewma Cargo Specialist - Clerk Admin -	Systems SED m n SPEC 25MM	Perso Rank All rank CPL CPL SGT SGT	nnel Levels OLOC s 2 0.167 0.25 0.25	System ARA 2 0.167 0.167 0.25 0.25	Leve			
Personnel SED Streat All selected items ASLAV - Crewma ASLAV - Crewma Cargo Specialist - Clerk Admin - M113 -	Systems SED	Perso Rank All rank CPL CPL SGT SGT CPI	nnel Levels OLOC s 2 0.167 0.167 0.25 0.25 0.333	System ARA 2 0.167 0.167 0.25 0.25 0.333	Leve			
Personnel SED Streat All selected items ASLAV - Crewma ASLAV - Crewma Cargo Specialist - Clerk Admin - M113 - MBT - Crewman	Systems SED am n SPEC 25MM	Perso Rank All rank CPL CPL SGT SGT CPL CPL CPL	nnel Levels OLOC s 2 0.167 0.25 0.25 0.333 0.167	System ARA 2 0.167 0.25 0.25 0.333 0.167	Leve			
Personnel SED Strea All selected items ASLAV - Crewma ASLAV - Crewma Cargo Specialist - Clerk Admin - M113 - MBT - Crewman MBT - Crewman	Systems SED am n SPEC 25MM 	Perso Rank All rank CPL CPL SGT SGT CPL CPL CPL	nnel Levels OLOC s 2 0.167 0.25 0.25 0.333 0.167 0.167	System ARA 2 0.167 0.25 0.25 0.333 0.167 0.167	Leve			
Personnel SED Strea All selected items ASLAV - Crewma ASLAV - Crewma Cargo Specialist - Clerk Admin - M113 - MBT - Crewman MBT - Crewman Parachute Ringer	Systems SED am n n SPEC 25MM - Leopard SPEC 120MM (	Perso Rank All rank CPL CPL SGT SGT CPL CPL CPL SGT	nnel Levels OLOC s 2 0.167 0.25 0.25 0.333 0.167 0.167 0.25	System ARA 2 0.167 0.25 0.25 0.333 0.167 0.167 0.25	Leve			
Personnel SED Strea All selected items ASLAV - Crewma ASLAV - Crewma Cargo Specialist - Clerk Admin - M113 - MBT - Crewman MBT - Crewman Parachute Rigger Supervisor Trans Supervisor Trans	Systems SED am n n SPEC 25MM - Leopard SPEC 120MM ( - port - Driver	Perso Rank All rank CPL CPL SGT SGT CPL CPL SGT SGT	nnel Levels OLOC s 2 0.167 0.25 0.25 0.333 0.167 0.167 0.25 0.012	System ARA 2 0.167 0.25 0.25 0.333 0.167 0.25 0.25 0.012	Leve			

*Figure 6: Model Personnel Levels Derived from User-assigned Levels* 

Important properties in relation to initial population<sup>7</sup>, target<sup>8</sup> and availability levels<sup>9</sup> are set under the *Personnel Levels* tab (Figure 7). They can be set at any level in the unit hierarchy by selecting the desired unit/sub-unit (i.e. for all elements in a formation or for a specific subunit) and by trade specialisation/rank by selecting any entry in the list of streams/ranks displayed (or across all trade specialisations and ranks in the selected unit by clicking *All selected items*). Note a stream must be selected before the text boxes for the properties appear. The properties are:

- *Initial* levels are the population levels at the start of the time horizon being investigated. This allows the investigation of scenarios where populations start over or under the target level. Set as a % of entitlement.
- *Mobilising* targets are the population targets that come into effect when mobilising or during a deployment. Set as a % of entitlement.
- *Non-mobilising* targets are the target population levels when units are neither deployed nor mobilising. Set as a % of entitlement.
- *Ringfenced* personnel are not available to reinforce other higher-priority units. Ringfencing levels can be set anywhere from 0 to 100%.
- *Trainees* level limits the proportion of the selected personnel that can be trained at any given time as a % of entitlement. Thus if a subunit has an entitlement of 20 personnel and the *Trainees* level is set to 25%, a maximum of 5 people will be able to undertake training at any time. Note that this limit is calculated based upon entitlement, regardless of the actual population.
- *Training Staff* dictates the maximum proportion of the group that can be used as training staff (either instructors or support staff) at any given time. Similarly to trainees, the training staff limit is applied based on the entitlement level.

These properties can all be set as a percentage value for both ARA and GRES personnel. The *Reset All* button restores the default values for all properties.

The *Personnel* tab shows the target population numbers for each stream and rank for a selected unit/sub-unit and month; the slider at the bottom of the page can be used to change the month (e.g. Figure 8 is showing the 18 month point of the time horizon). The first row, *All Personnel*, can be expanded (by clicking on the *[+]*) to show a breakdown of personnel by location.

<sup>&</sup>lt;sup>7</sup> Initial population will usually correlate to the current population and represents the level of personnel asset at the start of the model run.

<sup>&</sup>lt;sup>8</sup> Target is the level of personnel to which a unit is entitled; it is usually sourced from the SED documents from PMKeyS but can also be set within the A-SMART application.

<sup>&</sup>lt;sup>9</sup> Availability levels are used to set limits, as a percentage of entitlement targets at any level in the force structure organisational tree, of personnel that are available to be instructors or students.

MART													
Current Scenari	o: User Ma	nual Scenario											
Scenarios	Setup	Migration Plan											
Force	New	Formation Prove Brick	Unlink Expand Expand All Collapse All - Export As										
perations	Sec.	In the Party of	Timeline	Enterted	Test that					1			
istics Setup	scor	Unity brick	08 09 10 11 12 13 14 15 1		Tress serve					1			
tics Output	8	Formation 1	DOWN NEW YORK OTHER ADDRESS OF ADDRESS	Active Rang	0 10 2000	000000			1000	100000			
tics Editor		E Unito		Location	Unknown	location			Y	Set Al			
del Rates		Test Unit 1		Readiness	On Call				Y	Set Al			
The Model		Test Unit 2	the second s						123				
nd Percha		I Test Unit 3		rype	oter					Ser 14			
ne kesus		Test Unit 4		Personnel	SED 5	istems SED	Personnel	Levels	System Lev	els	Personnel	Icon	
ng wasuits		Test Unit 5				Pauk	0.00		Initial				
ITT FUESCIES		I Test Unit 7		All calastas	items.	All ranks	200.05	760	404	100		~~~~	
ng Courses		Test Unit 8		Air Dispatc	her -	SGT	0.5	0.5	A54		V 78, 80	out	
g Streams		If Test Unit 9		Aircraft Life	Support Fi	t W02	0.429	0.4	GRES	100	🗧 %, in	MLOC	0
Tools				Aircraft Life	Support Fi	t SGT	2.273	2.2	in the second second				
				Aircraft Str	actural Fitte	SOT NOT	2 2 2 3	2.2	Mobilising		(a)		
				Arcrewma		SGT	3	3	ARA	100	🖯 %, in	OLOC	3
				Aircrewma	1-	LCPL	3	3	GRES	100	1 % in	MLOC	0
				Aircrewma	1-	PTE	3	3			(W) 100 100		
				Ammunitio	n Technician	SGT	2	2	NonMobile	ing			
				AMS-WLR	AMS	W01	0.053	0.0	ARA	100	2 %, in	OLOC	3
				AMS-WLR	AMS	SGT	0.5	0.5		100			
				AND-WLR	WLR	WU1 SCT	0.053	0.5	OVE2		V8, 81	PRO-	
				Army Emer	gency Resp	WD1	0.05	0.0	100000				
				Army Emer	gency Resp	s SGT	2	2	Ringlences				
				ASLAV - Cr	ewman	WO1	0.123	0.1	ARA	0	🗸 %, in	OLOC	0
				ASLAV - Cr	ewman	CPL	1.833	1.8	GRES	0	3 % in	MLOC	0
				ASLAV - Cr	ewman SPE	W01	0.123	0.1			1.01		
				ASLAV - Cr	ewman SPE	SGT	0.333	0.3	Trainers				
				Assistant D	evenan see ental - Den	WDI	0.657	0.6	ARA	100	3 %, in	OLOC	3
				Assistant D	ental - Dent	SGT	1	1	COPES	100	04.1	MIOC	0
				Assistant D	ental - Dent	t CPL	2	2	unas.		V 74, 11	1999	
				Assistant D	ental - Deni	W01	0.667	0.6	- March - Laboration				
				Assistant D	ental - Den	CPL	2	2	Training50	AT .	0	0.000000	1.2
1.2				Cargo Spec	ialist -	507	0.5	0.5	ARA	0	🐨 %, in	OLOC	0
<i>H</i> .				Clerk Admi	n-	W01	0.05	0.0	GRES	0	3 %, in	MLOC	0
				Combat Cle	rk -	CPI	2	2 4			100 CO.		
TADT				E CANTERIE CA		51		15	Reset All				

Figure 7: Setting Personnel Levels and Training Availability

Personnel SED Systems SED Personr	nel Levels	System	n Levels	Personnel	Icon	1
Stream/Rank	REG	Mobilising	Non-Mobili	sing Trainee	Traine	^
All Personnel	321	321.01	321.01	321.01	321.0	
Air Dispatcher - / SGT	0.5	0.5	0.5	0.5	0.5	Ξ
Aircraft Life Support Fitter - / WO2	0.43	0.43	0.43	0.43	0.43	
Aircraft Life Support Fitter - / SGT	2.27	2.27	2.27	2.27	2.27	-1
Aircraft Structural Fitter - / WO2	0.43	0.43	0.43	0.43	0.43	
Aircraft Structural Fitter - / SGT	2.27	2.27	2.27	2.27	2.27	
Aircrewman - / SGT	3	3	3	3	3	
Aircrewman - / LCPL	3	3	3	3	3	
Aircrewman - / PTE	3	3	3	3	3	
Ammunition Technician - / WO1	1	1	1	1	1	
Ammunition Technician - / SGT	2	2	2	2	2	
AMS-WLR - AMS / WO1	0.06	0.06	0.06	0.06	0.06	
AMS-WLR - AMS / SGT	0.5	0.5	0.5	0.5	0.5	
AMS-WLR - WLR / WO1	0.06	0.06	0.06	0.06	0.06	
AMS-WLR - WLR / SGT	0.5	0.5	0.5	0.5	0.5	
Army Emergency Responder - / WO1	0.05	0.05	0.05	0.05	0.05	
Army Emergency Responder - / SGT	2	2	2	2	2	
ASLAV - Crewman / WO1	0.12	0.12	0.12	0.12	0.12	
ASLAV - Crewman / SGT	0.33	0.33	0.33	0.33	0.33	
ASLAV - Crewman / CPL	1.83	1.83	1.83	1.83	1.83	
ASLAV - Crewman SPEC 25MM GNR / WO1	0.12	0.12	0.12	0.12	0.12	
ASLAV - Crewman SPEC 25MM GNR / SGT	0.33	0.33	0.33	0.33	0.33	
ASLAV - Crewman SPEC 25MM GNR / CPL	1.83	1.83	1.83	1.83	1.83	
Assistant Dental - Dental Assistant ECN 025	9-1 0.67	0.67	0.67	0.67	0.67	
Assistant Dental - Dental Assistant ECN 025	9-11	1	1	1	1	
Assistant Dental - Dental Assistant ECN 029	9-12	2	2	2	2	
Assistant Dental - Dental Hygienist ECN 02	9-2 0.62	0.67	0.67	0.67	0.67	~
< III					>	
18						_

Figure 8: Monthly Personnel Target Numbers

#### 2.2.5 Viewing and Adjusting Major Systems Levels

The method of setting major systems entitlement levels for units/subunit is virtually identical to the way levels are set for personnel. In the *Systems SED* tab (Figure 9), available systems can be searched for by Stock Item Group Code (SIGC) or title, and added to the *Systems In Group* (Figure 9). The quantities of each system initially assigned will again default as 1 but can be altered by changing the numbers in the OLOC field below the *Systems In Group* box.

In the *System Levels* tab, initial, mobilising, non-mobilising and ringfenced levels can be set for each system as a percentage of the entitlement for the selected units/subunits.

Personnel SED	Systems	s SED	Personnel Le	evels S	vstem Levels Personnel Icon				
Systems In Group Available Systems									
System SIGC		SIGC	OLOC ML	<- Add	← Filter				
TANK COMBAT N	11A1 AIM	2350-0025	1 1		SIGC Title				
TANK COMBAT,	FULL TRA	2350-716C	1 1	Remove					
RADIO PORTABL	E VHF 14	5820-0025	1 1		System/Type	~			
GEN SET DIESEL	ENCINE :	6115-0035	1 1		T MG 30 CAL				
HOWITZER LICH		1015-0032	1 1		MG .50 CAL FLEX				
REERIGERATOR	TRANSPC	4110-0005	1 1		A VEHICLES				
REFRIGERATION	UNIT ME	4110-0003	1 1		AIRCRAFT				
WORKSHOP CON	1BAT NET	5411-0056	1 1		ANTI ACFT WPNS				
TRUCK UTILITY	LTWT MC	2320-0003	1 1		RAPIER	=			
TRUCK CARGO L	T MC4	2320-0004	1 1		RBS 70				
TRUCK RECON L	TWT FFR	2320-0156	1 1		ANTI ARMD WPNS				
					B VEHICLES				
					C VEHICLES				
					Tracked - Cat 933				
					<ul> <li>Iracked - D3C</li> <li>Tracked - Even at an OAT 2251</li> </ul>				
					Tracked - Excavator CAT 325L     Tracked - Kanabas D155				
					Tracked - Komatsu D155				
					TRACKED - TD 15     TRACKED - TD 15     TRACKED - TD 15				
					Wheeled - Case 580F				
<			>		Wheeled - Grader 130G				
					Wheeled - Hitachi LX120/W36				
System TANK CO	MBAT M1A1	l AIM			Wheeled - Proline 48-16L Hino				
OLOC 1			<u>&gt;</u>		Wheeled - Rollers				
			<u> </u>		Wheeled - Scraper				
MLOC 1					COMMUNICATIONS				
FTF 1			×		CONTAINERS				
					<ul> <li>BMSS Type A</li> </ul>				
LE 0	LE 0				BMSS Type B     B     B     B     S	~			
			✓		<				

Figure 9: The Systems SED Tab

#### 2.2.6 Creating Icons

An icon can be created for units/subunits at any level in the force hierarchy, allowing for easy visual identification, by selecting a unit and clicking on the *Icon* tab to open the icon editor (Figure 10).

Icons are created by selecting different icon parts from the boxes on the right of the screen. Labels can be added at different positions around the icon, and the current result is displayed in the preview screen. The icon is displayed next to the unit name.



Figure 10: The Icon Editor

## 2.2.7 Force Structure Migration Plan Output

By first clicking the *Migration Plan* tab at the top of the screen and then the *Refresh* button, the user can generate an output similar to the Master Migration Plan currently produced by AHQ for the proposed Army force structure changes (Figure 11). It displays as a Gantt chart and shows the changes in population over time at unit level (but not subunit). On the left hand side of the display is a list of the units at the start of the time horizon and on the right hand side the corresponding unit names, if any, at the end of the time horizon. The population is shown in yearly intervals as part of the chart; a background colour of blue indicates that the population is changing in that year, grey indicates no change. This plan can be exported as a separate file into Excel by clicking on the *Save* button.



Figure 11: Force Personnel Entitlement Migration Plan

# 2.3 Defining Task Groups

Once the scenario's in-barracks force structure and initiation parameters have been defined, the next step in the scenario set-up process is to create task groups using the *Task Groups* screen (Figure 12). Task groups are collections of units and subunits from the force tree that can be deployed together during an operation. A unit or subunit can be assigned to multiple task groups, but if two task groups containing the same unit are deployed at the same time, a clone of the unit is created to meet the needs of the second deployment; however, the over-allocation is highlighted to the user (see Section 2.4.9.). During the model run, personnel are automatically sourced from other parts of the force (if available) to fill the cloned unit.

#### 2.3.1 Viewing, Adding and Removing Task Groups

The collection of task groups for the current scenario is displayed in a similar manner to the force tree in the *Force* screen. A list of task groups is displayed on the left of the screen. Each task group can have multiple subgroups (which may, for example, represent rotational groups) and, where defined, these can be viewed by clicking the *[+]* button next to the name of the task group.

New task groups can be added by clicking *New Group* and deleted with the *Remove* button. Sub-elements can be added to a task group, say to represent rotations, by selecting the group and then clicking the *New Group* button. All elements in the tree can be renamed by selecting the element and editing the name in the *Unit* text field.



Figure 12: The Task Groups Screen

## 2.3.2 Adding Force Elements to a Task Group

The *Bricks* box, on the right hand side of the screen, shows the in-barracks force structure (including the complete unit hierarchy) that was set-up previously in the *Force* tab. These units and subunits can be filtered by their title. To add a subunit to a task group, select the task group, then the subunit to be added, and then click the *<- Add* button. To add a unit and all of its subordinate subunits, select the unit and click *<- Add Tree*.

When defining the make-up of a task group, users may want to know whether a unit has already been assigned to any other task groups. Selecting a unit/subunit from the tree in the *Bricks* box on the right hand side of the screen will cause all task groups to which that particular unit has been allocated to be displayed in the lower-right hand corner of the screen. For example, in Figure 13, *Subunit* 0/3 has been assigned to two task groups; Rotation 2 of Task Group One and Rotation 1 of Task Group Three.

Units and subunits that have been added to the selected task group/sub-group are automatically displayed in the box in the middle of the screen under the *Bricks* tab. Units/subunits can be deleted from a task group by selecting them in this box and then clicking the *Remove* button.

Bricks	^
Test Unit 0	Ξ
🗄 Subunit 0 / 0	-
Subunit 0 / 1	
Subunit 0 / 2	
Subunit 0 / 3	
Subunit 0 / 4	
Subunit 0 / 5	$\sim$
Already assigned to task group(s)	
Group	
Task Group One	
Rotation 2	
Subunit 0 / 3	
Task Group Three	
rotation 1	
Subunit 0 / 3	

Figure 13: Viewing Assignments to Task Groups

## 2.3.3 Viewing personnel and Systems Assigned to a Task Group

Once units have been assigned to a task group it is possible to see listings of all personnel and equipment entitlements (i.e. the personnel and equipment levels that will be deployed during a model run if that task group participates in an operation). Selecting a task force and clicking the *Personnel* and *Systems* tabs respectively will display this information. Personnel are organised into corps, rank and trade streams, and populations are listed as OLOC, MLOC and GRES values. Systems are listed by type and variant.

Because of force migration (via units being linked, see Section 2.2.2), the makeup of units assigned to a task group may change over the time period of the scenario. This may in turn mean that personnel and equipment entitlement levels for the task group vary over time. To view entitlement data at a specific point in the time horizon for the scenario, adjust the slider at the bottom of the screen. The number to the left of the slider shows the scenario month for which populations will be displayed.

#### 2.3.4 Creating Icons

Icons can be created for task groups and rotations in the same way as for units in the *Force* screen. Click the Icon tab to access the icon editor (Figure 10).

# 2.4 Setting-up Operations

In each scenario any number of operations can be defined, and the setup for these takes place in the *Operations* screen (Figure 14). Operations can contain multiple phases, which define different casualty rates. This allows for the setup of an operation involving, for example, an initial high intensity phase followed by a prolonged medium intensity phase. Task groups are dedicated to operations, not specific phases, and can be setup with any number of rotating partners. The setup procedures are discussed in detail below.



Figure 14: The Operations Screen

#### 2.4.1 Creating Operations and Phases

New operations can be created by clicking the *New Operation* button. This will create a new item in the timeline in the upper-left hand corner of the screen, represented by a grey bar. All newly-created operations have one phase by default, and this is displayed as the green bar below the operation.

Clicking on the operation name and selecting the *Info* tab on the right of the screen will bring up some basic options relating to the operation; the name of the operation can be changed and the Maximum Tour of Duty can be altered (from its default value of 12 months). Tour lengths control how long personnel can be deployed before entering reconstitution. If a rotational policy has been defined, the tour length dictates how long the task group will be deployed before it rotates out and is replaced by another task group (rotational setup is discussed further below in Section 2.4.3.). Nominal Reconstitution and Nominal Training periods can also be set; these control the duration of reconstitution and collective training that personnel will undertake between and after deployment periods. Warning Time relates to the amount of notice time that will be available for task groups to start mobilising for an operation. If warning time is less than the particular unit readiness level period then there will be insufficient time to build-up; in this case the software warns the user in the *Review* tab Gantt charts (see Section 2.4.9) but does not impact on the model run. Location refers to the geographical area of operations, and is used in logistics calculations. Clicking the *Location* button brings up a world map in a separate window; right clicking on this map brings up a list of close locations that can then be selected. *Colour* is used in the *Simplified* tab to show different operations in different colours, and can be set by the user by clicking the *Colour* button. There is also a check box, Operation Enabled, which when ticked (the default) means

that an operation will be a part of the scenario and when not the operation will not form part of the scenario during the model run. This functionality allows a user to remove an operation from the scenario without deleting all of the setup details, thus giving the user the flexibility to quickly re-include it.

Additional phases can be added to an operation by selecting the operation and clicking the *New Phase* button. An operation's starting time is set by the starting time of its first phase, and the total length of an operation is determined by the combined length of each of its phases (note that phases cannot overlap and must be sequential). Clicking on a phase and the *Info* tab on the right of the screen brings up some basic options relating to the phase setup; phases can be renamed and their start and end months edited. Phase lengths can also be adjusted by clicking and dragging the ends of the green phase bars in the relevant Gantt chart timeline.

#### 2.4.2 Assigning Task Groups to an Operation

Task groups are assigned to operations and not at the phase level, and assigning a task group is always to an entire operation including every phase in that operation (until rotations are setup). To assign a task group to a operation, select any part of the operation (the operation or any of its phases) from the timeline in the upper-left hand corner of the screen, then click the *Rotations* tab in the lower-left of the screen (if not already selected). A list of task groups, setup previously (see Section 2.3), will be displayed in the box below this tab. Now click the checkbox alongside the task group(s) to be assigned to the operation; noting that multiple task groups can be added to an operation. The *Deployments* timeline to the right of the checkbox should update to show that the task group has been assigned to the phase. This is represented by a bar consisting of a number of colours.<sup>10</sup> The colours represent the following activities:

- Dark blue: the task group is deployed for this period.
- Light blue: the task group has been deployed for longer than the specified maximum tour of duty length for the operation. Consider changing the tour length policy or add another rotation to the deployment schedule.
- Pink: the task group has entered its reconstitution phase after completing a deployment tour.
- Green: the task force is conducting collective training in preparation for a deployment. Note that this only occurs between deployments in an operation.

Multiple task groups can be deployed concurrently during a single operation, and any number of phases can be added to an operation allowing for the setup of highly detailed scenarios. To assist the user in the setup of complex scenarios, especially test scenarios, functionality has been included that allows for task groups to be added to continuing rotations, which is discussed in the next section.

<sup>&</sup>lt;sup>10</sup> Note that the build-up periods towards the first rotation are not displayed here as a task group may be made up of multiple units with different Readiness Notices (see Review tab for display of mobilisation cycles by unit).

#### 2.4.3 Setting-up Rotations

To facilitate the setup of operations, the tool allows the user to create rotations and then automatically rotate them for the duration of the operation using information inputted as part of the deployment policy settings (operation, tour, collective training and reconstitution lengths). The model automatically deploys the first rotation for the duration of the set *Maximum Tour of Duty* (if less than the operation duration), at which time it will move into reconstitution and the second rotation will deploy, and so on until either all rotations have deployed or the operation is complete; if not complete, the first rotation will deploy again after completing collective training and so on until the operation is complete.

To set up a rotation, select the *Rotations Tab* and the relevant operation (or indeed any of its phases). Now click the *[+] Rotation* button which is on the left of the screen (Figure 14). This adds a rotation to the deployment schedule. To add task groups to this new rotation, ensure the correct rotation is selected and then click the checkboxes corresponding to the task groups to be added. The *Deployments* timeline should update to show the effect of the new rotation. Additional rotations can be added in the same way, and rotations can be removed using the *[-] Rotation* button.

Note that if no rotations are added and a unit is setup to deploy for longer than the tour length (in this case the deployment timeline will show a light blue section) the model will still run; it will make use of the task group entitlement data to generate targets for the duration of the deployment. When run, the model will attempt to source personnel from elsewhere in the force (i.e. from units that have a lower priority for personnel) to fill the empty units.

#### 2.4.4 Viewing Personnel and System Levels by Task Group

To view personnel targets for a specific task group, first select the task group to be examined in the *Rotations* tab. Now click the *Personnel Levels* tab. This will display a list of personnel, broken down into stream and rank, allocated to the selected task group in the sub-window below. Clicking on an item in the list will bring up a series of options allowing users to change the *Initial, Mobilising, Non-Mobilising, Ringfenced, Trainees* and *Training Staff* levels for the stream/rank combination in the selected task group. For the personnel selected here, changing these levels will override the levels set earlier in the *Force* screen; presumably the levels of most interest to the user at this point in the setup are the mobilising levels (Figure 15).

System targets can also be viewed in the same way by selecting a task group, and then clicking on the *System Levels* tab. Systems are listed by type and variant, and levels can be set in terms of systems entitlement.

A-SMART																	-	8
Current ORBAT	User Manual O	RBAT Scenario: U	Jser Manual S	cenario														
Scenarios	Setup Re	eview Simplified																
Force Task groups	New Oper	ation 🦳 New Phase	e 😭 Remon	e Export A	B													
Operations	Na			A			In the second second			Info	Casualty	Datum to Dut	Datum	In Canlina	Curtan			
Logistics Setup	B Opeation	One	10 1		11 17	1	10 17			Operatio	Constanty	Present to both			31000			_
Logistics Output	phase	1	_							[v]oper	ation Enabled	Operation Te	OV.					
Logistics Editor	- Operation	Two			COLUMN TWO IS NOT		-			Marries a	Terretor	12	(6)					
Model Rates	phase	à								Maturna	in rour or bucy							
Run The Model										Nominal	Reconstitution	6	3					
Personnel Results										Nominal	Training	6						
Training Results																		
Training Courses										Warning	Time	3	Y					
Training Streams										Location		Unknown loca	stion					
Tools										Colour								
													_					
						-				<u>.</u>		-		1		(		-
	Rotations	Personnel Levels	System L	evels M	ND							-	Personnel	Systems	Task groups	Force		
		Stream		Rank	OLOC		Initial	Lana						Stream	/Rank	REG	Mobilisir	ng Non
	All selected if AMS-WLR - A	tems MS		All ranks WO2	1 0.063	1	ARA	100	- %, in	OLO	C 1		E All Perso Comma	ndo + / PTE		117	117 36.99	117 36.5
	AMS-WLR - V	WLR		W02	0.063	0.	GRES	100	3%, in	MLO	C 0		Mechan	ised Rifleman	- / PTE	3.52	3.52	3.52
	Explosive Do	neer - o Handler -		W02 W02	0.063	0.	Mahilaina						Resource	e Storeman -	/ LCPL	12.59	12.59	12.1
	Field Gunner	y - Detatchment		W02	0.076	0.	494	100	A N In	0.0	c 1		Resource	e Storeman -	/ PTE	8.9	8.9	6.9
	Ground Base	d Air Defence -		W02 W02	0.049	0.		100	-				Rifleman	+ / PTE		50	50	50
	M113 -			WO2	0.125	0.	6805		🐨 %, in	MLU	C 0							
	MBT - Crewn MBT - Crewn	nan Leopard nan SPEC 120MM GN	R	W02 W02	0.063	0.	NonMobilisin	0										
	Mechanised i	Rifleman -		W02	0.008	0.	ARA	100	3 %, in	0.0	C 1							
	Supervisor Er	noineering Services -	Electrician	WO2 WO2	0.063	0.	GRES	100	3% in	MLO	c 0							
	Supervisor Er	ngineering Services -	Plumber	W02	0.063	a.												
							Ringfenced		100									
							ARA	0	3 %, in	OLO	C 0							
							GRES	0	🔤 %, in 🗆	MLO	C 0							
22							-											
ANU.							404	100	A	0.0								
ALC: NO								100			1		٢.					3
ASMART	*	44				>	GRES		V 76, 81	MLO	~ 0	1	8			3		

Figure 15: Viewing Entitlement Levels by Task Group

#### 2.4.5 Viewing Deployment Targets Over the Time Horizon

Targets may vary during a deployment due to rotations of task groups as well as due to force migration. Deployment targets may be viewed at different points over the time horizon by using the slider located at the bottom right of the screen; the slider allows users to step through the months of the scenario. The box in the bottom right of the screen displays the entitlement levels for the task groups of the selected operation that are deployed at the month selected with the slider; note a black line appears on the Gantt charts to provide perspective on the timeline. Four tabs allow users to view deployed *Personnel* and *Systems* targets, as well as deployed units and subunits organised by either *Task Groups* or *Force* structure. A future addition to the tool may be to allow the user to view the units available for deployment (i.e. not committed).

Deployments can also be viewed geographically over the time horizon. Clicking the *Map* tab in the group of tabs on the lower-left of the screen brings up a map of the world (Figure 16); adjusting the time slider allows users to view operational locations at any given month with a box showing the location and the overall numbers of both personnel and major systems. The map can be zoomed using an additional slider in the bottom left hand corner of the screen.



Figure 16: Viewing Deployments Geographically

#### 2.4.6 Defining Operation Settings

In order for the model to accurately assess the affect of casualties and system failures on sustainability, some information needs to be provided for each operation and phase in the scenario. To set these parameters for an operation, select an operation and click the *Casualty* tab in the upper right-hand corner of the screen. The *Operation Type* can be set as either a conventional war or a peacekeeping/peace enforcement type operation. Users can also choose the most appropriate description for the *Terrain* and *Climate* in the area of operations (Figure 17).

In the *Return to Duty* tab, users can create a profile detailing when and how many casualties will return to active duty. Clearly, the higher the return to duty rate, the lower the effective casualty rate will be. Personnel that have become casualties are removed from the model until the point at which they return to duty (if ever). Clicking on an entry in the return to duty list allows users to change the month at which a proportion of casualties will return, and the percentage that will return at that month. For example, in Figure 18 a profile has been set up such that 10% of casualties return to duty almost immediately, with an additional 43% returning after 1 month, 18% after 2 months and 15% after 3 months. Additional return months can be added to the list, as well as removed, and the list can be reset to the default profile. Below the list, the percentage of casualties that will never return to duty is displayed.

A-SMART		💷 🖻 🔀
Current ORBAT:	: 2009-2018 ORBAT Scenario: New Scenario	
Scenarios	Setup Review	
Force	New Operation New Direct Concerning Furget Ac	
Task groups	New Operation wew Phase we remove Export As	
Operations	Name 09 10 11 12 13 14 15 16 17 18 19	Info Casualty Return to Duty Return to Service System
Logistics Setup	B RWG	Operation Type Conventional War
Logistics Output	priase 1	Torrain Eint Date
Logistics Editor		
Model Rates		Climate Dry, Overcast, Temperate
Run The Model		
Personnel Results		
Function Results		
Training Courses		
Training Streams		
Tools		
	Rotations Personnel Levels System Levels Map	Personnel Systems Task groups Force
	[4] Potation	Deployments Stream/Rank REG Mobilising Non-M
		2 13 14 15 16 17 18 19 All Personnel 0 0 0
	MRTF4	
	Rotation 1	
X.I.Z		
Where the second		()
A-SMART		

Figure 17: Setting Casualty Rate Parameters at the Operation Level

Info Casualty Return to Duty	Return to Service System
Add Remove Reset	
At Month % Return	Return To Duty
0 10	Month 2
1 43	Amount 18
3 15	
14 % never return to duty	

Figure 18: Setting Return to Duty Rates

The *Return to Service* tab allows the setting of return rates for equipment. Each system variant can have its own distinct return to service profile. Return rates dictate the proportion of equipment of a specific class that, having failed, can eventually be fully repaired. Clicking on a system class will bring up the currently specified return rates. Rates can be added and removed by clicking the Add and Remove buttons. Clicking on a rate allows the user to edit the month at which a proportion of the equipment will be returned to service, and the percentage that will be returned at that month. Thus a possible system return profile may be that out of all failed equipment of a certain class, 25% returns to duty after 1 month, 30% returns after 2 months and an additional 20% returns after 4 months. This would mean that

25% of failed equipment never returns to service. If no settings are inputted then it is assumed that 100% of failures will not return to service.

Selecting the *System* tab brings up a list of all equipment types and variants allocated to the selected operation. Selecting an equipment type/variant from the list brings up text fields for *Availability Rate (%), Time Between Deep Maintenance (month)* and *Quarantine Period (month)* and the user can enter settings for these input parameters that are specific to the operation and equipment type (Figure 19).<sup>11</sup> For example, a vehicle deployed on a high intensity operation may require more frequent deep maintenance schedules than on a low intensity operation. The *Availability Rate* relates to percentage of equipment that is available at any time due to unscheduled or light grade maintenance. *Time Between Deep Maintenance* relates to the frequency of scheduled maintenance. *Quarantine Period* is the time that equipment is unavailable at the completion of an operation; it is not applied at movements into scheduled maintenance.



Figure 19: Setting Equipment Operation Parameters

#### 2.4.7 Setting Phase Parameters

Selecting a phase reveals a new set of tabs in the upper right-hand section of the screen. The *Casualty* tab allows users to set the rates for battle and non-battle casualties for the selected phase (Figure 20). The options presented for setting battle casualty rates vary depending on whether the operation to which the selected phase belongs is a conventional war or a peacekeeping/peace enforcement type operation. In the case of the former, users can select an appropriate *Sophistication* factor, which represents how much more sophisticated Australian forces are compared to opposition forces. The *Combat Power* factor relative to the opposition should also be set. These two factors, along with some of the settings already defined for the

<sup>&</sup>lt;sup>11</sup> Note that default data for current Army equipment has not yet been sourced; however, default data will be sourced from the MIMS Maintenance Module or fleet managers, as necessary.

operation, are used to calculate a battle casualty rate for the phase. The calculated rate is displayed in a non-editable box below these two options. Casualty rates are expressed as a percentage of the force that will become casualties per day during the phase.

Battle Casualty Rate					
Use Custom Rate 0 % per day					
Conventional War					
Sophistication To Opposition Minor Advantage					
Combat Power To Opposition Even Match					
Calculated Rate 0.036					
Non Battle Casualty Rate					
Historical Rate Slipper-0.034% per day 🗸					

Figure 20: Setting Casualty Rates for a Conventional War Type Operation

For peacekeeping/peace enforcement type operations, users can select a historical rate from a list of past operations. For either operation type, users can also elect to set their own custom daily battle casualty rate. To do this, click the *Use Custom Rate* checkbox and enter a value in the corresponding text box.

Non-battle casualty rates are set by reference to historical rates from past operations. Again, users have the option to set their own custom rate if they prefer.

The model does not apply the battle casualty rate evenly across all deployed personnel. Historically, it has been demonstrated that certain types of units suffer a disproportionately high percentage of the total casualty rate. In the *Casualty Distribution* tab, users can apportion the total battle casualty rate between force types; force types are set for units in the *Force* page (Section 2.2.1). Clicking on a specific force type will bring up the option to override the default apportionment settings. The *Clear All Overrides* button will restore the default profile.

The *Failures* tab allows users to set failure rates for equipment. By default the equipment failure rate is calculated based on the battle casualty rate for personnel by applying a multiplication factor (historically there has been shown to be a strong correlation between personnel and equipment casualty rates). For example, tank failure rates have shown to be up to 6 times the relevant personnel casualty rates. Alternatively, the user also has the option to enter a custom failure rate. Again, all default values can be restored using the *Clear All Overrides* button.

#### 2.4.8 Reviewing Unit Allocations to Deployments

The *Review* tab (Figure 21) displays deployments for the entire in-barracks force structure tree and across the whole of the scenario timeline in the form of a Gantt chart. The colours used for

the operational phases in the Gantt charts are the same as those used in the deployment and rotation displays in the *Setup* tab (dark blue for deployed, pink for reconstitution, green for collective training, light blue for over-deployment, red for over-allocation<sup>12</sup>), with the addition of an orange build-up phase (the length of which is determined by the unit readiness notice). When a unit is selected, there is a display in a sub-window at the top of the screen which indicates the operation(s) to which a unit has been assigned, as well as the timelines; solid colour represents deployed, light colour build-up, reconstitution or collective training, using the operation colour selected in the *Setup* tab. The *Review* display is useful when the user is exploring concurrency issues, as well as the availability of units for allocation to further operations.

The *Show Deployed* button will expand the list of units to show all deployed units. The *Calculate Traffic Lights* button when clicked will calculate and display a traffic light (either green or red) next to any unit that is deployed at any point; a red signal highlights that a unit or one (or more) of its subordinate units, are assigned to concurrent deployments and/or are deployed for longer than the maximum tour of duty period (set as part of the operation setup).

A-SMART		
Current Scenari	o: User Manual Scenario	
Scenarios	Setup Review Simplified	
Force	Show Deployed Calcualate Traffic Lights	
Task groups	Operation	Deployments
Operations	Opeartion One	
Logistics Output	Operation Two	
Logistics Editor	Bricks	Deployments
Model Rates	Unknown formation	
Run The Model	Ounit0	
Personnel Results	OTest Unit 1	
Training Results	OTest Unit 2	
System Results	OTest Unit 3     OTest Unit 4	
Training Courses	OTest Unit 5	
Training Streams	OTest Unit 6	
Tools	OTest Unit 7     OTest Unit 8	
	O Test Unit 9	
510050		
AND HAVE		
SANE		
A-SMART		

Figure 21: The Deployment Review tab

## 2.5 Setting Personnel Recruitment and Separation Rates

Two of the key input parameters for the personnel module, are recruitment and separation rates; by selecting the *Model Rates* page and *Personnel Rates* tab, the user can view and, if desired, amend these rates. Recruitment rates for ARA and GRES refer to the absolute number

<sup>&</sup>lt;sup>12</sup> Note that *over-deployment* refers to a unit (or task group) being setup to deploy for longer than the policy setting for maximum tour length whereas *over-allocation* refers to a unit (or task group) being setup to deploy on multiple operations at the same time.
of personnel recruited per year for the selected group (note that, for the model run, this rate is transferred into a monthly rate by simply dividing by 12). In contrast, the non-deployed separation rates are set as a proportional loss of the total personnel level per year; for example, where a setting of 0.10 corresponds to a 10% loss per annum (note that again this rate is transferred into a monthly rate for the model run by dividing by 12).

At the top of the screen is a drop down box allowing users to switch between viewing ARA recruitment figures, GRES<sup>13</sup> recruitment, and Non-deployed separation rates. At the bottom of the screen is a slider that can be adjusted to show rates for any month in the time horizon, since rates can vary across the model run time (see below).

Rates can be set for *Officers* and *Other Ranks* by selecting the appropriate tab. Base rates can be edited independently for every combination of corps, trade stream and rank by double clicking the appropriate figure in the table to the left of the sliders. For example, in Figure 22, the recruitment rate for RACT Cargo Specialist Privates has been set to 120 per year.

Current Scenari	o: User Man	ual Scenario																		
Scenarios	Personnel	Rates System	Rates																	
Force	Import	Export																		
Task groups	Target Rat	e: Recruitr	nent ARA (persons per	year)																v
Looistics Setup	Other R	officant	Bark Adjustment	Vanin	Datas	1														
Logistics Output	Come	Trade	Cream	Dermit	OTE	100		CCT	14400	14104		Data Adjustment			Darrad	OTE	100	<b>CD</b>	0.07	( wind)
Logistics Editor	- Corps	trace	Seream.	Rectur	PIE	un	un	501	muz	0		roade wajusament			ROCTUR	PIE	a	- Cri	561	H
Model Rates	RAERE	Alicran Life Support	n Huter -								1			100%	č					
Run The Model	RAEME	Aircraft Structural I	itter -	0	0	0	0	0	0	0	-			100%	0	0	0	0	0	0
Personnel Results	RAEME	Technician Artificer	Aircraft -	0	0	0	0	0	0	0	-			100%	0	0	0	0	0	0
System Regults	RAEME	Technician Artificer	Avionics -	0	0	0	0	0	0	0	-		_	100%	0	0	0	0	0	0
Training Courses	RACT	Air Dispatcher		0	0	0	0	0	0	0	_			100%	0	0	0	0	0	0
Training Streams	RACT	Cargo Specialist -		0	120	0	0	0	0	•			_	85%	0	102	0	0	0	0
Tools	RACT	Marine Specialist -		0	0	0	0	0	0	0	-		-	100%	0	0	0	0	0	0
	RACT	Operator Movemer	its -	0	0	0	0	0	0	0				100%	0	0	0	0	0	0
	RACT	Supervisor Transpo	at - Driver	0	0	0	0	0	0	0			_	100%	0	0	0	0	0	0
	RACT	Supervisor Transpo	ert - Operator Special	0	0	0	0	0	0	0			_	100%	0	0	0	0	0	0
	AAAVN	Aircrewman -		0	0	0	0	0	0	0				100%	0	0	0	0	0	0
	AAAVN	Groundcrewman -	Aircraft Support	0	0	0	0	0	0	0				100%	0	0	0	0	0	0
	AAAVN	Groundcrewman -	Mission Support	0	0	0	0	0	0	0		-		100%	0	0	0	0	0	0
	AACC	Cook -		0	0	0	0	0	0	0				100%	0	0	0	٥	0	0
	AACC	Steward -		0	0	0	0	0	0	0			_	100%	0	0	0	0	D	0
	RAAPC	Clerk Finance -		0	0	0	0	0	0	o —				100%	0	0	0	0	0	0
	AUSTIN	Intelligence Operat	ions -	0	0	0	0	0	0	o —				100%	0	0	0	0	0	0
	AUSTIN	Linguist Intelligence	ė	0	0	8	0	0	0	0				100%	0	0	0	0	0	0
	RAINE	Operations Clerk -		0	0	0	0	0	0	0				100%	0	0	0	0	0	0
	RAINE	Resource Storeman		0	0	0	0	0	0	0				100%	0	0	0	0	0	0
	PATHE	Machanizari Diffam			0	0	0							100%				0		
	DATAF	Different o		×	0	0	č	0		0	H			1004	0		0	0	0	ě.
and the second	SCHUMP.	sondriden s			9	9		3	9		H			100%		J	9	9.	a	~
C. AL	<																			>

Figure 22: Editing Recruitment Rates

#### 2.5.1 Rate Adjustment

The base rates set for each corps/trade stream/rank combination can be adjusted in a number of ways (to facilitate 'what-if' analysis). The *Rate Adjustment* slider to the right of the base rate table in the *Officers* and *Other Ranks* tabs allows adjustment of rates of all ranks for a specific corps and trade stream. For example, in Figure 22 above, the RACT Cargo Specialist rank/trade stream combination has been scaled to 85% of the base rate. The results of this modification are shown in the table to the right of the sliders (for Privates, 0.85 x 120 = 102).

<sup>&</sup>lt;sup>13</sup> Note that currently only regular personnel are modelled and the GRES data is not used during the model run at this point.

Base rates can also be varied by rank in the *Rank Adjustment* tab. This may be useful if, for example, a user wants to explore the effect on sustainability of a proportional increase in recruitment into a specific rank across the entire land force.

Finally, recruitment and separation rates can also be varied over time. The *Varying Rates* tab consists of a tree on the left where trade stream/rank combinations can be selected (or, indeed, specific ranks for all corps/trade streams), as well as two graphs. The graphs adjust the base rates previously discussed for the selected item as a percentage of the base rate. Once a particular trade stream/rank has been selected, users can plot points on the top graph to create a cyclical 12 month pattern. This pattern will be applied for every year of the scenario, and is used to simulate seasonal variations in recruitment and separation rates. For example, Figure 23 shows a 12 month cyclic varying rate for the RACT Cargo Specialist/Private trade stream/rank combination. The graph shows recruitment peaking in month 2 and month 8 every year.



Figure 23: Varying Rates Over Time as a Percentage (Y-Axis) of the Base Rates

The bottom graph on the *Varying Rates* screen allows users to set trends that span the entire scenario. This may be useful to simulate more strategic factors such as the effect of particular economic conditions, expected demographic changes, or operational tempo on recruitment rates. If multiple adjustments are made to the base rate, the applied rate for a given month is calculated by multiplying all of the adjustments set at that time step. So the formula for determining the final applied rate can be expressed as:

Applied rate = [base rate] x [corps/trade stream adjustment] x [rank adjustment] x [cyclic 12-month adjustment] x [long-term adjustment]

The applied rate, taking all of these factors into account, is always displayed to the right of the sliders in the *Officers* and *Other Ranks* tabs. Note that if varying rates have been set the applied rate will change during the model run; there is a slider bar at the bottom of the page which

allows the user to select any month during the model run and the corresponding rates for the selected month will be displayed.

# 2.6 Setting Major System Rates

The layout of non-deployed *System Rates* is similar to that for *Personnel Rates*. The *Normal Rates* represent the base rates or normal settings/configurations, whereas the *Varying Rates* allow the user to simulate changes due to unusual events (for example, a change in the deep maintenance capacity level due to contractor availability or *Time Between Deep Maintenance* is shorter as systems age). Note that the analogous rates for deployed equipment were discussed earlier in Section 2.4 and are set under the *Operations* tab.<sup>14</sup>

### 2.6.1 Base Rates

For non-deployed equipment, inputs are organised into two groups; *System/Class* and *Readiness Level. System/Class* inputs are set for the selected system (Figure 24) and can be set at any level of the tree (for example, across all *A VEHICLES*, all *IMV – Bushmaster* or specifically for the *LIGHT ARMD VEH INF MOB COMMAND* variant) but cannot be further set by specific readiness level. *Readiness Level* inputs are set for each selected system (again at any level in the tree) and for each specific readiness level. Consequently, *System/Class* inputs influence all of the systems of the relevant type across the entire fleet, whereas *Readiness Level* inputs are applied to only those systems that are allocated to units which have readiness notice set at the corresponding level.<sup>15</sup> The *System/Class* and *Readiness Level* inputs are listed in the Table 1 below; a description of each input is provided.

<sup>&</sup>lt;sup>14</sup> By selecting an operation and the *System* tab, the inputs *Availability Rate, Time Between Deep Maintenance* and *Quarantine Period*, can all be set for the particular operation or by selecting an operational phase and the *Failures* tab, casualty rates for equipment can be set, for the particular phase. <sup>15</sup> Note that the *Readiness Level* inputs are for non-deployed equipment only and are applied similarly to the rates for deployed equipment (discussed above in Section 2.4); they are inputted in a different part of the interface to assist in the usability of the software.

Current Scenario	c Default								
Scenarios	Personnel Rates System Rates								
Force	Rose Robert Life for Robert								
Task croups	base kaces varying kates								
Operations	System/Class	Deep Maintenance Capacity	Deep Maintenance Period	Procurement	Repair Stock	Attrition Stock	Loan Sto	Readiness Level	
I anisting Cature	B A VEHICLES	0	0	0	0	0	0	On Call	
Cogeneration of the second	MBT	0	0	0	0	0	0	High	
Logence Output	# AVLD	0	0	0	0	0	0	Medium	
Logistics Editor	# M113	9	9	0	P	0	0		
Model Rates	I ACV	0	0	0	0	0	0		
Run The Model	I ARVL	0	0	0	0	0	0		
Personnel Results	I TLC	0	0	0	0	0	0		
Training Results	E APCA	0	0	0	0	0	0		
System Results	a ASLAV	0	0	0	0	0	0		
Tanining Country	WHLD ARMD VEH RECON T	1 W/EQUI 1000	3	7	0	0	0		
Training Courses	WHLD ARMO VEH MULTI RO	XE T-11 0	0	0	0	0	0		
I reining Streems	WHEELED ARMD VEHICLE,	RECOVERY 0	0	0	0	0	0		
Tools	WHEELED ARMD VEHICLE, I	SUPVETIA 0	0	0	0	0	0		
	# AIRCRAFT	0	0	õ	õ	0	õ		
	B VEHICLES	0	0	0	0	0	0		
	C VEHICLES	0	0	0	0	0	0		
	COMMUNICATIONS	0	0	0	0	0	0		
	CUNS	0	0	0	0	0	ě.		
	@ PERS WEAPONS	0	0	0	0	0	0		
	REFRIGERATION	0	0	0	0	0	0		
	TRAILERS	0	0	0	0	0	0		
	UNIT STORES	0	0	0	0	0	0		
	WATER PURIFICATION	0	0	0	0	0	0	Availability Rate (%)	90
	@ WATERCRAFT	0	0	0	0	0	0	Time Extensor Deep Maintenance (month)	
								time occurrent book reactioners (news)	24
	1						- 18	Annual Loss Rate (%)	0
	120 00000 0000 0000 0000						10		
	Deep Maintenance Capacity (number)	1000					Ŷ		
	Deep Maintenance Period (month)	3					0		
	Annual Procurement (number)	7					-		
	Repair Stock (number)	0					1		
and they	Attrition Stock (number)	0					1		
All Car		0					- A		
SULFACE AND	Loan Stock (number)						1		

Figure 24: System Base Rates

Table 1:	Description	of Major	<b>Systems</b>	Rates
		. /		

Input Name	Input Type	Description
Deep Maintenance Capacity	System/Class	Maximum number of systems that can be serviced at any point in time.
Deep Maintenance Period	System/Class	Number of months that systems are not available due to deep maintenance.
Annual Procurement	System/Class	Number of new systems to be procured annually.
Repair Stock	System/Class	One-off procurement to compensate for preventive maintenance.
Attrition Stock	System/Class	One-off procurement to compensate for annual loss.
Loan Stock	System/Class	One-off procurement to account for training activities.
Availability Rate	Readiness Level	Percentage of systems populations available at any time.
Time Between Deep Maintenance	Readiness Level	Number of months that systems will be available before requiring deep maintenance.
Annual Loss Rate	Readiness Level	Percentage of systems lost (not recoverable or disposed of) annually.

# 2.6.2 Varying Rates

The base rates for procurement, loss rate, deep maintenance period, deep maintenance capacity, time between deep maintenance and availability rate can be adjusted by making use the varying rates functionality (Figure 25). Varying rates are introduced to simulate impacts

on system model due to collective training, operational tempo or unexpected events, and can be inputted either seasonally or over the duration of the scenario (in a similar fashion to personnel rates, see Section 2.5.1.).



Figure 25: System Varying Rates

# 2.7 Viewing and Editing Training Information

Training inputs can play a key part in influencing force sustainability in a scenario. If there are insufficient resources (which currently equates to instructors only in our modelling) to conduct required training courses or personnel are not available to undertake training, a situation may be reached where few, if any, personnel at a particular rank are promotable. This creates a bottleneck and may have flow-on effects to higher ranks, causing a failure to meet population targets.

The A-SMART tool has been released with comprehensive individual training input data included. Most users should therefore not need to edit any of the training information that has been set up. Nevertheless, users may wish to view training data to assist in analysing possible causes of a failure to sustain personnel levels or to make amendments due to, for example, updated training data becoming available (including adding in new courses).

# 2.7.1 The Training Courses Screen

Users can view and edit the input data for training courses by clicking the *Training Courses* tab (Figure 26). On the left side of the screen is a list showing all currently defined courses and modules, organised by corps. Each course can have one or more modules, each with different lengths and requiring different resources.

Selecting a course displays a number of course options on the right hand side of the screen; the course *Title* can be edited, as well as its *Stream, ECN* and *School* (Figure 26). Currently,

these inputs/values are for information purposes only and do not affect how the training model is run. One important parameter is the *Students per course*, which, in conjunction with other module settings discussed below, will affect the level of resources that are required per student.

Come / Corner / Madula	2. INON SCENED		
Scenarios Corps / Course / Module	- Course Details		
Force Filter	Expand New Module	Remove Course	
Task groups Corps / Cour	rse / Module		
Operations AABC		ROBC - Mod 1 Tactics	
Logistics Setup AALC	Stream	MBT	
constice Output	ECN	POSC .	
H Logistics		1.0000	
Model Dates	Courses per year	1	
E AACC	Studends per course	é [12	
H RAA			
Ensonnel Results III RAVG	5chool	School of Armour	
raining Results	Contraction of the second s		
System Results 🛞 Crewman Comd MBT -	Tactics module		
raining Courses III Crewman Driver ASLAV	V - Corps Training		
raining Streams	(Corps Training Phase)	this course	
Tools Crewman Specialist 25	imm Ger	500 Cha	an Overani
A LAPV Crewman Comm	ander - Mod 3		THE LOSS NUMBER
H MIAI 120MM Gunner	8	Stream	
H ROBC ASALY Mod 3 G H ROBC ASALY Mod 4 C H ROBC MIT - Mod 2 G H ROBC MIT - Mod 3 G H ROBC MIT - Mod 3 G H ROBC MIT - Mod 3 G H ROBC MIT - Mod 1 H ROBC MIT - M	umeny mimunications interpandi Services interpandi Services Mod Mod Commonder ASUAY - C Commonder ASUAY - C Commonder ASUAY - C Common Module - Services Module - Services Module - Services - Services - Services - Services - Services - Services - Servi		

Figure 26: The Training Courses Screen

Selecting a module reveals more settings (each course must have at least one module, Figure 27). Again, the *Title* of the module can be changed. The *Length* of the course affects the amount of time for which students and instructors must be available to complete the course. The model assumes that there are 20 training days per month (allowing for weekends and leave). The *Instructors to Students* ratio is not currently used as a model input. Each training module requires a specific level of resources. The Personnel, Systems, Ammunition, Facilities, POL<sup>16</sup>, and Rations tabs allow users to view and edit the amount of these resources a module requires. Only the information in the *Personnel* tab is currently used during the model run. By clicking the *Personnel* tab, users can choose to add required personnel by *Issue Type* by clicking the <- Add button; Issue Type refers to the unit by which the resource is allocated (Course, Module, Staff, Trainee, Crew, Section, Spare or Enemy Crew) and is usually set to Course. Selecting the newly added entry allows users to specify personnel as instructors or support staff<sup>17</sup> (by ticking the check box for instructor or not for support staff), the number required (clicking the up/down arrows next to the Required box allows the number of personnel to be increased/decreased), the corps (by selecting the desired corps from the list in the bottom right of the screen and clicking Add ECN), rank (initially the rank of Recruit will be added, this can be altered by clicking on the word *Recruit* to reveal a drop-down menu, from which the desired rank can be selected from the list) and ECN(s) from which they can be drawn (by clicking under *ECN* a box is revealed into which the desired *ECN* can be typed). If the

<sup>&</sup>lt;sup>16</sup> Note that POL is an acronym for Petrol, Oil and Lubricants.

<sup>&</sup>lt;sup>17</sup> Note that in the display, if the instructor field is listed as *True* then this entry represents an instructor and if *False* then it represents support staff.

instructor/support staff position can be filled by personnel from more than one ECN then multiple ECNs can be added by following the same procedure (i.e. with the relevant *Personnel* entry selected, choose the desired corps, click *Add ECN* etc.). Other resource types are added in a similar fashion, although this information is not currently used in the model run.



Figure 27: The Training Module Courses Screen

#### 2.7.2 The Training Streams screen

The notion of a training career profile is centred around the idea of a linear progression through a series of training courses, On-the-Job-Training (OJT) and Time-In-Rank (TIR) experience, that personnel must satisfy in order to qualify for promotion. Each trade specialisation (in each corps) in the army is represented by a generic training stream; these profiles represent preferred qualifications and experience and are based upon information obtained from relevant Training Management Package (TMP) and Manual of Army Employment (MAE) documents. These streams describe the courses that must be completed at each rank in order to progress to the next, the minimum TIR in order to undertake a course and the minimum TIR to qualify for promotion. The *Training Streams* screen allows users to view, add and alter training profiles for all defined streams.

On the left side of the screen is a tree that lists training streams, organised by corps. Selecting a stream will display an overview of the career profile defined for that particular stream. *Career Time* refers to the minimum time that personnel can take to reach a given rank. The relevant ECNs and courses that need to be completed are also displayed. For example, in Figure 28, the information for the *RAAC: ASLAV – Crewman* stream is displayed. From it we can see that it takes a minimum of 3 years and 6 months for a new recruit to progress to Lance Corporal, and that at this rank there is one course that needs to be undertaken and a further 12 months TIR before promotion can occur.



Figure 28: Viewing Training Stream Information

Selecting a specific rank within a training stream allows editing of the requirements for that rank. On the right of the screen a list of all defined courses allows users to add courses to the requirements for the selected rank/stream combination. Current course requirements are listed in the box in the centre of the screen, along with a message advising on the minimum time in rank value, if any, which must be satisfied in order to qualify for promotion. Clicking on the course names or the message, *Available for promotion after x months Time-In-Rank*, allows the user to amend the periods required to undertake the particular course (i.e. effectively the period of OJT required before personnel can be panelled on a course) or the TIR period required to qualify for promotion, respectively, by entering the period in the box at the bottom of the screen (Figure 29).

Courses also have a *Weighting* associated with them. If all personnel must conduct the particular course then it is set to 1; however, if there is an option between more than one course (as in an elective course) then the user should add all of the relevant courses, multi-select the courses and then click on *Merge Courses*. The fraction of personnel expected to undertake the different course options should then be entered in the *Weighting* box at the bottom of the screen for each course; note that the sum of weightings across all of the merged courses should equal one (if not the tool highlights them in red). If personnel must take more than one of a list of courses then the list of courses must be added and merged multiple times.

The order of courses can be changed by selecting a course and clicking the *Move Up* or *Move Down* button. Users can also select a course and click the *Show Course* button to be taken to the relevant entry in the *Training Courses* screen.



Figure 29: Amending On-the-Job-Training and Time-In-Rank Periods

# 3. Supplies & Strategic Lift Module

In addition to the personnel and major systems modules, A-SMART can assist in forecasting expected supply levels and strategic lift requirements. The supplies and strategic lift module is based on the Joint Operational Logistics Tool Suite (JOLTS) project, and given appropriate inputs, can forecast the following expected logistics outputs:

- Daily weight requirements for each class of supply (food/water, fuel, ammunition etc) and operational phase (average over phase or option to view by single month)
- Total stocks of each class of supply to be held in the area of operations for each operational phase
- A timeline of strategic lift utilisation for each available platform and operational phase
- Plots displaying logistics requirements over time by class of supply and operational phase
- Plots displaying levels of major systems that require transportation by operational phase
- Plots showing unused air lift and sea lift capacity

# 3.1 The Logistics Setup Screen

The *Logistics Setup* screen allows the user to enter information that will determine the logistics requirements for each class of supply. Nine classes are covered by the tool, and these correspond to the classes of supply defined in Army doctrine:

- Class 1 consists of subsistence items, including foodstuffs, water, health and welfare items
- Class 2 consists of general stores including clothing, individual equipment, tentage, tool sets and kits, hand tools, stationery, administrative and housekeeping items

- Class 3 consists of POL
- Class 4 consists of construction items including construction materials and all fortifications and barrier materials but excluding explosive devices
- Class 5 consists of ammunition
- Class 6 consists of personal demand items including canteen supplies and non-scaled military items
- Class 7 consists of principal items which are a final combination of products ready for their intended use including vehicles, weapons and major equipment assemblies such as engines, transmissions and radios
- Class 8 consists of medical stores (including medical repair parts) and dental stores
- Class 9 consists of repair parts

When the *Logistics Setup* screen <u>and</u> an operational *Phase* are selected, the *Operation Settings* tab will be displayed. In this tab, the user can set factors that will affect all classes of supply for the selected operational phase, including basic assumptions. For example, the weight of 1000 litres of fuel can be set in this screen, and this will subsequently affect the weight profile of all fuels to be transported.

In addition to the *Operation Settings* tab, there are tabs for class 1 inputs, class 3 inputs and a single tab for the inputs of all other classes (2 and 4-9). In the *Class 1* tab, users can specify the proportion of personnel consuming different types of rations, as well as inputs affecting water usage (Figure 30).

In the *Class 3* tab, users can set a number of values that affect the combat factor. The combat factor is a multiplier that influences base fuel usage rates for all vehicles, and is based on conditions such as terrain and climate in the area of operations. On the right of the screen, lists of vehicles and aviation platforms are displayed. The *Average number deployed* value for each vehicle/aircraft is derived from the entitlement level of major systems assigned to the relevant task forces participating in the selected phase (this information is sourced from the unit SED data which underpins the force structure, discussed earlier in Section 2.2). As the number of vehicles/aircraft deployed may change during a phase, due to the rotation of units, these levels are an average across the timeline of the particular phase selected. The *Operation time per day* is based on the *Average Use (Hours/Day)* value for the vehicle in the *Logistics Editor* screen (see Section 3.3 below); however, the value can be edited in the *Class 3* tab by double clicking the number for the desired vehicle.

Of the remaining classes, only class 4 and class 5 have specific factors used to forecast usage levels. For class 4, one of the major determining factors is the length of the phase. Any phase that is setup to continue longer than 6 months requires a much higher per-person value for construction items; i.e. Company Defence Packs, which represent an additional logistics requirement during the initial setup stage of the deployment. Class 5 requirements vary depending on the type of phase (advance/deliberate attack/quick attack/defence/withdrawal/covering/protracted/inactive). Each system deployed has a rate of ammunition usage per day for each applicable phase type.

Opeartion One	Operation Settings Class 1 Class 3 Class 2/4/5/6/7/8/9	DoS Levels	Route Pla	nner Strate	egic Lift T	ransport Availability		
i≊ Opeation One These 1 phase 2 Operation Two phase a	Operation Settings         Class 1         Class 3         Class 3	005 Levels           60           30           70           9           10           2           2           0	Koute Pla	Iner Strate Water requiren The dally water Include medic. Is there a loca What is the to	gic Lift T ments er requiremental water use? al water sourc: tal daily produ	ransport Availability t per person in litres? a available for use? Jucation capacity of the loc	al water source?	15 Litres No Ves V 0 Litres
	Class 1 Assumptions A fresh ration A single 1-man Combat Ration Pack A single 5-man Combat Ration Pack	weighs 1 weighs 1	3.5 Kg 1.35 Kg 8 Kg	Results Number of fr Weight of fr Number of 1 Weight of 1- Number of 5 Weight of 5- Total weight Daily water of	resh rations esh rations r l-man CRPs r -man CRPs r -man CRPs r t of class 1 fo requirement	required daily. equired daily. equired daily. equired daily. equired daily. equired daily. ood requirements. to be brought in for th	ie force.	1528 5345.3 Kilogram 284 382.4 Kilogram 133 1057.4 Kilogram 6785.1 Kilogram 34710 Litres

Figure 30: Specifying Class 1 Inputs

For the remaining classes of supply (6-9), users can select from predefined rates for each class, expressed as kg/person/day.

Within the logistics setup tab, the user can copy settings between phases (including phases from other scenarios) by clicking the *Copy From*... button (Figure 31), located at the bottom left of the screen below the phase list (note that if current values already exist they will be written over and not recoverable). Only values for the Class tab that is currently selected are copied from the other phase; for example, if Class 1 is selected only its values will be copied over, leaving the values of all other classes unchanged.



Figure 31: Logistics Copy From button

# 3.1.1 Setting Days of Supply levels

The *DoS Levels* tab allows users to set the level of reserve stores that will be held in the area of operations (Figure 32). The practical effect of altering these levels will be to change the amounts of stores that will need to be delivered at the beginning of each phase. Reserves can be set at the *Formation Level, Unit Level,* as well as by separate *Commander's Reserves* category. The combination of all of the levels (Unit, Formation and Commander's reserve), for any given class, provides the total number of days of supply required at the start of each phase. The number of days is multiplied by the forecast daily requirement, for the particular supply class, to give the overall forecasted initial requirement by weight or volume.



Figure 32: Setting Days of Supply Levels

# 3.1.2 Route Planner

The route planner allows users to specify air and sea routes from the staging area to the area of operations (Figure 33). The route type (*Sea* or *Air*) to be edited is selected from the *Route Type* radio buttons on the right hand side of the screen. A map of the world allows users to select route waypoints. Right clicking a point on the map will bring up a list of nearby ports (for a sea route) or airports (for an air route). Left clicking an item in the list will add it to the route. Points on the route can be deleted by selecting the location in the *Route Path* box on the right of the screen and clicking on the *Delete Selected* button at the tope right of the screen.

To make selection of waypoints easier, the map can be zoomed by using the slider on the bottom of the screen, and panned by left clicking and dragging any point on the map. Locations can also be sorted and selected by name and country using the *Search* button at the top of the screen.



Figure 33: Setting an Air Route

# 3.1.3 Setting Strategic Lift Options

To forecast if/how strategic lift requirements can/will be met, the model uses information on available platforms, as well as how lift requirements will be divided between air and sea platforms. The user can input this information by first selecting the Strategic Lift tab (Figure 34). One of the key inputs is the percentage of personnel that will be transported by air; this will affect the distribution of passenger transport requirements amongst air and sea assets. Inputs specific to air transport include loading/unloading time, percentage of aircraft continuously serviceable, lead time for chartering civilian aircraft and whether to use only military aircraft, only civilian aircraft or a combination of the two. The Bulk Out Factor refers to a percentage limit (of the aircraft weight capacity rating) imposed to simulate space constraints on air transport. Capacity ratings for aircraft are expressed in terms of weight whereas, in reality, when loaded an aircraft usually runs out of physical storage space before the weight limit is achieved; for example, a 40% bulk out factor indicates that the aircraft will only be able to carry 60% of its rated weight capacity. On the top right hand side of the screen is a panel displaying the current air lift requirements based on the logistics inputs specified so far; including trip distance, number of personnel to be transported, initial and sustainment weights, and the maximum number of days allowable to transport initial supplies and major systems.

Inputs for sea lift include load/unload time, lead time for chartering vessels and fleet type used (military, civilian or both). Similarly to the air lift, sea lift information is displayed on the right hand side of the screen (distances, personnel to be moved, amount and maximum time

allowable to transport initial supplies and major systems). Note that sea lift capacity is calculated in terms of area and not weight.

peration settings	Class 1	Class 3	Class 2/4/5	5/6/7/8/9	DoS Levels	Route Planner	Strategic Lift	Transpo	ort Availability	
Air lift inputs					Air li <del>0</del>	information				
Air int inputs Aircraft will transpor	t what % of i	nersonnel?		25	One w	av distance per trip	is	1411	.0.6 Km / 7619	.1 NM
How long is being a	located to loa	ading, unload	ing and	1	Numb	er of Pax to be mov	ed?		4	7 PAX
maintenance for eac	h round trip?	ang, anoda	ing and	1	Hours Weigh	t of initial deployme	ent to be moved by	/ air.	72191	5 Kg
What percentage of serviceable?	aircraft will b	e continuous	ly	85 ,	% Daily water.	veight of sustainme	nt stocks, this incl	udes	675	7 Kg
What is the lead tim	e for charteri	ng civil aircra	ft?	7 [	Days Deploy	ment time.			1	5 Day
What is the aircraft	Bulk Out Fact	or?		40 9	%					
Fleet type to use			Militan	y and Civilia	in 🗸					
Sea lift inputs					Sea lif	t information	ic		0.6m / 0	0.00
Sea lift inputs How long will it take	to load the s	hip?		20 1	Hours Sea life One w Numb	t information ay distance per trip ar of Pax to be mov	is ed7		0 Km / 0	0 NN 5 PA)
Sea lift inputs How long will it take How long will it take	to load the s	:hip? e ship?		20 1	Hours Sea lif One w Hours Numb	t information ay distance per trip er of Pax to be mov t of initial deployme	is ed? nt to be moved by	/ sea.	0 Km / 0 11 321	0 NN 5 PAX 3 m^
Sea lift inputs How long will it take How long will it take What is the lead tim	to load the s to unload th e for charteri	ship? e ship? ng civil vesse	ls?	20   24   7	Hours Sea lif One w Hours Numb Days Deploy	t information ay distance per trip er of Pax to be mov of initial deployme ment time.	is ed? ant to be moved by	ý sea.	0 Km / 0 14 32/ 11	0 NN 5 PA) 3 m^ 5 Day

*Figure 34:* Strategic Lift

# 3.1.4 Aircraft and Ship Transport Availability

The next step in setting up air lift inputs is to specify when and the number of lift platforms that are available. When the *Transport Availability* tab is selected, a list displays the capacity limits and other information for various aircraft and ship platforms (Figure 35). Note that the number of aircraft/ships specified as available at this stage is a global setting that affects the whole scenario; that is, these settings will be applied across all operations and for this reason the operation/phase selection list is hidden for the *Transport Availability*. The most important consequence of this is that if two or more operational phases are set to run concurrently, they will share available lift assets.

Current Scenari	o: Logistics Test										
Scenarios	Operation Settings	Class 1	Class 3	Class 2/4/5	/6/7/8/9	DoS Levels	Route P	lanner	Strategic Lift	Transport Availability	
Force									Denlo	ment Timeline	
Task groups	Name	Ca	apacity (Kg)	Capacity (Pax)	Speed (Knot	s) Fleet type	Available	hillinghund	10 10 11	International 13 million of 14 million	15
Operations	747 -400F (AL-1A)	1	13000	0	467	Civilian					
ogistics Setup							15				
onistics Output							10				
ogistics Editor	420084		1000	260	460	Civilian	5				
ogistics cultor	A30004	1	1000	209	700	Militan					
Model Rates	ANTONOV 124	1	23057	100	430	Civilian					
Run The Model	B 737	0	23537	110	460	Civilian					
ersonnel Results	B 747 400	1	3000	439	467	Civilian					
Training Paculte	B 767	1	9000	217	450	Civilian					
Training Results	B707-302B	0		141	470	Military					
System Results	B 737	2	0000	140	450	Military					
Fraining Courses	B 747	9	1000	380	498	Military					
raining Streams	B 767	1.	4000	210	462	Military					
Tools	C 130 H (NZ)	1.	4000	70	300	Military					
10013	C 130E	1.	4000	70	280	Military					
	C 130H	1.	4000	70	300	Military					
	C 130J	1.	4000	70	320	Military					
	C 141	4	1200	0	432	Military					
	C 17	7	7400	102	450	Military					
	C 273	1	1500	68	280	Military					
	C 5	1.	20000	88	450	Military					
	CN 235 CASA	0		48	230	Military					
	CIVIL AIR - Type 2	0		70	498	Civilian					
	CIVIL AIR - Type 3	0	Sec. 1	70	498	Civilian					
	DHC4	6	000	0	140	Military					
2 (F) 2	FA 10	0		0	460	Military					
At 41	F 111 ENI CON DAGO	0		0	460	Military					
	FALCON DA90	0	<i>C</i>		-100	military					

Figure 35: Setting Transport Availability

The availability of transport platforms can be set to change during the time horizon. The availability of each platform is assigned in blocks of time, with the number of platforms available during the time block set by the user. Each transport platform can have as many of these blocks setup as required and the blocks can overlap. To add an availability block, select the desired transport type and click on the *Add Block* button at the bottom left of the screen (Figure 35); this creates a block with zero platforms available along a default time line (repeating the process will add multiple blocks). Now select the newly created block (by opening the tree, by clicking the *[+]* next to the platform name, and clicking the desired Gantt chart) and enter the number of available transport platforms in the *Available* field at the bottom of the screen. The start and end times for the block can be changed by left-clicking and dragging either end of the block in the same way unit and operation start/end times can be edited.

# 3.2 The Logistics Output Screen

Once all of the inputs have been specified in the *Logistics Setup* tab, logistics results can be viewed by selecting the *Logistics Output* tab (Figure 36). By clicking the phase selection checkboxes on the left of the screen, results can be viewed by individual phase or multiple phases (in which case aggregate results are displayed).

Clicking the *Initial* tab displays the initial logistics requirements in tonnes for each class of supply. These initial requirements represent the forecast of what must be transported at the start of the operation and includes stores held in reserve. These values take into account sustainment stocks and the supplies required to build up the target days of supply stocks.



Figure 36: Logistics Output Screen

The *Sustainment* screen displays the ongoing daily amount forecasted for each class of supply that is required to sustain the operation. Initially, the average daily requirement for the selected phase(s) will be displayed (noting that the unit makeup may change during a phase, depending on the scenario, affecting the logistics requirement). The daily requirement for a particular month can be displayed by clicking the *Single Month* checkbox at the bottom of the screen; a slider appears when the checkbox is selected which allows users to select a particular month of the selected phase(s).

The *Strategic Lift* screen (Figure 37) displays a list of all lift platforms available for use. To the right of this list is a deployment timeline showing which platforms are assigned to be used for strategic lift and at what times. In the timeline, a blue bar represents the period that a platform has been assigned to move initial supplies to the area of operations (i.e., personnel, major systems, and reserve stocks for all classes of supply). A green bar represents the period the platform is assigned to transport ongoing sustainment supplies. Selecting an assigned platform will display information about its allocation below the deployment timeline within the *Transport details* box. Below this box, information is displayed with regards to the extent that strategic lift requirements of the selected phase(s) have been met (i.e. deficiencies in sea or air lift, based upon available platforms and the set period for the maximum time to deploy, is listed in percentage terms).

Selecting the *Plots* tab shows logistics information in chart form (Figure 38). Logistics requirements can be viewed in terms of average daily requirements (in kg) for each class of supply. In addition to logistics requirements, plots are available for the weight of major systems forecast to require transport over the time horizon, as well as unused air lift and sea lift capacity. The time horizon displayed for these plots can be modified by left clicking and dragging the ends of the green bar displayed below the graph.

ransport Deployments Name	Ava	ilable Type	Fleet type	Deployment Timeline
AN 74	6	Aircraft	Military	08 09 10 11 12 13 14 15 15
B 737	2	Aircraft	Civilian	1
CIVIL SHIP - Type 1	1	Ship	Civilian	. 0
CIVIL SHIP - Type 2	2	Ship	Civilian	
CIVIL SHIP - Type 3	1	Ship	Civilian	
HMAS JERVIS BAY	1	Ship	Military	
Transport details				
2 used for initial system o L used for ongoing sustai phase 1	deployment for phase 'p inment for phase 'p	se ' <i>phase 1</i> ' be hase 1 ' betwee	etween mont en months 2	hs 21 and 22 2 and 61

Figure 37: Viewing Strategic Lift Results



Figure 38: Viewing Logistics Requirements Plots

# 3.3 The Logistics Editor

The supplies module in A-SMART has been set up with the expectation that most users will obtain logistics results and use only the *Logistics Setup* and *Logistics Output* screens (using the default input parameters loaded). However, for those users who wish to conduct specific analysis, the *Logistics Editor* allows users to comprehensively revise underlying specifications for logistics inputs (Figure 39).

A-SMART													_ 7
Current ORBAT:	2008-16 ORBAT 50	enario: Default Scenario											
Scenarios	Transport Settings	Ammunition Settings	Location Setting	s System 1	Types Syste	m Classes							
Force	New Transmost	Persona Transmost											
Task groups	Contraction of Contra	g rearry manaport											
Operations	Title	Capacity (Passengers	) Capacity (kg)	Capacity (m^2)	Capacity (ISO)	Speed (knots)	Load Time (hours)	) Unload T	time (hours)	Fleet	Type	Cargo and Pax Cost Per Nm	0
Looking Cature	8 747 400	439	13000	0	0	467	0.5	0.5		Civilian	Aircraft	Individually 0	
Loginica Secur	8 767	217	19000	0	0	460	0.5	0.5		Ovilian	Aircraft	Individually 0	
Logistics Output	8707-3028	141	0	0	0	470	0.5	0.5		Matary	Aarcraft	Individually 0	
Logistics Editor	8 747	140	20000	0	0	490	0.5	0.5		Miran	Aircraft	Individually 0	100
Model Rates	8 767	210	14000	0	0	467	0.5	0.5		Military	Alerraft	Industrially 0	
Run The Model	C 130 H (NZ)	70	14000	ő	0	300	0.5	0.5		Miltary	Aircraft	Individually 0	
Descent Carefus	C 130E	70	14000	0	0	280	0.5	0.5		Miltary	Aircraft	Individually 0	
POWING NEWLS	C 130H	70	14000	0	0	300	0.5	0.5		Military	Aircraft	Individually 0	
Training Results	C 1303	70	14000	0	0	320	0.5	0.5		Military	Aircraft	Individually 0	
System Results	C 141	0	41200	0	0	432	0.5	0.5		Military	Aircraft	Individually 0	
Training Courses	C 17	102	77400	0	0	450	0.5	0.5		Military	Aircraft	Individually 0	
Tenining Channel	C 273	68	11500	0	0	280	0.5	0.5		Martary	Arcraft	Individually 0	
Transing Swearing	CN THE CAES	48	120000	0	0	220	0.5	0.5		MERIDARY	Aircraft	Individually 0	
Tools	CTUTI ATR - Tune 1	70		0	0	409	0.5	0.5		Chillion	Alectaft	Industrially 0	
	CTVTL AIR - Type 2	70	0	0	0	498	0.5	0.5		Chrilian	Aircraft	Individually 0	
	CIVIL AIR - Type 3	70	0	0	0	498	0.5	0.5		Ovilian	Aircraft	Individually 0	
	CIVIL SHIP - Type 1	300	0	1882	0	12	2	6		Civilian	Ship	Concurrently 0	
	CIVIL SHIP - Type 2	300	0	1882	0	12	2	6		Ovilian	Ship	Concurrently 0	
	CIVIL SHIP - Type 3	300	0	1882	0	12	2	6		Ovilian	Ship	Concurrently 0	
	DHO4	0	6000	32.5	0	140	0.5	0.5		Military	Aircraft	Individually 0	
	FA 18	0	0	0	0	480	0.5	0.5		Métary	Aircraft	Individually 0	
	P 111	0	0	0	0	950	0.5	0.5		Mercary	APCORT	Individually 0	
	HALCON DAYO	600	0	1933	0	100	0.5	0.5		Miltary	Arcraft	Concurrently D	
	HMAS KANIMELA	450	0	783	0	12	2	6		Miltary	Chin	Concurrently 0	
	HMAS MANOORA	450	0	783	0	12	2	6		Military	Sin	Concurrently 0	
	HMAS SUCCESS	0	0	0	0	12	2	6		Miltary	Ship	Concurrently 0	
	HMAS TOBRUK	350	0	1000	0	12	2	6		Miltary	Ship	Concurrently 0	
	HMAS WESTRALIA	0	0	0	0	12	2	6		Military	Ship	Concurrently 0	
	HMNZS CHARLES UPH	IAM 35	0	3875	0	12	2	6		Miltary	Ship	Concurrently 0	
	HMNZS ENDEAVOUR	0	0	0	0	12	2	6		Mitary	Ship	Concurrently 0	
	11-76T(I1-76TD) Ilyus	thin 0	35000	0	0	405	0.5	0.5		Ovilian	Aircraft	Individually 0	
	KC 10	0	0	0	0	537	0.5	0.5		Martary	Aircraft	Individually 0	
	NU 135	0	0	0	0	550	0.5	0.5		Manary	AFCTOR	Individually 0	
	P SC	v	×	¥	v	360	0.2	0.3		reactory	Archarc	subviolatity 0	M
	Title	HMAS JERVIS BAY					Load Time (	hours) 2					
	Capacity (Passengers)	600					Cunicad Time (i	hours) 6					
	Capacity (kg)	0						Floet Mit	tary				
Name	Canacity (m^2)	1832					-	Type Shie	2				
SHERE	Capacity (150)	0					Carpo and Range		Vi mentle				
A-SMART	compactify (caco)						the go and rase		- and the second				
	speed (knots)	12					Cost P	er remi D					

Figure 39: Logistics Editor Screen

In the *Transport Settings* tab, a list of all air and sea platforms available for strategic lift is displayed. Left clicking on a platform name allows various factors to be edited; including, capacity for transport of goods and passengers, speed and loading times. New platforms can be added and removed from the list as required by clicking *New Transport* or *Remove Transport*, respectively. All platforms defined in the list are available for selection in the *Logistics Setup* screen.

Selecting the *Ammunition Settings* tab provides a list of all of the ammunition types available for use in conjunction with defined major systems (Figure 40). Editable ammunition characteristics include physical dimensions, weight and cost per round. Note that ammunition usage rates are set by reference to weapon types, and are accessed through the *System Classes* tab discussed below (Figure 43).

Current ORBAT: Scenarios	2008-16 0	REAT Se	an animal Distant & Companya									And in case of the local division of the loc
Scenarios			enano: Delaut scenano					193				
and the second se	Transport	t Settings	Ammunition Settings	Location Settin	çs Syste	m Types 5	System Class	NS .				
Force	New A	mmunition	Remove Ammunition									
Task groups			Guanatoriana		1.00							1.0
Operations	ASN		Tide	Q	y Mass (kg	<ol> <li>Volume (m)</li> </ol>	-3) NEQ	Price	Length	Width	Height	0
Lookenice Setup	NIC-1	AGM Hell	fire ARH	0	0	0	0	0	0	0	ô	
Lociation Change	1164-01	ACT CHG	E SPAL LIAI	20	5	0.0081	29	21.86	0.3	0.15	0.18	1.0
Logence output	1164-02	ACT CHG	E SPAL LIAI	20	5	0.0081	29	21.80	0.3	0.15	0.18	
Logistics Editor	0215-00	ADADTED	COEN DOOL MLAD	45	23	0.0809176	0	3.65	0.779	0.340	0.376	
Model Rates	0215-19	ADAPTER	GREN PROJ MIA2 W/CART	F2 48	27	0.0809136	3	39.8	0.778	0.349	0.298	
Run The Model	1370-01	ADAPTOP	FUZE HOLE 2 INCH F5	45	0	0.0275784	0	7.55	0.445	0.313	5.198	
Descent Courts	3381-01	ADAPTOR	MOD FOR \$1MM MOR	0	0	0	0	0	0	0	0	
Personne keiults	3380-01	ADAPTOR	TV 184 FOR 81MM MOR	0	0	0	0	0	0	0	0	
Training Results	1341-06	AMMOND	UM NITRATE TECHNICAL PR	RILLED 50	58	0.084	0	0.79	0.7	0.48	0.25	
System Results	1341-07	AMMOND	UM NITRATE TECHNICAL PR	RILLED 25	33	0.084	0	0.79	0.7	0.48	0.25	
Training Country	1341-08	AMMONE	UM NITRATE TECHNICAL PR	ULLED 30	38	0.084	0	0.79	0.7	0.48	0.25	
training courses	1341-09	AMMONE	UM NITRATE TECHNICAL P	RILLED 40	48	0.084	0	0.79	0.7	0.48	0.25	
Training Streams	4342-01	BALLISTI	C DESC 260MM	1	0	0	11000	0	0	0	0	
Tools	1305-09	BANG TO	RP SECT 1 1/2 IN MK 1	4	31	0.00623878	2063	23.48	0.194	0.172	0.187	
	4317-01	BEANHOU	E CONNECTOR	0	0	0	1	0	0	0	0	
	4311-01	BLACK PC	TWOER	5	007	0	1000	0	0	0	0	
	4501-00	BOMB HE	CD 2000 LD PIX 01 PICUS	0	117	0	43545		0	0	0	
	4500.00	BOMB HE	GP 500 LB MK 82	0	227	0	87090	0	0	ő	0	
	1238-01	BOOSTER	MINE ATK X2E1	30	15	0.0308671	76.7	10	0.533	0.381	0.152	
	4171-01	BURSTER	TUBE DIVER RECALL KIT	25/	0.7	0.0029925	0	0	0.71	0.15	0.095	
	2230-01	CANISTE	R CS IRRT LIA4	12	4	0.0081	0	31.43	0.3	0.15	0.18	
	2230-11	CANISTE	R CS IRRT LIAS	0	0	0	0	0	0	0	0	
	0102-01	CART .22	CAL BALL F5	500	0 19	0.0119629	0.17	0.05	0.387	0.184	0.168	
	0102-11	CART .22	CAL BALL 5522LR	500	6 19	0.0119629	0.17	0.05	0.387	0.184	0.168	
	0101-01	CART .22	CAL 58 CAP	400	0 12	0.0081	0.1	0.05	0.3	0.15	0.18	
	0100-01	CART .22	CAL BLANK	100	00 8	0.00976579	0.13	0.06	0.432	0.178	0.127	
	2001-01	CART .22	HORNET	100	0 10	0.0081	0.4	0	0.3	0.15	0.18	
	2050-01	CART .22	3 REM 64GR PP	500	8	0.01072	1.7	0.43	0.335	0.16	0.2	
	2051-06	CART .22	3 REMINGTON R223R1 55	GRAIN PSP 360	0	0.0081	1.62	0	0.3	0.15	0.18	
	2010-01	CART .25	DALL CARDINE	100	0 9	0.0081	1.5	0.50	0.3	0.15	0.18	
	2020-01	CART 30	BALL PIL CARBINE	- 25	13	0.0081	3.6	0.89	0.3	0.15	0.18	
	0105-09	CART 30	CAL ROALL M2 11NA M25	100	0 50	0.0275784	3.6	0.96	0.471	0.313	0.198	
	0104-01	CART 30	CAL BALL M2 LNK	10	0 36.7	0.0275784	3.6	0.00	0.445	0.241	5.198	
	0106-11	CADT 30	CAL BLANK MED LINK	80	34	0.0255084	0.68	23.0	0.445	0.205	0.105	
	0661-23	CART 30	CAL TEACER M25	120	0 35	0.0401433	3.6	0.85	0.565	0.29	0.245	
	2070-01	CART .30	8 WIN 130GR HP	260		0.0081	2.9	0.9	0.3	0.15	0.18	*
		ADM MICH	CANADOS Y CARDON NAVE							0.000 00	6	1000
		AON NUC	· · · · · · · · · · · · · · · · · · ·					_	NC	A ba Ad	0.	
100		TIDE AGM	Heinre ARH						Price	per round	0	
Miller	Que	antity 0							0 U	ength (m)	0	
C 10 2	Mass pe	r Qty 0								Width	0	
ASMART	Volume per	r Qty 0								Height	0	

Figure 40: Editing Ammunition Settings

In the *Location Settings* tab (Figure 41), the locations and names of major military and transport hubs can be altered. Clicking the *AIR*, *SEA*, *RAIL*, *ROAD* and/or *BASE* checkboxes on and off alters the types of hubs shown in the locations list on the left of the map. Clicking on a location on the list allows users to edit the title, location, country code and location code of the site, as well as the location type. Locations can also be selected by right clicking a point on the map and selecting the location from the popup list of nearby sites. New locations can be added to the list by clicking the *New Location* button. Airfields and sea ports defined in this tab are available for selection in the *Route Planner* in the *Logistics Setup* screen (as discussed earlier, Figure 39).

Selecting the *System Types* tab (Figure 42), allows users to view and edit all major systems variants. Titles, SIGCs (Stock Item Group Codes) and the System Class to which the variant belongs can all be altered. New variants can also be added by clicking *New System Type*.

#### DSTO-GD-0595



*Figure 41: Editing Locations* 

A-SMART					_ C
Current ORBAT	User Manual ORBAT Scenario: User Ma	anual Scenario			
Scenarios	Transport Settings Ammunition Setti	ings Location Settings	System Types	System Classes	
Force	New System Type				
Task groups	Contraction of the second second				
Operations	Tibe	SIGC System Class			10
Looktics Setup	TRUCK, RANGE SAFETY, SINGLE LI	2320-0192 Miscellaneous			
Localities Change	TRUCK CARGO MED MC3, 4X4 4TONN	2320-0194 Miscellaneous			
Luginia Galia	TRUCK BIEL TANKER HAV MC3 AV	2320-0393 Miscelaneous			
Logistics Ealtor	TRUCK CARGO MOM CL MC3	2320-0210 Miscellaneous			
Model Rates	TRUCK, RECON, SF, LTWT, FPR, W	2320-0213 Miscellaneous			
Run The Model	TRUCK, CARGO MOM WITH CRANE 8.	2320-0217 Miscellaneous			
Personnel Results	TRUCK WRECKER HEAVY MC2	2320-701C Mack Recov			
Training Parada	WHLD ARMD VEH RECON T-1 W/EQUI	2320-711C ASLAV			
maning Kesula	WHED ARMD VEH MULTI ROLE T-II	2320-712C ASLAV			
System Repults	WHEELED ADMO VEHICLE, RECOVERT	3330.715C ACLEV			
Training Courses	WHEELED ARMD VEHICLE, SURVEILA	2320-716C ASLAV			
Training Streams	WHEELED ARMD VEHICLE, AMBULANC	2320-717C Miscellaneous			
Tools	WHEELED ARMD VEHICLE, PERSONNE	2320-718C Miscellaneous			
	WHEELED ARMD VEHICLE, COMMAND,	2320-719C Miscellaneous			
	WHEELED ARMD VEHICLE, RECON, T	2320-720C Miscellaneous			
	LIGHT ARMO VEH INF MOB TROOP B	2320-721C Miscellaneous			
	LIGHT ARRO VEH INF MOS COMMAND	1220-725C Miscelaneous			
	LIGHT ARMO VEH INF MOR DEWS B/	2320-726C Miscellaneous			
	TRUCK WRECKER HEAVY MC3 HRV	2320-727C Miscellaneous			
	TRAILER CARGO LTWT MC3	2330-0002 0.5/1t			
	DOLLY TRAILER CONVERTER SEMI-T	2330-0004 Specialist			
	TRAILER CARGO MDM, MC4 2-1/2 TO	2330-0005 0.5/1t			
	TRAILER CARGO LTWT 1/2 TONNE M	2330-0010 0.5/1t			
	TRAILER CARGO LT 1 TONNE MC2	2330-0012 0.5/10			
	CHASSIS TRATER IT 1 TONNE MC2	2330-0015 0.5/11			
	DOLLY SET LIFT TRANSPORTABLE S	2330-0019 Specialist			
	SEMITRAILER TANK WATER HVY MC4	2330-0021 BWT			
	SEMITRAILER CARGO 12.5M MC4	2330-0022 Articulated			
	SEMITRAILER TANK FUEL HVY MC4	2330-0024 BLFT			
	SEMITRAILER TANK/PLANT TRANSPO	2330-0025 TK/PLT			
	TRAILER BRIDGING MDM GIRDER 3	2330-0033 Miscellaneous			
	TRAILER GUN CARRIAGE PUNERAL M	2330-0035 Miscellaneous			
	TRALER PUM PLANT TRANSPORTER	2330-0039 10/101			
	TRAILER SHOP FORT LTWT MC2	2330-0044 Specialist			
	TRAILER CABLE REEL LT MC2	2330-0045 Specialist			
	TRAILER BAGGAGE LI MO4	2330-0052 Specialist			
	TRAILER CARGO MOM MC3	2330-0054 8t			0
sales.	TRAILER BOAT OFF SHORE SAFETY	2330-0055 Specialist			
Saller .	FIDE WHU ARROVER RECOVER	-1 MACON			
A-SMART	Sustan Class AD AV				

Figure 42: System Types Input Screen

In the *System Classes* tab (Figure 43) a list of all classes is displayed on the left hand side of the screen. Selecting a class allows the user to change the type to which the class belongs. For example, the 9 mm pistol class (of which there may be multiple variants as defined in the *System Types* screen) is of the PERS WEAPONS type. Selecting a class also allows users to edit the dimensions/weight of the system and whether it can be transported by air, sea and/or road, at the bottom left of the screen. If a vehicle class is selected, additional settings are accessible on the right hand side of the screen that relate to fuel usage. Selecting a weapon class (or a vehicle class fitted with a weapon) reveals options relating to ammunition usage rates for each ammunition type associated with the weapon. Users can define a high and medium level usage rate; the rate applied in calculating the model outputs depends on the *Operation Settings* inputs specified in the *Logistics Setup* screen (Figure 39).

Current ORBAT	: User Manual ORBAT Scenario: U	ser Manual Scenario						
Scenarios	Transport Settings Ammunition	n Settings Location S	ettings	Syste	em Types	System Classes		
Force	New Sustan Clare					77 SH 2		Valida Midu na
Task groups	New System Class						-	Construction of the state of th
Operations	Tide	Type	Length	Width	Height	Weight	0	Chart searce amontal R searche amonte
Looistics Setup	.50 CAL MG	PERS WEAPONS	0	0	0	0		Title ASLAV
Locistics Output	0.5/1t 105mm	TRAILERS	3.15	1.57	1.37	255.5		Fuel Type DIESEL
Logistics Editor	150 cu ft	REFRIGERATION	0	0	0	0		Eval Castacity (L) 0
Madel Dates	155mm	GUNS	7.51	2.79	2.13	8030		( as advert) (a)
Picoel Hates	20t/ Recov	TRAILERS	7.71	2.5	3.27	28100		Range (KM) 0
Run The Model	7.62mm Sniper Rifle	PERS WEAPONS	0	0	0	0		Fuel Usage (L/Hour) 0
Personnel Results	84mm	PERS WEAPONS	0	0	0	0		A construction of the second sec
Training Results	8t	TRAILERS	5.875	2.5	2.44	11980		Average use (noun/bay) o
System Results	9 mm Pistol	PERS WEAPONS	0	0	0	0		Fuel Usage (L/KM) 0
Training Courses	Abrams	A VEHICLES	9.83	3,65	2.44	63500		A service that Arman A a
Training Streams	Abrams wpn system	A VEHICLES	4.87	2.68	2.57	11050		www.age.ose.(www.bay).0
Tools	Ambulance	B VEHICLES	6	2.16	2.59	5800		Oil per Service 74.25
1006	APCA	A VEHICLES	4.87	2.69	2.53	11500		
	APCF	A VEHICLES	4.87	2.69	3.12	11050		Weepon Attributes
	APCM	A VEHICLES	4.87	2.69	2.53	11500		New ammunition attribute
	ARH (fuel usage)	AIRCRAFT	0	0	0	0		
	Articulated	TEATI FRS	7.71	2.5	3.27	28100		Ammunition type [Prist Line holdings] Advance ; Quick Attack ; Deliberate Attack ; Defence ; Withor
	ARVI,	A VEHICLES	0	0	0	0	25	CART 25MM MPT SD MK2 LNK 0 0 0 0 0 0
	A53508	AURCRAFT	0	0	0	0		
	ASLAV	A VEHICLES	6.57	2.775	2.65	13200		
	AVLB	A VEHICLES	10.56	3.25	3.5	45300		
	8200	ATECDART	0	0.60	0.09	39800		
	8707	AIRCRAFT	ő	0	0	0		
	BATH UNIT	WATER PURIFICATION	0	0	0	0		
	BUFT	TRAILERS	12.1	2.5	3.43	7670		
	BMSS Type A	CONTAINERS	2.44	1.83	1.93	770		
	Briss Type B B. Billister Storage Casacity (450 I)	WATER DURING ATION	2.43	1.83	1.93	4540		
	Richmaster	A VEHICLES	6.9	2.5	2.65	14000	4	
	Tide ASLAV	H TRIMERCE			1.500		1	
	System Class Type A VEHICLES						v	
	Length (m) 6.57							
	Midth (m) 2 775							
	hisiaht (m) 2.65							
	Holden (m) 2200							
All	weight (kg) 13200							
NUMBER	Air Transportable							
A CHEADY	Sea Transportable 🗹							
a state of the second	Road Transportable							

Figure 43: System Classes Input Screen

# 4. Running the Models and Viewing Results

# 4.1 Personnel Module

# 4.1.1 Running the Personnel Module

Once the set up of the scenario has been finalised, the model is ready to be run. In the *Run The Model* page (Figure 44), there is a *Run The Personnel Model* button that will initiate this process. If the *Simulate Delayed Separations In Reconstitution* checkbox is ticked, during the model run as rotational tours are completed a block of personnel will separate as the task group(s) enter reconstitution; i.e. at the point that voluntary separations are again allowed. It is assumed that the magnitude of separations at this point will be at the level of personnel who would have separated if voluntary separations were allowed during the period of build-up and deployment.



Figure 44: Running the Model

There is also a checkbox that allows users to turn training requirements on or off. By default training is turned on, and this means that promotions are limited based on whether personnel have completed all required courses (constrained by the number of training staff that are available), as well as satisfying minimum time in rank requirements. If training is turned off, promotions are limited only by whether personnel have reached the required minimum time in rank. If the *Model All Staff* checkbox is ticked, the personnel model will require all staff to be present for a course instead of just those defined as Instructors.

Different *Training Models* can be selected, these change the algorithm used to determine training priorities; it is recommended to use only the *Multi-Pass* and *Non-Constrained* models. *Multi-Pass* is the standard training/promotion model and takes into account TIR and completed courses to promote personnel. In contrast the *Non-Constrained* model will promote as many personnel as required, based on time in rank, to fill all gaps; this should only be used to provide an indication of the number of trainers, courses and personnel required for a certain force structure.

The model can also compensate for choke points in a trade stream; a choke point is a rank with a much smaller entitlement of personnel than both the rank above and below. This can lead to the situation where a large number of personnel are qualified for promotion but the number which can be promoted at each time step is limited by the size of the gap in the rank causing the choke point. In this situation, if the *Compensate for choke points* checkbox is ticked, the model will promote extra people into the smaller rank so the larger rank above it can still be provided with enough personnel. If choke point compensation is not used the larger rank will be starved of personnel because not enough personnel can be promoted through the choke point caused by the smaller rank. Note that issues with choke points could also be dealt with by altering the force structure and increasing the entitlement for personnel at the particular rank (see Section 2.2).

#### 4.1.2 Viewing Personnel Results

As discussed below, the primary means by which the personnel model displays outputs from a model run is as charts showing population levels against time in months; these charts can be viewed in the Personnel Results screen (Figure 45). Once users have run the model for the first time, it is recommended that they spend some time getting accustomed to how the results are shown, and how they can manipulate the interface to show relevant information.



Figure 45: The Personnel Results Screen

Plots show solid lines for calculated results, and dotted lines for target populations; target populations are calculated from the entitlement data, discussed in Section 2.2, aggregated from all of the relevant units (i.e. if there is no filter selected then all of the Army). The legend at the top of the graph indicates which line colours correspond to particular unit readiness levels. By moving the mouse pointer over the plot, the coordinates of the mouse position on the plot will be displayed next to the mouse pointer.

To analyse the chart in finer detail, it can be zoomed and panned to expand and move the area of interest. When zooming or panning, a target symbol will appear on the chart indicating the centre of zoom.

#### Zooming:

- Activate zoom •
- Increase vertical zoom
- Decrease vertical zoom
  - Increase horizontal zoom

- Decrease horizontal zoom •
- Hold down right mouse button - Move mouse up
- Move mouse down
- Move mouse to the right
- Move mouse to the left

#### Panning:

- Activate pan - Hold down left mouse button
- Pan to the left
- Pan to the right
- Move mouse to the right
- Move mouse to the left - Move mouse down
- Pan up • Pan down

- Move mouse up

Double left-clicking the chart will restore the original un-zoomed view. By clicking the *Copy* button in the top left of the chart, the plot can also be copied to the clipboard, or saved as a .png file. Alternatively, the plot can be saved as an image (by clicking the Save Image... button) or the raw data saved to Excel (by clicking the *Save Excel...* button).

# 4.1.2.1 Browsing Personnel Level Results

When the Personnel Results screen is first opened, it will display a plot of the entire force over the time horizon. Usually, it will be necessary to investigate smaller components of the total force in isolation to determine if there are specific issues forecast by A-SMART of the population dropping below target.

There are a number of ways of filtering the personnel results. Selecting tabs on the left of the screen allows the viewing of results by Stream (results can be displayed by corps, trade specialisation and rank), Units (results can be displayed down to sub-unit level) or Operations. Further, combinations of these filters can be applied; for example, in Figure 46, the results have been filtered so that the chart displays only AACC Cooks at the Corporal rank allocated to Operation Three.

The filters currently being applied are displayed in the title of the graph and if a filter is set on a *Stream, Unit* or *Operations* the title for that tab will be highlighted in red. In this way, users can keep track of what the chart is displaying at any time. Often, if a chart does not seem to match the results the user expects to see it is because a filter has unintentionally been set or left on. To clear a filter, go to the relevant tab and click the *Clear Filter* button at the top of that tab.

At the bottom left of the screen, the data tab allows the results to be filtered further; by selecting the tick boxes (or not) the results can be set to display (or not) personnel that are ring-fenced, not ring-fenced, promotable or not promotable.



Figure 46: Filtering Results by Corps, Rank and Operation

The presentation of chart information can be further manipulated by selecting the *Options* tab, located at the top of the chart (see Figure 47). By default, the chart shows all readiness levels and targets. In addition to these lines, a *Total* level and target can also be shown. The lines displayed can be turned on or off by clicking the box to the left hand side of the name of the particular readiness level. The target can be turned on or off by clicking the box immediately to the right of the name of the particular readiness level; note this will leave the line for the forecast population. When selected, the *Only* buttons display only the lines for the particular readiness level; this can be useful when interpreting complex results.

The *Show as* % option allows results to be viewed as a percentage of the target level rather than absolute numbers of personnel. *Show Deficit* only shows the gaps where the actual personnel level has fallen away from the target level. *Matthew Mode* allows a more simplified view by grouping personnel across all unit readiness levels into a single line for non-mobilising units and also for units building-up for deployment; this mode can facilitate results interpretation and presentation, especially for force sets which have many readiness levels set. If *Stack Results* is selected population levels will build on top of each other and the largest population value is equal to the total population. This option shows a clear view of the break down of the total population into readiness levels and mobilisation phases. By clicking the *Accumulate* option, the plot will add each month's population as a running total such that the total (say, separations) can be viewed up to any time step.

Stream	Units	Operations	Options							
Readiness Levels										
Read	iness	Target Only								
Dep     Train     On (         V High         V Med         V Low         V Base         V Initia         V Recc         V Excee         Onitian	loyed ning Call ium e al onstitution ess									
Option	s									
Show a Deficit Stack F Accum Show 1 Matthe	Option         Show as % (population)         Deficit (population)         Stack Results         Accumulate (movements)         Show Total (population)         Matthew Mode									
🔿 Data Se	ts									
Dat	a									
Ringfen     Non Rin     Promota     Non Pro	ced gfenced ible motable									

Figure 47: The Personnel Results Options Tab

# 4.1.2.2 Using Traffic Lights to Identify Failure Points

A 'traffic light' system allows users to see at a glance which parts of the force have not been fully sustained; colours are assigned to each node in the tree on the left of the chart. Depending on the results tab selected, traffic lights are displayed by *Stream, Units* or *Operation*s. The colours displayed – green, yellow and red – provide a quick way to assess problem areas where the actual population of a group has dropped below the target population at some point in the time horizon.

A unit which sustains 100% of its target population throughout the time horizon shows green, 80% shows yellow, and 50% or below shows red. Intermediate shades are used to allow users to approximate the failure rate for the item concerned; i.e. the closer to 100% the closer the shade of amber tends to green and, vice versa, if a population is close to 50% of target the closer the shade tends to red. If any of the subunits within a unit are failing, then the traffic light for the unit will show as a failure.

# 4.1.2.3 Browsing Personnel Movement Results

In addition to viewing the personnel *levels* of a particular group over the time horizon, results can also be viewed in terms of *movements* into and out of a group. Firstly, select the *Movements* tab at the top of the chart. A number of different types of movement can be displayed by selecting them in the *Movement Type* list at the bottom left of the screen; *Recruitment, Separations, Promotions Out, Reinforcements Out, Promotions In* and *Reinforcements In*. Note that *Separations* plots can, on some occasions, fall into the negative; this is due to *return to duty* personnel re-entering the force, after an absence (to simulate medical care and recuperation), at a greater rate than personnel leaving the force.

Clicking the checkbox to the right of a movement type will cause it to display as a negative value. This is useful to allow comparison of two complementary movement types. In Figure 48, the chart has been set up to display promotions in and promotions out for Lance Corporals in the Rifleman trade stream of RAINF. Promotions out has been set to display as a negative number so that the two movement types can be differentiated between and contrasted. From the chart we can readily conclude that there are more people being promoted into the rank than being promoted out of it.



Figure 48: Viewing Promotions In and Out over the Time Horizon

# 4.1.2.4 Browsing Time In Rank Results

Selecting the *Time In Rank* tab will show the average number of months personnel have been at a selected rank. This plot gives an insight into how rapidly personnel are being promoted and whether an ageing population is an issue.

# 4.1.2.5 Browsing Staff Years Results

By selecting the *Staff Years* tab the user can view summary data on the personnel population levels for the scenario; these results can be filtered by rank/corps, unit and operation

(Figure 49). The results are broken down by mobilisation phase and readiness level, as well as by target population, actual population, the ratio of actual to target populations, and the deficit population level. The numbers are sums across the entire length of the scenario expressed in staff years; one staff year is equivalent to the requirement of, say, 1 person for 1 year or 12 personnel for 1 month. The staff year results allow the user to compare the sustainability of different scenarios across the time horizon of the model run using a single metric; the staff year results effectively show the chronic shortfall of forecast personnel populations across the duration of the model run.



Figure 49: Staff Years Results

# 4.1.2.6 Browsing Corp/Rank Matrix

Selecting the *Corps/Rank Matrix* tab provides a higher level view of the personnel module outputs (Figure 51). It is designed to provide a summary of the results from the scenario by rank and corps. Figure 50 provides the colour key to the charts. As previously, an all green matrix indicates a fully sustained scenario. The metric used to generate the matrix is simply the lowest level reached by the particular rank and corps throughout the period of the model run. That is, a red colour indicates that the associated rank/corps population dropped to a level at or below 20% of the target level. A grey square in the matrix indicates no positions were present in the deployed group at that rank/corps. Note the results can be filtered; for example, by operation. The personnel sustainability matrix serves as a quick guide to the location and severity of failures in meeting personnel targets in the force over the scenario (Figure 51); for more detailed information on personnel sustainability, the personnel level results need to be investigated (Section 4.1.2.1).



Figure 50: Key to Sustainability Matrix

A-SMART													
Current ORBAT	: User Manual ORBAT Scenario: User Manual Scen	ario											
Scenarios	Coty Save Image Save Excel Scenario	User M	anual Scenario': A	Data, All stream	s, All Units, All Op	erations							
Force	Stream Linits Operations Options		Populations	Movements	Time In Bank	Staff Years	Corps/Rank Matrix						
Task groups	[These Dibard		Reco	at PTE	LCPL	CPL SGT	W02	WO1	LT.	CAPT	MAJ	LTCOL	CORPT
Lookenice Setup	Cost meet	101	RACT				10 10						
Logistics Output	Name R ALISTINT	-							_				
Logistics Editor	C NO CORPS		KALML										
Model Rates	E CRAA		AAAVN										
Run The Model	IE ORAACHD												
Personnel Results	C CRAADC		RAADC										
Training Results	B CRAAMC		RAA					_					
System Results	B RAAOC												
Training Courses	H CRAAPC		RAE										
Tools	B RACT		2440										
	E CAE												
	C RAINF		RAADC										
	• • RASIGS				-								0
			RAAPC										
			RAINF										
												-	
			AACC										
			AAPSYCH										
				6			2.0.0			2		8	6
			AUSTINT					1					
			RAAMC										
					_			-					2
			RACMP										
					_			_		-			
			NONE TU										
		~	RAAEC										
	< _	>								_			
	Data Sets		RAANC									-	
Num.	Data		RASIGS				10 400						
No. of Contract	Ringfenced		The second second										
A-SMART	Promotable		NO CORPS										
Constanting of the	V Non Promotable		_	-									

Figure 51: Personnel Sustainability Matrix

#### 4.1.3 Training Results

If the personnel results for a scenario show serious failure for some specific trade stream/rank combinations, the cause of the failure may be training related. The *Training Results* screen (Figure 52) allows exploration of training results by *Courses, Instructors* or *Students*; each set of plots makes use of a traffic light system to assist users in highlighting problem areas.

By Courses. Course plots are grouped by corps and display five lines:

- i. *Trained*: number of personnel who have undertaken training for the particular course selected
- ii. *Unable To Train:* number of personnel available to train but for which there are insufficient instructors available to conduct training
- iii. Courses: number of courses that are run
- iv. Other Staff: number of support staff allocated to conduct training
- v. Instructors: number of instructors allocated to conduct training

The relevant courses are listed on the *Courses* tree for a particular corps, and opening the tree of a course reveals a list of *Instructors* and *Students* allocated to the particular course. These

lists can be further opened to show the types of instructors and students that have been allocated/undertake the particular course. Selecting a name at any point in the tree shows the relevant plot.



Figure 52: Viewing Training Results

By *Instructors*. Instructor plots are grouped by corps and can be broken down by stream, rank and then the relevant courses (for the particular stream/rank). The number of *Available trainers* is displayed as well as a staked plot of the level of instructors allocated to each course; in this way the level of unallocated instructors, if any, can be quickly assessed. A shortfall of instructors is not displayed and should be interpreted by an inability (if any) to train available students (discussed below).

By *Students.* Student plots are grouped by corps and can be viewed by stream, rank and then by the relevant courses (for the particular stream/rank). The plots display the number of personnel *Unable To Train,* as well as the number of personnel *Trained* and the corresponding number of courses that are conducted.

Note that both the instructor and student plots correlate with the corresponding *Courses* plots and, in many ways, provide an easier method to assess the same information. Functionality also exists to link readily between the plot types (clicking on the buttons, *Show Course, Show Student* or *Show Instructor*, transfers the view to the relevant plot) and also to the relevant information in the *Training Courses* and *Training Streams* tabs when *Course Details* and *Personnel Results* buttons are clicked. For example, the user may be investigating plots for a particular course and wish to investigate the allocation of an instructor type over all courses; clicking *Show Instructor* will take the user to the relevant plot in the *Instructors* menu.

As an example, assume that in the personnel results for the Rifleman trade stream there was a failure to sustain the target population of Lance Corporals. At the same time, the population

of Privates for the stream was fully sustained. The common conclusion to be drawn in such a situation is that not enough Privates are being promoted to Lance Corporal to offset the separation rate. One of the reasons for such a failure could be that not enough Privates are being trained to the point where they qualify for promotion. As discussed above, the *Training* Streams screen displays the courses for each trade stream and rank. Using this screen we can see that Privates in the Rifleman stream must complete the SUBJ1 CPL course before they can be promoted. In the Training Results screen, the training data for this course is displayed. The number of trainees who undertook the course each month is shown in yellow, while the number who were available to take the course, but who could not because of a lack of instructors (capacity constraints), is shown in red. If the number of people unable to train is large as no personnel receive training, this indicates that there are not enough instructors available to provide an adequate number of courses. The user can then drill down further to determine the types of instructors that are not available in sufficient numbers and also, by transferring to the *Instructors* plots (by clicking *Show Instructor)*, view which other courses to which the particular type of instructor has been allocated. In doing so, the user may decide to alter the course details (for example, replace the type of instructor), by clicking *Course Details* and amending the inputs, and re-run the model to investigate the affects of the policy/input change.

# 4.2 Major Systems Module

# 4.2.1 Running the Major Systems Module

Clicking the *Run Major Systems Model* button on the *Run The Model* screen will initiate the major systems module calculation.

#### 4.2.2 Viewing Major System Results

The *Systems Results* screen (Figure 53) has similar functionality to the *Personnel Results* screen. Filters can be set to display plots for the forecast population levels for specific systems (under the *Major Systems* tab). These plots can then be broken down further for specific units (by selecting the *Unit* tab) and operations (by selecting the *Operations* tab). There is also an *Options* tab (Figure 54) that allows populations to be displayed for mobilisation phases and readiness levels (including all non-deployed unit readiness levels that are relevant, as well as, deployed, collective training, quarantine and the excess pool), as well as for maintenance pools (*Delayed Deep Maintenance, Deep Maintenance* and *Light Grade Maintenance*)<sup>18</sup> by selecting the relevant check box. These populations can be further broken down by selecting the check boxes for *Ringfenced* and *Non Ringfenced* at the bottom of the screen. There is also the option to *Stack Results* for the population plots such that when population options are multi-selected the lines will be stacked on top of each other.

<sup>&</sup>lt;sup>18</sup> Deep Maintenance refers to major systems unavailable due to planned long-term maintenance. Delayed Deep Maintenance are those major systems that are unavailable as they are scheduled for Deep Maintenance but there is insufficient capacity in the repair centres. Light Grade Maintenance refers to major systems unavailable due to unforeseen maintenance requirements. Loss refers to major systems rendered unusable due to accident or enemy attack.

#### DSTO-GD-0595

A-SMART										
Current Scenar	io: Default									
Scenarios	Major Systems Unit Operations Options	Popu	Population Movement							
Force	(Class Eller)		Population for scenario "Default" All Data ATRCRAFT, All Units, All Operations							
Task croups	Cear Hiter									
Operations	Title	14	N							
Longthe Schut	H OA VEHICLES									
Longer and Descent	E MBI	1.000								
Logeous Colput	AVBM	Ň.	30							
Logistics Editor	H M113									
Model Rates	E ACV									
Run The Model	B CARVL	- 10	12							
Personnel Results	B CAPCM									
Training Results	E OAPCA									
System Results	VAL2A	19	9							
Training Courses	E AIRCRAFT									
Training Streams	H OUHO									
Tools	HELICOPTER UTILITY BLACKHAWK	7.								
1005	H OB VEHICLES									
	B C VEIGCLES									
	B COMMUNICATIONS	N.	N N N N N N N N N N N N N N N N N N N							
	€ €GUNS									
	PERS WEAPONS	8								
	REFRIGERATION	20	•							
	B CINIT STOPPS	2 -								
	R Ourknown									
	WATER PURIFICATION	1.22								
	H WATERCRAFT									
		~								
	1									
	Poulation Data Sets									
Sale	Data	0								
A-SMART	Ringfenced		0 12 24 36 48 60 72 84							
	V Non Ringfenced		Time (Month)							

Figure 53: Viewing Major Systems Results



Figure 54: Setting Options for Major Systems Results

Selecting the *Movement* tab allows plots for the movement of systems due to *Loss, Procurement, Return To Service, Reinforcement Out* and *Reinforcement In* for the selected class or classes to be displayed. *Movement* plots can also be accumulated (say, to investigate the total losses over time).

# 5. Interpreting Model Outputs

# 5.1 Personnel Module

This section is intended to assist users new to the A-SMART tool in understanding how to interpret the results of a scenario. The output of the personnel module is necessarily complex and the tool provides a number of different ways in which results can be viewed. Initially, trying to understand the meaning behind the results – and uncovering the relevant information – can be difficult. Becoming familiar with the results interface is vital, as this will allow users to quickly access the information that they require.

This section goes through the process of investigating personnel and training results from a scenario, showing how data can be interpreted, and what conclusions may be drawn. Once users can come to grips with simple scenarios such as those detailed below, interpreting their own results will become much easier.

#### 5.1.1 Interpreting the Results

Sometimes when one looks at the results displayed in graph form and there is a failure to sustain a group's target population, it is difficult to tell whether the results represent a moderate widespread deficiency or a critical failure confined to a specific subset of the total population. Determining the extent of the failure is usually the first step in working out possible explanations for personnel deficiencies. There are two primary means of identifying the nature of a personnel failure; by selecting appropriate filters to view subsets of the total force and by calculating traffic lights. Often these techniques are used in conjunction. Consider the following results screenshot (note the lines in this plots are stacked):

From the Corps/Rank filter selection list on the left hand side of the screen, we can see that the force in the current scenario consists entirely of personnel from the RAA and RAINF corps. Turning to the chart itself, we can see that there is a population at the medium readiness level (displayed in green) and a population at the on-call readiness level (displayed in red). There is also an excess population, shown in black.

While the on-call population is almost completely sustained for the length of the scenario, the medium readiness population drops away considerably from its target. From the chart alone, however, it is not possible to tell how the force structure maps to the readiness levels. We have no way of telling, for example, whether the medium level is composed of RAA personnel, RAINF personnel, or a combination of the two corps. Further, is the failure shown confined to certain ranks, or spread evenly across all ranks?

#### DSTO-GD-0595



Figure 55: Unsustained RAA, RAINF Populations

To answer these questions, one can look at the traffic lights for all corps, stream and rank combinations (Figure 56). Traffic lights assign all force elements a colour based on the extent to which their constituent populations have been sustained over the time horizon. 100% sustainment displays as green, tending towards yellow for 80% sustainment, and moving towards an orange or red colour for lower levels of sustainment. Levels of 50% or less will show as a solid red light.

The red circles beside the RAA corps indicate that *at least* one of the ranks within that corps is failing critically, the yellow light next to RAINF indicates at least one of RAINF's ranks is partially failing. Expanding the tree to view the rank-level traffic lights gives a more detailed picture (Figure 56).

The traffic lights show that all RAA personnel in the current scenario are in the Ground Based Air Defence stream. Within this stream, the Lance Corporal and Warrant Officer Class 2 ranks are failing critically. The Sergeant and Corporal ranks are also experiencing a moderate failure.

Within the RAINF corps, only the Warrant Officer Class 2 rank in the Rifleman stream has failed to maintain its population target. All other ranks are fully sustained.

Stream	Units	Operations	Options								
Clear Filter											
Name											
🖃 🔴 RAA											
O Officer RAA -											
O CORPT											
OLTCOL											
OMAJ											
	O CAPT										
	OLT										
	Ground Bas	ed Air Defence -									
	WO1										
	• WO2										
	OSGT										
	OPecruit										
	VECIUIL										
	) Officer RAI	NF -									
	OCORPT										
	LTCOL										
	• MAJ										
	CAPT										
	●LT										
🖃 🔾 F	tifleman -										
	• W01										
	O WO2										
	• SGT										
	CPL										
	O De anuit										
	CRECIUIT										

Figure 56: Traffic Light System for Personnel Results

Selecting a results filter to view results for a specific corps is one way to determine how corps populations are represented in the different readiness levels displayed by the graph. Figure 57 shows the RAA corps in the corps/rank filter list.

The medium readiness level target for the RAA corps is 400, which corresponds to the medium level target for the entire force. In addition, there are about 70 members of the RAA corps in the excess pool. Since there is only one other corps in this scenario, all personnel not shown in the graph above must belong to that (RAINF) corps, including all on-call personnel. We can therefore conclude that the RAA corps has suffered a serious failure to sustain its population, while the RAINF corps is largely sustained over the time horizon.

#### DSTO-GD-0595



Figure 57: Unsustained RAA Population

Naturally, more realistic scenarios that attempt to model the entire Australian force will be more complicated and require a deeper analysis to determine specific points of failure. Nevertheless, the techniques described in this section can be used in such cases to identify specific failure points and provide insights into their causes (especially for individual training issues).

# 5.1.2 Achieving a Sustainable Force

Once a specific area of sustainability breakdown has been identified, users will often wish to explore ways, or policy options, that rectify the deficiency. Several factors relating to the way the scenario has been set up have an impact on sustainability. For example, changing the force structure – particularly by increasing the initial populations of units – may increase sustainability. For scenarios involving operations, reducing the operational tempo may reduce the impact of high attrition rates.

In addition to these structural scenario alterations, there are two fundamental methods of increasing force sustainability; modifying recruitment and separation rates, and eliminating training bottlenecks. These processes are detailed below.

# 5.1.2.1 Adjusting Personnel Model Rates

A general failure to sustain the target population across a stream or corps may be caused by a low recruitment rate or, alternately, a high separation rate.

Figure 58 shows the total population for the RAA corps. In this scenario, all RAA personnel (not in excess) are at medium readiness level, which is represented by the green line. The target population for the corps is a little over 400, but the actual population quickly drops away from the target as the scenario progresses.


Figure 58: High Separation Rate Causing Unsustained Population Levels

In this case, the black (excess) line represents recruits not yet ready to advance to private. Although these personnel are counted as part of the 'excess' priority level, the presence of this population does not indicate that the corps has met or exceeded its population targets. This can be seen from the traffic lights displayed for the various ranks to the left of the graph, which are predominately red, indicating serious failure in almost all ranks. In other scenarios, the excess pool may also include excess populations in other ranks.

The *Model Rates* screen (see Section 2.5) allows users to set recruitment and separation rates. Note that in scenarios where one or more operations are scheduled, the deployed attrition rate may also be a decisive factor in personnel sustainability; these rates can be set in the *Operations* screen.

In this scenario (Figure 59), the recruitment rate for the Ground Based Air Defence stream is 60 recruits per year (an adjustment of 60% of a base rate of 100). There is no lateral recruitment into higher ranks.

Rate: Recruitment ARA (person	s per yea	ir)															×
r Ranks Officers Rank A	djustmer	nt I	Varying P	Rates													
s Trade Stream	Recruit	PTE	LCPL	CPL	SGT	WO2	WO1		Rate Adjustment		Recruit	PTE	LCPL	CPL	SGT	WO2	WO1
F Rifleman -	100	0	0	0	0	0	0		3	00%	300	0	0	0	0	0	0
Ground Based Air Defence -	100							—- <b> </b> -		60%	60.00						0
	Rate: Recruitment ARA (person r Ranka Officers Rank A s Trade Stream F Rifeman - Ground Based Air Defence -	Rate: Recruitment ARA (persons per yez r Ranks Officers Rank Adjustmer s Trade Stream Recruit F Rifeman - 100 Ground Based Air Defence - 100	Rate: Recruitment ARA (persons per year) r Rank Officers Rank Adjustment 1 s Trade Stream Recruit PTE F Rifleman - 100 0 Ground Based Air Defence - 100 0	Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying F           s         Trade Stream         Recruit         PTE         LCPL           F Rifleman -         100         0         0           Ground Based Air Defence -         100         0         0	Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit!         PTE         LCPL         CPL           F. Rifleman -         100         0         0         0         0         Ground Based Air Defence -         100         0	Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         CPL         SGT           F         Rifleman -         100         0         0         0         0         Ground Based Air Defence -         100         0<	Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         CPL         SCT         WO2           F. Rifleman -         100         0         0         0         0         0         0           Ground Based Air Defence -         100         0         0         0         0         0	Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           si         Trade Stream         Recruit         PTE         LCPL         CPL         SGT         WO2         WO1           F Rifleman         100         0         0         0         0         0         0           Ground Based Air Defence -         100         0         0         0         0         0	Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           si         Trade Stream         Recruit         PTE         LCPL         CPL         SGT         W02         W01           F         Rifleman -         100         0         0         0         0         0         Ground Based Air Defence -         100         <	Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         CPL         SGT         WO2         WO1         Rate Adjustment           F         Rifleman -         100         0         0         0         0         0         0         Ground Based Air Defence -         100         0 </td <td>Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         CPL         SGT         WO2         WO1         Rate Adjustment           F         Rifleman -         100         0</td> <td>Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         CPL         SOT         WO2         WO1         Rate Adjustment         Recruit           F         Rifleman -         100         0<td>Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         CPL         SGT         W02         W01         Rate Adjustment         Recruit         Recruit         PTE           F         Rifleman -         100         0         0         0         0         0         300%         300 0         Ground Based Air Defence -         100         0</td><td>Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         SGT         WO2         WO1         Rate Adjustment         Recruit         PTE         LCPL           F         Rifleman -         100         0&lt;</td><td>Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         CPL           F. Rifleman -         100         0</td><td>Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         CPL         SGT           F. Rifleman         100         0</td><td>Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         CPL         SGT         WO2           F. Rifleman -         100         0</td></td>	Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         CPL         SGT         WO2         WO1         Rate Adjustment           F         Rifleman -         100         0	Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         CPL         SOT         WO2         WO1         Rate Adjustment         Recruit           F         Rifleman -         100         0 <td>Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         CPL         SGT         W02         W01         Rate Adjustment         Recruit         Recruit         PTE           F         Rifleman -         100         0         0         0         0         0         300%         300 0         Ground Based Air Defence -         100         0</td> <td>Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         SGT         WO2         WO1         Rate Adjustment         Recruit         PTE         LCPL           F         Rifleman -         100         0&lt;</td> <td>Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         CPL           F. Rifleman -         100         0</td> <td>Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         CPL         SGT           F. Rifleman         100         0</td> <td>Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         CPL         SGT         WO2           F. Rifleman -         100         0</td>	Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         CPL         SGT         W02         W01         Rate Adjustment         Recruit         Recruit         PTE           F         Rifleman -         100         0         0         0         0         0         300%         300 0         Ground Based Air Defence -         100         0	Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         SGT         WO2         WO1         Rate Adjustment         Recruit         PTE         LCPL           F         Rifleman -         100         0<	Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         CPL           F. Rifleman -         100         0	Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         CPL         SGT           F. Rifleman         100         0	Rate:         Recruitment ARA (persons per year)           r Ranks         Officers         Rank Adjustment         Varying Rates           s         Trade Stream         Recruit         PTE         LCPL         CPL         SGT         WO2           F. Rifleman -         100         0

Figure 59: RAA Ground Based Air Defence Recruitment Rate

By increasing the rate at which recruits come into the stream (Figure 60), we increase the pool from which personnel can be promoted to fill higher ranks (assuming sufficient training capacity – see below). Recruitment can also occur directly into higher ranks. For now, we will change the rate adjustment slider so that 100 base recruits per year come into the stream.

rget Ra	te: Recruitme	nt ARA (persor	ns per yea	ar)													
Other R	anks Offic	ers Rank /	Adjustmer	nt i	Varying i	Rates											
Corps	Trade S	itream	Recruit	PTE	LCPL	CPL	SGT	WO2	WO1	Rate Adjustment	Recruit	PTE	LCPL	CPL	SGT	WO2	WO1
AINF	Rifleman -		100	0	0	0	0	0	0	300%	300	0	0	0	0	0	0
laa	Ground Based	Air Defence -	100	0	0	0	0	0	0	100%	100	0	0	0	0	0	0

Figure 60: Increasing the RAA Ground Based Air Defence Recruitment Rate

Equally as important to overall force sustainability is the separation rate (Figure 61). The current separation rate profile for the Ground Based Air Defence stream is very high, with 60% of all Privates separating every year, and 30% of personnel at all other ranks separating yearly (300% of the base rate).

Target R Other	ate: Non-Deployed Separatio	n Rate (* Adjustme	% per y	ear) Varying I	Rates											~
Corps	Trade Stream	Recruit	PTE	LCPL	CPL	SGT	WO2	WO1	Rate Adjustment	Recruit	PTE	LCPL	CPL	SGT	WO2	WO1
RAINF	Rifleman -	0	20	20	20	20	20	20	100%	0	20	20	20	20	20	20
RAA	Ground Based Air Defence -	0	20	10	10	10	10	10		0	60	30	30	30	30	30
									-							

Figure 61: RAA Ground Based Air Defence Separation Rate

Cutting the separation rate by a third for all ranks (i.e. back to 100% of the base rate) should translate to a marked improvement in sustainability for this stream (Figure 62).

Target Ra	rget Rate: Non-Deployed Separation Rate (% per year)																~
Other I	Ranks Officers Rank	Adjustme	ent	Varying I	Rates												
Corps	Trade Stream	Recruit	PTE	LCPL	CPL	SGT	WO2	WO1	Rate Adjustment		Recruit	PTE	LCPL	CPL	SGT	WO2	WO1
RAINF	Rifleman -	0	20	20	20	20	20	20		100%	0	20	20	20	20	20	20
RAA	Ground Based Air Defence -		20	10	10	10	10	10		100%		20	10	10	10	10	10

Figure 62: Decreasing the RAA Ground Based Air Defence Separation Rate

Rerunning the model, we observe an almost complete sustainment of the target population for the stream over the time horizon (Figure 63).



Figure 63: Almost sustained RAA Population Levels

The ability to adjust model rates is useful for performing 'what-if' analysis. For example, a lower separation rate could be specified to investigate the possible impact of introducing retention bonuses for army personnel. Small adjustments in recruitment or separation rates (or a combination of these) may produce large and sometimes unexpected effects.

## 5.1.3 Eliminating Training Bottlenecks

Sometimes a failure to sustain a target population is caused by an insufficient flow of personnel from lower ranks to higher ones. In the scenario below, (Figure 64) the RAA corps population is not sustained. We can see from the traffic lights that two ranks in particular (Lance Corporal and Corporal) are failing badly.



Figure 64: Unsustained RAA Population Level

Looking only at the Private rank in the RAA corps, we can see that the target level has been fully sustained – indeed, there is a considerable excess Figure 65).



Figure 65: Sustained PTE Population Level with Excess

In contrast, the Lance Corporal rank is failing critically (Figure 66).



Figure 66: Critically Failing LCPL Population Level

The fact that there is an overabundance of Privates at the same time as there is a dire shortage of Lance corporals indicates that something is preventing promotion from the lower rank to the higher one. Qualified personnel in one rank will always be promoted if there is a gap between the target and actual levels in the rank above.

Personnel of a particular rank qualify for promotion if they have reached the specified minimum time in rank level and have completed all required training. Any issues with respect to meeting minimum time in rank requirements should resolve themselves over the course of a ten year scenario, since most minimum TIR periods are significantly less than ten years. In this case, there must be a training bottleneck that is preventing promotion.

One can find out which courses are causing the bottleneck for promotions on the *Training Results* screen (Figure 67), which also uses a traffic light system to show training bottlenecks in a similar way to the *Personnel Results* screen.

The red line shows the number of people who were available to train, but could not due to capacity constraints (that is, a lack of instructors). The yellow line represents the number of people who undertook the course. The green line shows how many times the course is run each month.

The screenshot in Figure 67 has expanded the Courses/Corps list to show, Students $\rightarrow$ RAA $\rightarrow$ Ground Based Air Defence $\rightarrow$ PTE $\rightarrow$ SUBJ 1 CPL. From this we can see the other courses required for training at the PTE level are all being fulfilled because they have solid green lights next to them; only SUBJ 1 CPL is causing problems. Clicking on the *Course Details* button at the top of the screen while SUBJ 1 CPL is selected shows the details of that course (Figure 68).



Figure 67: Training Results for Ground Based Air Defence, PTE, SUBJ 1 CPL

A-SMART	and the state of t		
Current Scenario	b: User Manual Example		
Scenarios	Corps / Course / Module	Module Details	
Force	Riter Expand	🗑 Remove Module	
Task groups	Corps / Course / Module	Title Current Con	
Operations	AABC	300101	
Logistics Setup	AALC	Length 42 Days	
Logistics Output	AUSTINT	Instruction to Students 1, 8	
Logistics Editor	CIV		
Model Rates	Logistics	Requirements	
Run The Model	E Army Partick Course - ARA	Personnel Systems Ammunition Facilities POL Rations	
Personnel Results	E RSM	Testnator Required Load Teste	
Training Results	C SUBJ 1 CPL	True 1 100 Course RAA/CAPT	
System Regults	SUBJ 1 CPL	Remove	
Training Courses	# 5080 1 561 # 5083 1 WO		
Training Courses	RAACHD		
Interning Surgario	RAADC		
Tools	RAANC		
1 mar 1 m	AAAVN		
	AACC		
	E RAA	s	
	RAAOC	Issue Crimte	
	RACMP	Instructor	
	RACT	Required 1	
	DAEME		
	(E RAINF	Loading 100 0%	
	RASIGS	ECNs	
	RAAEC	Corps ECN Rank Add ECN Corps	100
	AAPRS	RAA OFFR CAPT Remove ECN BAA	2
	0040725.0	RAAC	
		RAADC	
		RACMP	
		RAE .	
		RAEME	
		RAINE	
		RASIGS RAAEC	
All		RAAMC	
No. 14		AAPRS	4
NAMES OF TAXABLE PARTY.			

Figure 68: Personnel Requirements for SUBJ 1 CPL Course

We can see that the SUBJ 1 CPL course requires an RAA Officer, with rank of Captain but there are currently no RAA Officers in the scenario population so this course cannot be run and is causing the bottleneck (note that this is a fictitious example for the purposes of illustration only). To fix this problem, we can change the instructor requirement to an RAINF Officer, with rank of Captain.

Re-running the model, the personnel model results for the RAA corps show a large difference (Figure 69). Although there is a slight departure from the target level at the beginning of the scenario, this is soon corrected, and thereafter the force is fully sustained.



Figure 69: Sustained Ground Based Air Defence Population Level

For the Lance Corporal rank, the actual population drops off initially as before but soon recovers as Privates are promoted into the gaps created by Lance Corporal separations (Figure 70).



Figure 70: Sustained LCPL Population Level

One way of confirming that promotions are taking place to fill the gaps in the Lance Corporal rank is by viewing the *Movements* plots. Selecting the Lance Corporal rank, clicking the *Movements* tab, and clicking the *Promotions In* option displays Figure 71, which shows the number of promotions from Private to Lance Corporal for every month in the time horizon.



Figure 71: Promotions into LCPL

By increasing the training throughput we have improved the sustainability of the force. One final point to note in relation to individual training is that a scenario with a high personnel turnover rate (that is, a high recruitment rate coupled with a high separation rate) will require a larger training throughput to achieve sustainability than a scenario with a low turnover rate. This is because recruits enter the force with no training, while those separating will include personnel of higher ranks and greater training which are more difficult to replace.

## 5.1.4 Interpreting Scenarios Involving Deployments

Results from scenarios involving operational deployments can be complex. This is especially true when an operation involves multiple rotations for different task groups (note that plots can be filtered by *Stream, Units,* and *Operations,* and lines can be turned on or off under the *Options* tab to simplify the plots). Consider the scenario in Figure 72.

As in previous examples, this scenario involves personnel from the RAA and RAINF corps. Again, the RAA corps begins the scenario at the medium readiness level (shown in green) while the RAINF personnel are initially at the on-call level (shown in red).



Figure 72: Personnel Results Involving an Operation

An operation has been scheduled to begin at month 44 and continue until month 90. Two task groups have been set up to rotate with each other, with the maximum tour of duty length for each group set at 12 months. The first task group is composed solely of RAA members, and contains all of the other ranks personnel. The second task force is similarly comprised of RAINF other ranks personnel.

The RAA task force is the first to be deployed. Before it can be deployed, however, it must move up from medium readiness through the various readiness levels. The global settings for this scenario dictate that before a deployment, personnel in a task group must spend 12 months at medium readiness, six months at medium-high, three months at high, and three months at the on-call level. In Figure 74, working backwards from month 44, we can see a significant increase in the on-call population for three months (since both the RAINF and RAA personnel are temporarily at this level). Preceding this, the RAA personnel spend three months at high readiness (shown in orange), and six months at the medium-high level (shown in yellow).

This is shown more clearly if we filter the results to view only RAA personnel (Figure 73).

The time that the task force transitions from its initial medium readiness level to mediumhigh, as it starts to mobilise for the operation, is dictated by the time needed to move through all of the levels until it is ready to be deployed. In this case, 6 + 3 + 3 = 12, so mobilisation begins 12 months before the start of the operation, at month 32.



Figure 73: RAA Personnel Results

The first rotation remains deployed (shown in blue) for 12 months, after which it enters its reconstitution phase (shown in pink). The nominal reconstitution phase for the operation has been set to six months, so after deployment there is a six month period before personnel become available to reinforce operations. There is a six month collective training period (shown in dark green); note that during this period personnel are available for individual reinforcement.

Collective training only occurs between deployments within an operation, and will not occur after a task group's final deployment for an operation. Thus at month 80, the RAA task group's final deployment rotation concludes, and it enters into reconstitution for six months. The group's personnel then return to their initial medium readiness level from month 86 onwards, even though the operation does not end until month 90.

Now that the different elements of the results for the corps have been identified, we can turn our attention to what the results say about the sustainability of the corps across this scenario. It is often simpler to look at results for each individual corps, rather than the combined results for all corps. Since there is no transfer of personnel between corps, there are no external injections or drains on a corps' population. The only caveat to this is that instructors for courses undertaken by a certain corps may come from anywhere in the force, so if it seems as if there may be a training bottleneck, rectification may involve adjusting populations of certain ranks in other corps.

At the beginning of the scenario, the population level starts to drop below its target, before starting to increase around the 20 month mark. By the time that the operation commences at month 44, the corps population is almost fully sustained.

The population falls fairly rapidly during the deployment phases in the operation. This is because the attrition rate of the operation is higher than the separation rate that is applied to non-deployed personnel. By the end of the first rotation, the population level is around 340, down from a target of 410.

During the reconstitution periods (shown in pink), the population of the corps continues to drop. This is because reinforcement and promotion do not occur into a reconstituting group. Although the reconstitution population is dropping, the excess population grows significantly during these periods, since personnel are not being taken from the excess pool for reinforcement. When the reconstitution period is finished, there is a sudden movement as personnel are shifted from the excess pool into the collective training pool.

Overall, the results show a reasonably serious failure to sustain the target population for the corps. In general, scenarios involving deployments with high casualty rates will struggle to be sustainable unless recruitment and training rates are high, return to duty rates are high and/or personnel can be sourced from elsewhere in the force to reinforce the deployment(s). *Return to duty* rates specify the number of casualties that will rejoin the force after a certain period of time. Clearly, a high return to duty rate results in a lower effective attrition rate.

## 5.2 Major Systems Module

The major systems module provides analysis of the sustainability of major equipment in the army; it makes use of SED Data to generate target populations, procurement and loss rates, as well as inputs that relate to maintenance requirements including availability rates (to represent unforeseen or short-term preventative maintenance) and deep maintenance frequency/duration (to represent long-term preventative maintenance requirements). The module produces results in a similar format to the personnel module and shows the distribution of major systems to units, highlighting periods of under population (if any). The traffic light mechanism assists analysts to quickly identify major systems that need further investigation into why their forecast population fails to meet required targets. In this way the module can be used to investigate whether the fleet size is large enough to support the force requirements, in terms of both non-deployed and operational commitments, facilitating 'what-if' analysis across scenarios with changing operational tempo, directed readiness levels and planned force structure changes.

## 5.2.1 Insights into the Setup and Analysis of Major System Module Results

Major systems are either allocated to units (deployed or non-deployed) or are unavailable due to unforeseen or deep maintenance (including awaiting deep maintenance if insufficient capacity). The total population of a system fleet varies according to annual procurement and loss rates, which can be set by operational phase or unit readiness level. A high population of equipment in or awaiting maintenance, usually results in periods during which the population levels are below target. The frequency of deep maintenance depends on the readiness level or operational phase to which the system is allocated; for example, if a unit deploys its equipment will likely require deep maintenance more frequently. The population of unavailable equipment can also rise if the deep maintenance capacity is low. Deployed major systems are quarantined at the end of an operation. The quarantine period can be set by

operation and can have a significant impact on the available population. Concurrent activities will commonly cause equipment shortfalls in the major system module; increased time between activities will ease the expected major system population level shortfall.

## 5.2.1.1 Basic Example

In this example (Figure 74), a simple force has been set-up with two units, A and B. Unit A has been allocated 200 ASLAVs (from 2 variants), 10 105mm guns and 24 personal weapons. Unit B has been allocated 12 personal weapons and 300 Bushmasters (from 2 variants). There is no annual procurement, and only the A-vehicles have been set-up to have stock pools, loss rates, availability rates and deep maintenance requirements. No operations have been set-up at this point. The model results show populations broken down by readiness levels, deep maintenance, delayed deep maintenance, preventative maintenance, excess, and, if operations are set-up, deployed, collective training and quarantine. Similar options exist for the display of plots as for the personnel plots (i.e. breaking the populations down by units or operations, staking, accumulate lines etc.).



Figure 74: Major Systems Sample Force Structure

If we inspect Bushmaster-IMTV as an example (Figure 75), the total initial population is 400 (200 allocated to Unit B SED, as discussed above, and an additional 100 in each of the Repair and Attrition stocks). Note that when time equals zero, if deep maintenance periods have been set then the initialisation of the population will allocate vehicles across the cycle of deep maintenance, including to the deep maintenance pool; in this case as the deep maintenance period is 2 months and the time between deep maintenance is 36 months, then 200/(36+2) equals approx. 5.26 for each class or 10.53 vehicles in deep maintenance/delayed deep maintenance (as there are 2 monthly classes) and the remaining 189.47 allocated to the unit (evenly distributed between the 36 monthly classes). Redistribution then occurs and equipment is moved from the excess pool (including repair, attrition and loan stocks) to reinforce any depleted units and into light grade maintenance (purple line in Figure 75). Note

that this leads to a group or 'cohort' of vehicles which enter the force at the same time and, therefore, will in 36 months enter deep maintenance at the same time (causing the step patterns observable in some of the lines here). As deep maintenance has a capacity to service Bushmaster-IMTV set at 10, initially this causes a small build-up<sup>19</sup> in those vehicles queued each month (light blue line in Figure 77). At month 36, when those vehicles that reinforced Unit B at the start of the model run, move into the deep maintenance queue there is a significant increase in the number of vehicles delayed, awaiting deep maintenance; the same affect is observable every 36 months. Due to reinforcement from the excess vehicle pool, in this scenario Unit B is fully sustained (orange line in Figure 75).



Figure 75: Major Systems Sample Results – Bushmaster IMTV Population Levels

If an operation is added to the scenario (Figure 76), the IMTV population is adversely affected (Figure 77); a task group was set-up made up of Unit B and allocated to an operation of 12 months duration (24-36 months). Unit B has a non-deployed readiness level set to High (orange line); units move up through readiness levels before deploying. In this case, the duration of On Call (red line) is 3 months, which starts at 21 months (3 months before the operation begins at month 24). The loss rate has been set to increase on moving from High to On Call to Deployed (from 2 to 5 to 36% per annum), the availability to decrease (from 95 to 90 to 80%) and the time between deep maintenance to decrease (from 36 to 24 to 12 months). Until the operation begins the impact of changing the parameters on population and availability of the vehicles is small. Once the operation begins, the total population starts to drop and there is a significant build up in vehicles awaiting deep maintenance (light blue

<sup>&</sup>lt;sup>19</sup> The initial distribution across deep maintenance and delayed deep maintenance is 10.53 and as the capacity of deep maintenance is 10 this leaves 0.53 delayed (10.53 - 10 = 0.53).

line). At month 31 there are no longer sufficient vehicles in the excess pool to compensate for the loss and reduced availability, and the deployed group cannot be fully sustained for the last half of the operation. At the completion of the operation, the number of vehicles awaiting deep maintenance does not significantly change as there was no spare maintenance capacity. If the capacity of deep maintenance is increased by 50% to 15, the results show that there is still a build-up in delayed deep maintenance during the operation but the backlog is effectively removed by the 6 year point (Figure 78). The numbers of vehicles required in the fleet to fully sustain the deployment (with the given loss, availability and maintenance capacity inputs), can be experimented with. If the deep maintenance capacity is returned to 10, then an increase of the repair pool by 100 vehicles leads to an almost fully sustainable operation for this scenario (Figure 79). Note that some input parameters can be set to vary during the scenario run; procurement levels, loss rates, availability rates, as well as deep maintenance period, frequency and capacity. Movement plots can be used to assist in interpreting results and can show the number of major systems moving due to loss, procurement, return to service, reinforcement in or reinforcement out.



*Figure 76: Operation Set-up* 



Figure 77: Major Systems Sample Results – Bushmaster IMTV Population Levels with Operation



*Figure 78: Major Systems Sample Results – Bushmaster IMTV Population Levels with Operation and Increased Deep Maintenance Capacity* 



*Figure 79: Major Systems Sample Results – Bushmaster IMTV Population Levels with Operation and Increased Repair Pool* 

# 6. A-SMART Data Formats and Loading Procedures

## 6.1 Introduction

This section describes all of the formats for data that can be loaded into A-SMART using the built-in data loading tools. The most important of these are the entitlement data loading tools, for personnel and major systems, but this section also covers various other run-time configuration data which needs to be managed when using the tool (Figure 80).

## 6.1.1 Typographic conventions

The following typographic conventions are used within this section:





Figure 80: A-SMART Data Loading Tools

## 6.2 ORBAT Data

## 6.2.1 Quick Start

The data loader is accessed by going to the *Tools* tab as shown in Figure 81. The general procedure when loading a new set of entitlement data is to create a new ORBAT, load the personnel entitlements (which will fill out the migration structure) and, if they are available, to then load the matching major system entitlement data.

Once you have prepared your data (see Section 6.2.3, Data Files), the steps are as follows:

- 1. Select a structure file (see Section 6.2.3.1, Structure File).
- 2. Select a personnel control file (see Section 6.2.3.2, Personnel Entitlement Data).
- 3. Execute by clicking on Load Personnel.
- 4. Review the loading log, as shown in Figure 82.
- 5. Either accept the data load by clicking on Finish or cancel the load by pressing Cancel

A-SMART		G.
Current ORBAT:	Default Scenario Default	
Scenarios Force Task groups	Data Tools SQL Reports Old Tools SEDs Rates and Levels	
Operations	Tree Structure File	Select File
Logistics Output	Control File)	Select File
Logistics Editor Model Rates	[Load Personnel]	
Run The Model	Asset File	Select File
Personnel Results Training Results	Load Personnel Asset Levels	
System Results	Tree Structure File	Select File
Training Streams	System Control File	Select File
Tools	Load Systems	
New.		

Figure 81: Entitlement Data Load Screen Personnel

A-SMART	
Current ORBAT:	Default Orbat Scenario: Default Scenario
Scenarios	Data Tools SQL Reports Old Tools
Scenarios Force Task groups Operation Logistics Schup Logistics Schup Logistic	Data Tools         SQL Reports         Old Tools           Stop         Texts and Levels           Personal         Texts Structure File         CMDocuments and Settingstrootzed/My DocumentstarctSMARTAASISMARTAASIDocumentationexample-personnel-errorstinee-structure.csv           Setting         Chading tree file         ChDocuments and Settingstrootzed/My DocumentstarctSMARTAASIDocumentationexample-personnel-errorstinee-structure.csv           Setting         Chading versional         Settingstrootzed/My DocumentstarctSMARTAASIDocumentationexample-personnel-errorstinee-structure.csv           Setting         Chading versional         Settingstrootzed/My DocumentstarctSMARTAASIDocumentationexample-personnel-errorstinee-structure.csv           Setting         Settingstrootzed/My DocumentstarctSMARTAASISMARTAASIDocumentationexample-personnel-errorstinee-structure.cs
ASMAPT	Cancel Finsh

*Figure 82: Output Shown When Importing Personnel* 

## 6.2.1.1 Major Systems

Once you have prepared your data (see Section 6.2.3, Data Files), the steps are as follows:

- 1. Select a structure file (see Section 6.2.3.1, Structure File).
- 2. Select a major system control file (See Section 6.2.3.3, System Entitlement Data).
- 3. Execute by pressing Load Systems
- 4. Review the loading log, similar to that shown in Figure 82.
- 5. Accept the data load by pressing

Note that unlike the personnel loader, there is currently no opportunity to revert the load once it has finished. As the data is loaded it will populate the tree, so if it fails part way through the only way to proceed is to <u>delete the whole force structure</u>, and start again.

## 6.2.2 Data Sources

The entitlement data import functionality in A-SMART is intended only as a mechanism to import external data for modelling purposes. It is not possible to export the information in a compatible format, for editing in an external application. There is an export-only functionality for entitlement data that is accessed from the *Force* tab, and there is also a whole-scenario backup and restore feature accessible from the *Scenarios* tab, and finally an ad-hoc query facility based on Standard Query Language.

### 6.2.2.1 Personnel Entitlements

The ideal level of detail for personnel entitlement data contains all force structure information, department identifiers, individual 'Jobcodes', position numbers, start dates and expiration dates for all positions. This allows the A-SMART data loader to properly identify alterations to the force structure over time, and accurately represent the force hierarchy within the system. Only queries from PMKeyS can satisfy these requirements, but other potential data sources are listed here.

### 6.2.2.1.1 PMKeyS

The PMKeyS exports contain position information for all personnel of interest. Because it comes from a live system, the data is guaranteed a relatively high level of consistency. Exports can also be generated at an arbitrary date, so for example, if a query is run to obtain a snapshot at the beginning of each year, a set of data which reflects the changing force over time can be created. An additional query supplies the full set of live hierarchy information which can be immediately cross referenced with the SED data to produce a usable force structure. This is the only data format supported by A-SMART.

### 6.2.2.1.2 DPlans-A

The DPlans-A web site contains a selection of specific PMKeyS data exports for a future date. This export does not contain any data for subunits which have moved, changed significantly, or ceased to exist prior to the export date, and is therefore unsuitable for sustainability analysis in the intermediate time span — one of the primary design goals of the A-SMART application. The available data also has no up-to-date hierarchy information associated with it — a major limitation that would require some manual processing.

## 6.2.2.1.3 SED Review Team

The SED review team maintain a parallel set of data which is to be loaded into PMKeyS. Being manually edited there are many consistency problems and errors are not uncommon. But the primary issue is that the department identifiers are not known until the data has been loaded into PMKeyS. Without them, significant manual manipulation is required to encode the force hierarchy in a usable way.

## 6.2.2.2 Major Systems Entitlements

Ideally the major systems entitlement data would supply similar information to that available for personnel entitlement data: force structure, department identifiers, SIGC codes, activation and retirement dates, in an electronic format. Unfortunately not all of these requirements can be met by any current data source, and significant manual work is required to form suitable input files. One limitation is that there is no temporal information available in any of the data, although temporal information can be extracted from the personnel data set — once department identifiers have been added to the system data.

## 6.2.2.2.1 Defence Entitlements System

A system called the Defence Entitlements System (DES) stores the major systems entitlement data, but it only includes current information. Being an electronic database, the quality of the information is good, but it cannot be used on its own for modelling future dates. Unfortunately the DES system appears not to use the same department identifiers as PMKeyS; as a result there is no automated way to merge the two data sets.

## 6.2.2.2.2 SED Review Team

The SED review team creates data sets which are loaded into the DES system. As such they have a more up-to-date set of data. Unfortunately, as this data is manually entered it suffers from various consistency problems and data entry errors. As with the DES data, there are currently no department identifiers which can be used to link the major system entitlements to the personnel entitlements.

## 6.2.3 Data Files

For personnel data, a structure file, a control file, and a set of annual SED files are required. For major systems data, a structure file, a control file, and a set of per-unit data files are required. These are detailed in the following sections. All files must be stored within the same folder in the non-proprietary and efficient comma separated values (.CSV) format for the application to be able to read them.

## 6.2.3.1 Structure File

The structure file contains the hierarchical structure and names of all of the units and subunits as stored in PMKeyS. The structure will be mirrored completely inside A-SMART for all active units. The structure file contains every unit and every subunit to which personnel and systems are assigned. The Department ID is used to correlate the data, so every Department ID present in the Personnel File must have a corresponding entry in the Structure File or <u>the entire data</u> <u>set will refuse to load</u>. The Structure File may include subunits and units which are not referenced in any of the Personnel Files, these empty subunits will be pruned automatically.

## 6.2.3.1.1 Structure File Format

The structure file consists of rows of data, each of which specifies a single Unit, Brick or Subunit. The file must contain the header and columns as specified in Table 2.

A	В	С	D	E*	F*	G*	Н	I*	J*
Unit ID	Unit Desc	Sub Unit	Sub Uni Desc	tDeptID	Short Desc	ADO Unit Indic	Tree Level	TreeNd Num	Prnt Node
		ID	2 000			inuic	20101		Tioue
100001	DEFENCE			100001	DEFENCE	U	1	1	0
140000	OFFSEC/CDF			140000	OFFSEC/CDF	U	2	2	1

Table 2:Example Structure File

Note that only the columns indicated with (\*) are actually used, but they must be placed in the same positions as shown above. The numerical columns must be formatted as plain numbers when saved to a CSV format file (by default they appear to be formatted using thousands separators). The meaning of each significant column is:

- **DeptID** The unique identifier for the department. This is the master key which links personnel and systems to their brick and unit.
- **Short Desc** The name of the unit or brick. This is the name displayed within the A-SMART application.
- **ADO Unit Indic** U, S, or N. Units are U, bricks are N. S is for subunits, and is not used by A-SMART.
- **Tree Nd Num** Tree node number. A unique id within the tree file which is used to define the hierarchy.
- **Prnt Node** Defines the parent of the given node, by it's Tree Nd Num.

Units within the tree structure file must be listed in hierarchical order; that is parent units must be listed before their children. This format is the format supplied directly from PMKeyS, so should need no adjusting apart from saving in comma separated values format, and ensuring numbers are formatted simply.

## 6.2.3.2 Personnel Entitlement Data

The personnel entitlement data is transferred using a collection of data dumps from PMKeyS. These data dumps are stored one per year or more, in separate files, and contain all personnel entitlements which are active during the time period.<sup>20</sup> Time periods of less than one year cannot be loaded, although this only affects the termination date. Since there is no termination date present in the data files, the termination date of personnel is determined implicitly from the data using the year range the data represents. The position number is used to correlate entitlement positions over time and when the position number is no longer active in a given year range, it is assumed to have terminated at the end of the previous one.

## 6.2.3.2.1 Control file Format

The control file links multiple entitlement data files together. It specifies which years the given entitlement data covers and where to locate it. It must be stored in 'CSV' format. An example is shown in Table 3.

<sup>&</sup>lt;sup>20</sup> Currently this data is provided upon request by the Deputy Director Data Management (DPlans-A, AHQ) as separate Excel spreadsheets. It is correct at the date the query is conducted.

	А	В	С	D
1	Туре	Year Start	Year End	File Name
2	REG	2008	2009	2008 REG
3				

#### Table 3: Example Personnel Control File

- **Type** REG or RES. Indicates the force type stored in the file. Only REG positions are used.
- **Year Start** The start of the date range for which this data is valid.
- **Year End** The end of the date range for which this data is valid.
- **File Name** The file name of the data file, relative to the location of the control file, and without an extension. Normally the control file will reside in the same directory as the data files, so simply type the name of the file here. When loading, the application will automatically append '.csv' to the file name.

### 6.2.3.2.2 Entitlement File Format

Each personnel entitlement file consists of single rows of data, one for each entitlement that is active during the year the file represents. The file must contain the header row and columns as shown in Table 4.

### Table 4: Example Personnel Entitlement File

А	В	С	D	Е	F
Group	Sub Group	Unit ID	Unit Desc	Sub Unit ID	Sub Unit Descr
106695	OFF VCDF	106695	OFF VCDF		

#### ... continued

G*	Н	I*	J*
DeptID	Dept Descr	Position	Posn Descr
106695	OFF VCDF	999999	VICE CHIEF OF DEFENCE FORCE

#### ... continued

K*	L	M*	Ν	0
Job Cd	Jobcode Descr	Service Type	Rank	Corps
99999	REG SENOFF VCDF LTGEN	REG	LTGEN	SENOFF

#### ... continued

P*	Q	R*	S*	Т*	U
Location	Descr	MLOC	OLOC	Eff Date	Status
RUSS	Russell ACT	Y	Y	2007-11-24	Approved

As with the previous section, only those columns marked with (\*) are used, and the first row listed shows the names of the columns in a spreadsheet application. Again, this format matches the PMKeyS data dump format. The meaning of each significant column is:

• **DeptID** - The unique identifier for the department. This is the master key which links personnel and systems to their brick and unit. Each DeptID in this file must have a corresponding entry in the structure format file, otherwise the unit in question will not be considered.

- **Position** The position number. This is globally unique and represents the position of the entitlement. This number, together with the DeptID and file year, is used to construct the migration of units over time.
- **Job Cd** The jobcode of the position, specifying the qualifications required for the position. A-SMART will create new unlinked jobcodes for any new codes it encounters. An additional manual process is required to map these codes to careers and streams if they are to be modelled. See Section 6.3.2, Jobcode Map for further details.
- **Service Type** REG or RES; regular or reserve position. Only these are considered, others are ignored.
- **Location** The location of the position. A-SMART only tracks location by brick, so it assumes all personnel in the same brick are at the same location. The location is not used by A-SMART directly, but merely to provide additional data filtering options. If it is not known or not required this should be 'Unknown'.
- **MLOC** The AE value; active entitlement value. If this is 'Y' then 1 will be added to the MLOC value of this entitlement.
- **OLOC** The UE value; unit entitlement. If this is 'Y' then 1 will be added to the OLOC value of this entitlement.
- **Eff Date** The initial date the position became active. For a new position for a given year, the date must not be earlier than the date associated with the personnel file, except for the first year of data. This can be represented in a number of formats; ISO date format, 'YYYY-MM-DD', a standard Australian shortened date format 'DD/MM/YY', or a longer version 'DD/MM/YYY'.

## 6.2.3.3 System Entitlement Data

The system entitlement data either comes from DES dumps or from DES load material from the SED review team. In either case, significant additional work is required to prepare the data into a format suitable for loading into A-SMART. All data files must be converted to the strict canonical format described in Section 6.2.3.3.2, Entitlement File Format, taking particular care with the department identifier column which is used to associate personnel entitlements with the corresponding system entitlements at the brick level. And finally a control file must be created to indicate which files are to be loaded.

## 6.2.3.3.1 Control File Format

The control file format is an index which lists all files to be loaded. An example is shown in Table 5. This file must be stored in CSV format, and all file names will have '.csv' appended to construct the actual file name used. Lines where the file name is empty are ignored, but it may be useful to include the full structure for reference.

	А	В	С
1	DeptID	Short Desc	File Name
2	0	DEFENCE	
3	1	Unit A	unit-a
4	200	Brick A	
5	201	Brick B	
6	2	Unit B	unit-b
7			

Table 5:	Major	Systems	Control	File
	5	<i>J</i>		

## 6.2.3.3.2 Entitlement File Format

The format for the major system entitlement data is slightly more complicated than the personnel entitlement data and has some strict requirements. The file format parallels the important parts of both the DES data dumps and the SED review team data but is not compatible and requires significant manual editing. The main difference is that a department identifier must be supplied which matches the given brick to the matching brick in the personnel entitlement data and tree structure files.

Clarifying the previous paragraph; this data format is <u>not</u> the same as the DES dumps or the SED review team data. It has similarities but is more suited to automated data loading for the A-SMART tool. See Section 6.8, Major Systems Data Load for a summary of the process that has been used to load data to date. A similar process will be required to convert any available data into the format described in this section.

A	В	С	D	Е	F	G
* comment *						
Unit	Unit A					
Department ID	1					
DEPTID	SIGC/EGC	DESC	OLOC	MLOC	FTE	LE
200	Brick A					
1	1005-0001	PISTOL 9MM AUTO	3	3	3	0
2	5855-0022	NIGHT VISION GOGGLE	3	3	3	0
201	Brick B					
1	1005-0001	PISTOL 9MM AUTO	4	3	3	0

 Table 6:
 System Entitlement Data Sample

Table 6 shows an example file. It consists of the following sections:

- **Header Section** The header section includes every row until, and including, the one beginning with DEPTID. In contains various fields and comments, most of which are ignored by the current importer. The row containing DEPTID must be exactly as shown. If it is not, then the file will not be read by the importer.
- **Brick sub header** When the DESC column is empty, then the DEPTID column contains the department ID for the following entitlements, and the SIGC/EGC column contains the name of the department. All other columns are ignored.
- **Brick Entitlements** When the DESC column is non-empty, then the row contains the description of the entitlement. For these rows the DEPTID column is ignored. Within the entitlement section there are two parts. First a header which defines the brick to which the entitlements belong, followed by a number of entitlement entries. The columns are:
  - **DEPTID** When the **DESC** column is empty, then this is the department identifier for the following entitlements. Otherwise it is a paragraph indicator or user reference and not used by the data loader.
  - SIGC/EGC When the DESC is empty, then this is the title of the brick the following entitlements belong to. This case is not used by the data loader, but is used as a reference when using the raw data files. When DESC is non-empty, then this is the unique SIGC which identifies this piece of equipment.

- DESC A description of the equipment. Normally this is not used, but if the given SIGC is new to the system, this description will be used to add the given SIGC to the system.
- **OLOC** The operational level of capability.
- **MLOC** The minimum level of capability.
- **FTE** The full time entitlement component of this item to the target brick.
- LE The loan entitlement component of this item to the target brick.

## 6.3 Model Data

## 6.3.1 Personnel Asset

The initial personnel population can be defined by the Personnel Asset Data. It specifies the current population from the live PMKeyS system at a particular date. Positions are specified using jobcodes which the importer then re-maps to the streams and ranks modelled by the system.<sup>21</sup>

## 6.3.1.1 Quick Start

The personnel assets can be loaded via the *Data Tools* tab in the *Tools* page.

- 1. Go to the Data Tools tab in the Tools page.
- 2. Select an asset file.
- 3. Click on Load Personnel Asset Levels. Loading the Personnel Asset Data may take a few minutes.
- 4. Review the loading log, similar to that shown in Figure 82.
- 5. Accept the data load by pressing Finish

## 6.3.1.2 Personnel Asset Data Format

The Personnel Asset Data is the result of a PMKeyS query but must be stored in a CSV format file.

 Table 7:
 Personnel Asset Data File

	А	В	С	D	E	F	G
1	Group	Sub Group	Unit ID	Unit Descr	Sub Unit ID	Sub Unit Descr	DeptID
2	100039	CBT FORCES	100840	1 CSR			146172
3	100039	CBT FORCES	100840	1 CSR			146172
4	100039	CBT FORCES	101093	1 RAR			141092
5	100039	CBT FORCES	101093	1 RAR			141093

<sup>&</sup>lt;sup>21</sup> Note that, rather than using current personnel levels for initial populations for model runs, the user has the option to input levels themselves, set as a percentage of entitlement levels.

#### .. continued

•	commuca	continued								
	Η	Ι	J	K*	L	М				
1	Short Desc	Position	Posn Descr	Job Cd	Jobcode Descr	Service Type				
2	A VEH TP	782713	RFN	33862	REG RIFLEMAN None PTE	REG				
3	A VEH TP	123135	RFN	POOL27	ARMY Pool - Regular Paid	REG				
4	3 SECT	641367	RFN	33862	REG RIFLEMAN None PTE	REG				
5	4 SECT	572134	RFN	33862	REG RIFLEMAN None PTE	REG				

#### ... continued

	Ν	0	Р	Q	R	S	Т	U
1	Rank Code	Rank	Family	Corps	Location	Descr	MLOC	OLOC
2	E02	PTE (E)	FM00504	RA INF	ROBI	Robertson Bks Palmerston NT	N	Y
3	E02	PTE (E)	FM00504	RA INF	ROBI	Robertson Bks Palmerston NT	N	Y
4	E02	PTE (E)	FM00504	RA INF	LAVA	Lavarack Bks Tvl QLD	Y	Y
5	E02	PTE (E)	FM00504	RA INF	LAVA	Lavarack Bks Tvl QLD	Y	Y

#### ... continued

	V	W	X*	Y	Z	AA	AB
1	Eff Date	Status	ID	Service Type	Family	Family	Function
2	01/01/09	Operations	VACANT				
3	01/01/09	Operations	VACANT				
4	01/01/09	Approved	6238762	REG	FM00504	RA INF	FN60414
5	01/01/09	Approved	2631830	REG	FM00504	RA INF	FN60414

### .. continued

•	· continued									
	AC	AD	AE	AF	AG*	AH*	AI			
1	Function	Sub Function	Sub Function	Skill Grade	Skill Grade	Rank	Rank			
2	1									
3	,									
4	RIFLEMAN	SF60420	RIFLEMAN1	S160420	343-1	E03	PTE(P)			
5	RIFLEMAN	SF60420	RIFLEMAN1	S160420	343-1	E03	PTE(P)			

Table 7 shows an example fragment of a Personnel Asset Data file. Most of the columns are not used but are present in the query provided from PMKeyS. The significant columns are:

- **Job Cd** The jobcode of this position, or a pool name. If it is a number it is considered a jobcode.
- **ID** The employee position ID, or 'VACANT' if the position is unfilled. 'VACANT' positions are ignored.
- **Skill Grade** The ECN code for the skill grade of this position. If the Job Cd column indicates a pool position, this ECN is used to estimate the distribution of personnel in the same skill grade. Only the first 3 digits are considered.
- Rank The ranks are 'O00'-'O10' or 'E00'-'E10'. Only the ranks 'O01'-'O10' and 'E02'-'E10' are considered. Trainees and cadets are not counted (O00 = Officer Cadet OCDT, E00 = Recruit and E01 = Private Trainee).

The total population is calculated by adding up two sub-populations. First a direct count of all instances of a particular jobcode. To this is added an estimate of the contribution a given Skill Grade gives to the same jobcode for all pool positions. This is calculated as a weighted distribution of all positions which have the same Skill Grade and which specify a jobcode. If a given Skill Grade is referenced from a pool position, but does not define a jobcode, then that asset is not counted.

Once the total population for each jobcode is accumulated, the weighting values from the Jobcode Map are used to translate these populations into absolute populations in a given Stream and Rank. These are then stored as a multiplier of the target value set for the given Stream and Rank across all units in the current ORBAT. Because of the multiplicative nature of this data relationship, loading the initial asset levels can take a long time as there are many pieces of data which must be calculated and stored. This complex process is a side effect of storing personnel entitlement data by jobcode even though it is being modelled by Stream and Rank. It is also as a consequence of being able to specify asset populations relative to the target value of each Stream and Rank for every individual Brick.

## 6.3.2 Jobcode Map

The Jobcode Map defines how jobcode numbers (from PMKeyS) are transformed into A-SMART streams<sup>22</sup> for the purposes of modelling. Each jobcode may be associated with a number of streams and using a fixed statistical weighting. A-SMART provides a full editor for manipulating the jobcode map within the scenario screen (see Figure 83).



*Figure 83: Jobcode Editing* 

## 6.3.2.1 Quick Start

The Jobcode Map can be exported and imported via the *JobCode / ECN* tab in the *Scenarios* page. The process of importing jobcodes will merge the new set with the existing set, so importing can also be used to just add new jobcodes.

<sup>&</sup>lt;sup>22</sup> A-SMART streams link closely to Employment Category Numbers and define the career profile for a given trade. See Section 6.9, Training Streams for further information.

## 6.3.2.1.1 Export Jobcode Map

- 1. Go to the *JobCode / ECN* tab on the Scenarios page (see Figure 83).
- 2. Click on Export to Excel
- 3. Enter a file name to save as.
- 4. Save the file.

### 6.3.2.1.2 Import Jobcode Map

Once you have prepared a previous export for import:

- 1. Go to the *JobCode / ECN* tab on the Scenarios page (see Figure 83).
- 2. Click on Import from Excel
- 3. Select the file name of the prepared data file.
- 4. Then load it.

## 6.3.2.2 Jobcode Map Format

The Jobcode Map data file provides a static mapping from the unique job identifier used in the personnel entitlement data to the career stream and rank used when modelling within A-SMART. It consists of rows which define a single jobcode to stream and rank mapping. Each jobcode may have any number of mappings, although the over-all weight should add up to 1.0, but this is not enforced.

<i>Table 8:</i>	Example Jobcodes File Format
-----------------	------------------------------

	А	В	С	D	E	F
1	JobID	JobTitle	Rank	Corps	Stream	weight
2	40895	REG INF OFFR NONE CAPT	CAPT	RAINF	Officer RAINF -	1
3	43659	REG ANY RAAC/RAA/RA INF LT	LT	RAA	Officer RAA -	0.5
4	43659	REG ANY RAAC/RAA/RA INF LT	LT	RAINF	Officer RAINF -	0.5
5	43967	REC INSTR	CPL	RAINF	Rifleman -	1

Table 8 shows a partial example of a jobcode file, the columns are:

- JobID The jobcode/job identifier taken from the personnel SED data.
- **Job Title** The title of this jobcode/job description. If the jobcode already exists in the system then this is only for human reference, but if it doesn't, this title is used to create the new jobcode.
- **Rank** The corresponding rank for this position. One of the modelled ranks should be used, as visible in the Training Streams editor. If a jobcode maps to multiple streams then only the rank of the first will be used for all of them.
- **Corps** For reference only; the corps of the position.
- **Stream** The career stream this mapping applies to or blank if this jobcode has no mapping. The stream name is taken from that shown on the *Training Streams* page; it must match exactly including cAsE and internal spaces.

• **Weight** - Represents the statistical likelihood of this jobcode being filled by personnel from this stream. All weights from a given jobcode should add up to 1.0.

Additional details to note:

- If the jobcode does not map to any modelled stream, then leave columns D to F blank.
- New jobcodes can be added to this file, and they will be added during import.
- An existing jobcode will have its mapping reset to match the contents of the file.
- If the system has an existing jobcode which is not present in the loader file, it will remain in the system.

Currently the file is stored in the Excel 2007 XML format. Unfortunately this format is an undocumented, unsupported format, but Excel or other spreadsheet software should be able to load and save in this format for the foreseeable future. A future task may be to convert the application to use the more efficient CSV format.

## 6.3.3 Model Rates

The model inputs for recruitment and non-deployed separation can be imported and exported from within A-SMART using a separate file. This is accessed from the *Model Rates* page (Figure 84), within the *Personnel Rates* tab.

A-SMART						
Current ORBAT	Default Orbat Scenario: Default Sce	enario				
Scenarios	Personnel Rates System Rates					
Force	Import Export					
Task groups Operations	Target Rate: Recruitment AR	A (persons per year)				v
Logistics Setup	Other Ranks Officers Rank #	Adjustment Varying Rat	tes			
Logistics Output	Corps Trade Stream	Recr. PTE CPL SGT	W02 W01	Rate Adjustment R/	ecri PTE CPL SGT WO2	WO1
Logistics Editor	RAINF Rifleman -	0 0.01 0.01 0.01	0.01 0	125% 0	0.01 0.01 0.01 0.01	0
Bup The Model	RAA Ground Based Air Defence -	0 0 0 0	0 0	100% 0	0 0 0 0	0
Personnel Results				-		
Training Results						
System Results						
Training Courses						
Training Streams						
10015						
10.00						
. Mul						
3 AB 2	1					
A-SMART	0					

*Figure 84: Model Rates Page* 

## 6.3.3.1 Quick Start

Most of the data on the *Model Rates* page (personnel only) can be exported and imported from the page itself. The file created can be used to transfer the settings to another user or another scenario, although it is up to the user to ensure the streams and ranks are present in the target scenario, otherwise the data will refuse to load.

### 6.3.3.1.1 Export Model Rates

- 1. Go to the *Personnel Rates* tab on the *Model Rates* page (see Figure 84).
- 2. Click on Export .....
- 3. Enter a file name to save as.
- 4. Save the file.

## 6.3.3.1.2 Import Model Rates

- 1. Go to the *Personnel Rates* tab on the *Model Rates* page (see Figure 84).
- 2. Click on Import .....
- 3. Select the file name of a previously exported or created file.
- 4. Then load the file.

## 6.3.3.2 Model Rates Format

The model rates file format consists of a number of worksheets within the same spreadsheet, each of which stores a given set of data. The worksheets must be in a specific order, and be labelled with specific titles. There are two basic sets of information stored in the export file format; annual recruitment targets and non-operational separation rates. For each set of data, officer and other ranks data is stored on separate work sheets so in total there are four work sheets required. The work sheets must be in the following order with the exact names as listed all within the same spreadsheet.

Worksheet	Purpose
RECARA	Recruitment rates for other ranks from the regular army.
SEPND	Non-deployed separation rates for other ranks.
OFF RECAR	Recruitment rates for officer ranks.
OFF SEPND	Non-deployed separation rates for officer ranks.

### Table 9: Example Model Rates, Other Ranks

А	В	С	D	E	F	G	Н
Corps	Stream	Recruit	PTE	CPL	SGT	WO2	WO1
RAA	Ground Based Air Defence -	0	0	0	0	0	0
RAINF	Rifleman -	0	0.01	0.01	0.01	0.01	0.01

... continued ...

Ι	J	K	L	М	N	0	Р	Q	R	S	Т
short	long	short	long	short	long	short	long	short	long	short	long
Recruit	Recruit	PTE	PTE	CPL	CPL	SGT	SGT	WO2	WO2	WO1	WO1

An example of the other ranks rates is shown in Table 9. The same structure is shared between both the recruitment and non-deployed separation rate work sheets. The columns have the following meaning:

- **Corps** For reference; the corps this stream belongs to.
- **Stream** The name of the stream. This must exactly match the streams loaded into the system, including cAsE and internal spaces. Refer to the *Model Rates* page to view the streams available.
- **Recruit ... WO1** The rate for each rank of the stream in question. Blank implies a rate of zero.
- **short Recruit** ... When the variable rate adjustment has been set for a given stream and rank, this will include an array of points which describes the variable rate curve (see Section 6.3.3.2.1). If they are not set, these will be blank. This covers the 'short' rate which represents annual cycles; these values are repeated for each 12-month cycle of the model run.
- **long Recruit** ... This is the same as the 'short Recruit' column, but is the long-term adjustment curve (see Section 6.3.3.2.1). This curve will span the life of the corresponding scenario.

Table 10 shows an example of the officer rates. This is basically the same as the data for the other ranks, but with the appropriate ranks listed. Note that 'CORPT' is a special A-SMART rank which represents the combined ranks of colonel or higher. It is not modelled.

А	В	С	D	Е	F	G
Corps	Stream	LT	CAPT	MAJ	LTCOL	CORPT
RAA	Officer RAA -	0	0	0	0	0
RAINF	Officer RAINF -	0	0	0	0	0

Table 10: Example Model Rates, Officer Ranks

... continued ...

Н	I	J	K	L	М	Ν	0	Р	R
Short LT	Long LT	Short CAPT	Long CAPT	Short MAJ	Long MAJ	Short LTCOL	Long LTCOL	Short CORPT	Long CORPT

### 6.3.3.2.1 Variable Rates

Variable rates are encoded using a list of coordinate tuples enclosed in square brackets '[]'. Values within the range of those specified are calculated using linear interpolation.

[(month, rate adjust), (month, rate adjust), ...]

The *month* value is relative to the start of the scenario, which is zero. The *rate adjust* value is a multiplier applied at that point in time. For short rates, the maximum value allowed for *month* is 12. For long rates its value should not exceed the maximum length of the scenario. The values must be specified in strictly-ascending *month* order. The value of *rate adjust* should be within the range (0.0, 3.0) for it to be accessible in the editor. This represents a scaling of 0 times up to 3 times of the base value.

## 6.4 Backup and Restore

In addition to the load, save, and transfer of specific segments of data, there is also the ability to perform wholesale backups of the entire set of data, or by entire ORBAT/scenario set.

## 6.4.1 Database Backup

All data for A-SMART is stored in a PostgreSQL database. It comes with administrations tools to allow reliable backup and restoration of whole databases (Figure 85). Using the pgadminIII tool backing up the whole database is very simple. Restoring databases is slightly more complex, but is still relatively straightforward. pgadminIII is installed during the A-SMART installation process and should in the 'start' menu. Note that because the entire database is processed as a whole, separate data sets cannot be merged together using this procedure. Other tools provide some of this functionality such as the ORBAT Backup process (see Section 6.4.2).

## 6.4.1.1 Database Backup

- 1. Ensure you have quit out of all running instances of the A-SMART application.
- 2. Run the pgadmin III application.
- 3. Connect to the asm database by double-clicking on it.
- 4. Right-click on the asm database to bring up the context menu and select Backup...
- 5. If required, select the filename in the save window (See Figure 85).
- 6. Click on OK to save the backup file.

📑 Backup Database asm	
Filename E:\asm-2	20081127-pre-branch.backup
⊙ COMPRESS ◯ TAR	PLAIN
<pre>     Plobs     With OIDs     Insert commands     Disable \$ quoting     Disable \$ </pre>	PLAIN options Only data Only schema No owner Create DB Drop DB Disable Trigger
☑ Verbose messages	
Options Messages	<u>O</u> K <u>C</u> ancel

Figure 85: Backing Up the asm Database

🔃 Restore Databas	e asm	
Filename	e:\asm-20081127-pre-branch.t	
	🗌 Only data	
	Only schema	
	Single object	
	No owner	
	🗌 Disable Trigger	
	✓ Verbose messages	
Options Contents	Messages	
Help	<u>V</u> iew <u>O</u> K	ancel

Figure 86: Restoring the asm Database

## 6.4.1.2 Database Restore

Database restore is slightly more involved since there will often be data present in the database which must be removed first to avoid conflicts.

- 1. Ensure you have quit out of all running instances of the A-SMART application.
- 2. Run the pgadmin III application.
- 3. Connect to the asm database by double-clicking on it.
- 4. If the asm database holds data which must be retained, first backup the existing database.
- 5. First Delete the old asm database. Right-click on the asm database (if it exists) to bring up the context menu and then select Delete/Drop. Confirm the delete to remove the database. This process is irreversible, so ensure the data is not required or has been backed up first.
- 6. Create a new asm database. Right-click on the Databases row in the pgAdminIII Object browser to bring up the context menu and then select **New Database**...
- 7. Enter asm as the database name and also ensure the encoding is set to UTF8. Clicking on OK will create the database. See Figure 87.
- 8. Now restore the database. Right-click on the asm database to bring up the context menu and then select Restore...
- 9. If required, select the file name in the load window (See Figure 86).
- 10. Click on **OK** to begin the restoration process.

11. There will usually be some errors during the import that can be safely ignored; it is beyond the scope of this document to cover them. In short, after the database has been restored, you must click on Cancel to finish rather than OK.

间 New Database	
Properties Variable	es Privileges SQL
Name	asm
OID	
Owner	
Encoding	UTF8
Template	
Tablespace	<pre></pre> <pre></pre> <pre></pre>
Schema restriction	
Comment	
Нетр	

Figure 87: Creating a New asm Database

### 6.4.1.3 File Format

The file format is the binary database backup ('dump') format used by PostgreSQL, and will not be documented here.

### 6.4.2 ORBAT Backup

The ORBAT Backup facility is intended to provide functionality for transferring modelling data-sets between different installations of the application. It can extract a whole ORBAT/force tree with all associated entitlement, operational, and modelling data into a single text file. This can be transferred to another user, and loaded in a single step. To simplify the processing involved there are some strict limitations on the transferability of the data and in general the data files exported using this functionality may only be transferred to users using the identical version of the application. If the ability to transfer between versions is required, then the process described in Section 6.4.1, Database Backup must be used.

## 6.4.2.1 ORBAT Backup

- 1. Go to the **Scenarios** page, and select the ORBAT or scenario set you wish to backup.
- 2. Click on Backup ...
- 3. Enter or select a file name.
- 4. Then save the file.
- 5. The application may become unresponsive during the backup process.

## 6.4.2.2 ORBAT Restore

- 1. Go to the **Scenarios** page.
- 2. Click on Restore ...
- 3. Select the file name of a previously created backup.
- 4. Then load the file. The application may become unresponsive during the restore process.
- 5. The loaded ORBAT will be available at the end of the list of ORBATs, and may be selected manually.

## 6.4.2.3 File Format

The file is stored in a simple CSV format. Essentially it consists of the items of interest of every relevant table in the database, concatenated together. As the database tables change from version to version of the application, this format should be considered to be undocumented and only used to backup and restore ORBAT data between identical versions of the application. To emphasise, this file format is <u>undocumented</u> and not designed for manual manipulation.

## 6.5 Tree Structure Sample

Table 11 contains an example tree structure definition file used in the following sections. The tree stored in this file represents the structural hierarchy shown in Figure 88.

Unit ID	Unit Descr	Sub Unit ID	Sub Unit Descr	DeptID	Short Desc	ADO Unit Indic	Tree Level	TreeNd Num	Prnt Node
0	DEFENCE			0	DEFENCE	U	1	1	0
1	Unit A			1	Unit A	U	2	2	1
1	Unit A			200	Brick A	U	3	3	2
1	Unit A			201	Brick B	U	3	4	2
2	Unit B			2	Unit B	U	2	5	1
2	Unit B			202	Brick C	Ν	3	6	5
2	Unit B			203	Brick D	N	3	7	5

Table 11: tree-structure.csv



Figure 88: Tree Structure Hierarchy

## 6.6 Personnel Entitlement Sample

In this section a complete data-set for an example data load of personnel entitlement data is presented. This uses the structure file from Section 6.5, Tree Structure Sample. The control file is in Table 12. Table 13 contains the entitlement data for positions active between 2008 and 2009. Table 14 contains the positions active between 2009 and 2010. And finally Table 15 contains the positions active from 2010 until 2016. This demonstrates that the same file can be used to span years, when the data does not change, or when data for intervening years is unavailable. Various fields which are ignored are filled with 'na' to help the tables fit on the printed output (for this report).
Things to note:

- When a position number is no longer present in the following year, the active lifespan of the position will be inferred to be until the end of the previous year.
- The same data can span a range of years.
- If a brick's entitlements are changed in a given year, and the DeptID remains the same, the loader will automatically create the correct time-span for the brick and also link all associated bricks together.
- If a brick is changed in a subsequent year, and given a new DeptID, it will be treated as an independent brick.

	А	В	С	D
1	Туре	Year Start	Year End	File Name
2	REG	2008	2009	2008 REG
3	REG	2009	2010	2009 REG
4	REG	2010	2016	2010-2016 REG

Table 12: personnel-map.csv

### DSTO-GD-0595

# Table 13: 2008 REG.csv

	Group	Sub Group	Unit ID	Unit Descr	Sub Unit ID	Sub Unit	DeptID	Dept Descr	Position	Posn Descr	Job Cd	Jobcode Descr	Service Type	Rank	Corps	Location	Descr	MLOC	OLOC	Eff Date	Status
1	L	Unit A	200	Brick A			200	Brick A	1	grunt a	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
]	L	Unit A	200	Brick A			200	Brick A	2	grunt b	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
]	L	Unit A	200	Brick A			200	Brick A	3	grunt c	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
]	L	Unit A	200	Brick A			200	Brick A	4	grunt d	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
1	L	Unit A	201	Brick B			201	Brick B	5	grunt e	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
1	L	Unit A	201	Brick B			201	Brick B	6	grunt e	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
1	L	Unit A	201	Brick B			201	Brick B	7	grunt f	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
]	L	Unit A	201	Brick B			201	Brick B	8	grunt g	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
2	2	Unit B	202	Brick C			202	Brick C	9	grunt h	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
2	2	Unit B	202	Brick C			202	Brick C	10	grunt i	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
2	2	Unit B	202	Brick C			202	Brick C	11	grunt j	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
2	2	Unit B	202	Brick C			202	Brick C	12	grunt k	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na

#### DSTO-GD-0595

# Table 14: 2009 REG.csv

					1							1	1	1	1		1		1
Group	Sub Group	Unit ID	Unit Descr	Sub Unit ID	Sub Unit DeptID	Dept Descr	Position	Posn Descr	Job Cd	Jobcode Descr	Service Type	Rank	Corps	Location	Descr	MLOC	OLOC	Eff Date	Status
1	Unit A	200	Brick A		200	Brick A	1	grunt a	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
1	Unit A	200	Brick A		200	Brick A	3	grunt c	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
1	Unit A	200	Brick A		200	Brick A	4	grunt d	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
1	Unit A	201	Brick B		201	Brick B	6	grunt e	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
1	Unit A	201	Brick B		201	Brick B	7	grunt f	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
1	Unit A	201	Brick B		201	Brick B	8	grunt g	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
2	Unit B	202	Brick C		202	Brick C	9	grunt h	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
2	Unit B	202	Brick C		202	Brick C	10	grunt i	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
2	Unit B	202	Brick C		202	Brick C	11	grunt j	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
2	Unit B	202	Brick C		202	Brick C	12	grunt k	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
2	Unit B	203	Brick D		203	Brick D	13	grunt l	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2009	na
2	Unit B	203	Brick D		203	Brick D	14	grunt m	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2009	na
2	Unit B	203	Brick D		203	Brick D	15	grunt n	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2009	na
2	Unit B	203	Brick D		203	Brick D	16	grunt o	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2009	na

### DSTO-GD-0595

# Table 15: 2010-2016 REG.csv

Group	Sub Group	Unit ID	Unit Descr	Sub Unit ID	Sub Unit Descr	DeptID	Dept Descr	Position	Posn Descr	Job Cd	Jobcode Descr	Service Type	Rank	Corps	Location	Descr	MLOC	OLOC	Eff Date	Status
1	Unit A	200	Brick A			200	Brick A	1	grunt a	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
1	Unit A	200	Brick A			200	Brick A	17	grunt b	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2010	na
1	Unit A	200	Brick A			200	Brick A	3	grunt c	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
1	Unit A	200	Brick A			200	Brick A	4	grunt d	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
1	Unit A	201	Brick B			201	Brick B	18	grunt e	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2009	na
1	Unit A	201	Brick B			201	Brick B	6	grunt e	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
1	Unit A	201	Brick B			201	Brick B	7	grunt f	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
1	Unit A	201	Brick B			201	Brick B	8	grunt g	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2007	na
2	Unit B	203	Brick D			203	Brick D	13	grunt l	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2010	na
2	Unit B	203	Brick D			203	Brick D	14	grunt m	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2010	na
2	Unit B	203	Brick D			203	Brick D	15	grunt n	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2010	na
2	Unit B	203	Brick D			203	Brick D	16	grunt o	33862	na	REG	PTE	RAINF	RUSS	na	Y	Y	1/01/2010	na

# 6.7 Major Systems Entitlement Sample

This section contains the data for a major systems entitlement sample. It uses the same structure file from Section 6.5, Tree Structure Sample and completes the entitlement data for major systems to match the personnel data from Figure 88. The control file is shown in Table 16; it references the two unit data files described in Table 17 and Table 18.

	А	В	С
1	DeptID	Short Desc	File Name
2	0	DEFENCE	
3	1	Unit A	unit-a
4	200	Brick A	
5	201	Brick B	
6	2	Unit B	unit-b
7	202	Brick C	
8	203	Brick D	

#### Table 16: system-map.csv

Tabl	e 17:	unit-a.csv

* comment *						
Unit	Unit A					
Department ID	1					
DEPTID	SIGC/EGC	DESC	OLOC	MLOC	FTE	LE
200	Brick A					
1	1005-0001	PISTOL 9MM AUTO	3	3	3	0
2	5855-0022	NIGHT VISION GOGGLE	3	3	3	0
3	2320-0003	TRUCK UTILITY LTWT MC4	1	1	1	0
201	Brick B					
1	1005-0001	PISTOL 9MM AUTO	4	3	3	0
2	5855-0022	NIGHT VISION GOGGLE	4	3	3	0
3	2320-0164	TRUCK CARRYALL LTWT MC3	2	2	2	0
	* comment * Unit Department ID DEPTID 200 1 2 3 201 1 2 3 3 201 1 2 3 3	* comment *   Unit Unit A   Department ID 1   DEPTID SIGC/EGC   200 Brick A   1 1005-0001   2 5855-0022   3 2320-0003   201 Brick B   1 1005-0001   2 5855-0022   3 2320-0164	* comment *Image: Comment *UnitUnit ADepartment ID1DEPTIDSIGC/EGCDESC200Brick A11005-0001PISTOL 9MM AUTO25855-002232320-0003TRUCK UTILITY LTWT MC4201Brick B11005-0001PISTOL 9MM AUTO25855-0022NIGHT VISION GOGGLE32320-01432320-0164201Same and the second secon	* comment *	* comment *   Image: Comment *	* comment *   Image: Comment *

Table 18: unit-b.csv

* comment *						
Unit	Unit B					
Department ID	2					
DEPTID	SIGC/EGC	DESC	OLOC	MLOC	FTE	LE
202	Brick C					
1	1005-0001	PISTOL 9MM AUTO	1	1	1	0
2	5855-0022	NIGHT VISION GOGGLE	1	1	1	0
3	2320-0003	TRUCK UTILITY LTWT MC4	1	1	1	0
203	Brick D					
1	1005-0001	PISTOL 9MM AUTO	2	2	2	0
2	5855-0022	NIGHT VISION GOGGLE	2	2	2	0
3	2320-0164	TRUCK CARRYALL LTWT MC3	1	1	1	0

# 6.8 Major Systems Data Load

This section lists some of the issues experienced when trying to load Major Systems Data and point to the need for some consistent machine-readable interchange format. It makes neither conclusions nor recommendations but merely relates the experiences of the A-SMART team performing this task.

The Major Systems Data has typically arrived in a set of spreadsheets with one file per unit.<sup>23</sup> Within each spreadsheet a list of bricks are defined followed by the system components identified by an SIGC and the entitlement quantities. On the face of it this electronic format would appear ideally suited to an automated loading process, however there were often inconsistencies and other problems which hampered this task:

- Different meanings for columns in different spreadsheets even within the same data set.
- Inconsistent titles for the same pieces of equipment and other typographical errors.
- Sometimes the use of multiple worksheets within the same spreadsheet file, only one of which was relevant.
- Inconsistent section markers separating summary information from the raw data.
- Different file naming conventions even within the same data set making it impossible to automatically link the file names to the units they relate to.
- Different unit and brick naming conventions compared to the personnel entitlement and ORBAT data.

Apart from these basic data inconsistencies there is also important data that is simply not available, such as a corresponding department identifier for individual bricks which matches that specified in personnel entitlement data. The inconsistency issues were addressed in different ways depending on the size of the problem, either through adjusting the loader code or by manual editing. The association of department identifiers to individual bricks was performed using a process of manual editing with software support to reduce the size of the problem.

# 6.8.1 Custom Loaders

Custom loaders were developed when there was a common inconsistency problem affecting many files. Work-a-rounds and special case code were added to the most recent loader to attempt to make sense of the inconsistencies. This was a practical approach when the expectation was that on-going updates would occur, having most of the same problems; it was hoped that by fixing the problem once then data inconsistencies would not need to be fixed everywhere they occurred. Unfortunately this generally did not happen and almost every time a new set of data was delivered it was different enough from the previous set that new work-a-rounds needed to be applied. Sometimes new 'fixes' broke old work-a-rounds. Changing requirements or on-going changes to application database tables and data structures often meant the entire loader itself needed to be rewritten.

<sup>&</sup>lt;sup>23</sup> The major systems equipment data was obtained from the Organisation cell within DPlans-A, AHQ.

# 6.8.2 Linking Department ID

The other major problem was linking the major systems data to the corresponding personnel entitlement data. Firstly, because the personnel and major systems tracking software are separate systems such linking information data is not available directly. Secondly, because the unit naming conventions differ they could not be used to correlate corresponding units automatically. Consequently, the process required significant manual labour. Being a manual process creating the department identifier linkage was slow, tedious and error prone. The following steps were repeated until the data-set loaded without data errors:

- 1. Produce or obtain a list of units and bricks in a given order.
- 2. Extract the list of bricks under a given unit and find the corresponding list of bricks from the Major Systems Data for that unit manually.
- 3. Attempt to match up both lists of bricks (personnel and major systems) in the order of the first. This must be done manually as the naming conventions often differ.
- 4. Any bricks that are not present in the Major Systems Data should have empty bricks inserted with the correct name.
- 5. Repeat steps 2–4 for all units.
- 6. Attempt to load the data and if errors are found address them by modifying the code or the source of the errors. The loading process matches up bricks from each list by their position in the list. Another control file is needed to match the file name to the unit identifier.

Additional manual checking is still required to validate the load. The end result was a series of incompatible custom-written data loaders filled with many messy work-a-rounds written in low quality code that was only used once to perform the load in question and then discarded.

# 6.9 Training Streams

This section describes the process that was used to load the training stream information into A-SMART. This was a one-off process using ad-hoc software that was not further developed and integrated into the main application as a user feature, as such it is only included for reference purposes. It may assist with any future efforts to develop this functionality. Note that there is a full editor built-in to A-SMART which allows for the creation and editing of streams and courses.

# 6.9.1 Data Capture

One of the more complex data-sets required for full career modelling is an electronic description of careers and course requirements. Since no such database existed at the time<sup>24</sup>, an effort was made within the A-SMART project to capture as much information as possible and create a baseline data-set. A contractor with Army experience gathered this information and stored it in a large number of spreadsheets. The information gathered during the collection stage fell roughly into the following categories:

<sup>&</sup>lt;sup>24</sup> The Training Management Pack Support System (TPMSS) is now operational. However, a cursory scan of the TMP data suggests that it is not in a format which would support automated processing easily (if at all).

- **Streams** The streams represent the career profiles of individual trades.
- **Training Steps** The training steps represent stages in career streams. To make the modelling and analysis of the streams practical, career profiles were converted to simple linear sequences of stages. Each stage can specify an on-the-job training period and/or a set of elective courses required for advancement. A fixed weighting is applied to determine relative course load when more than one elective is defined.
- **Courses** Each course describes a training school, capacity, and includes a number of training modules.
- Modules Each module describes the resources required to run each course.
- **Resources** Resources that are required to run a single course. These resources may include instructors, support staff (such as medical personnel), facilities, weapons, vehicles and other specialised equipment, ammunition and other consumable supplies. Notionally resources can be specified per-course, per-group, or per-student.
- **Instructors** Instructors and support staff were specified by ECN, Corps and Rank. Where a staff position could be filled by different instructors, multiple options were specified.
- **Staff** Were specified similarly to instructors.
- **Ammunition** Ammunition details were specified by ammunition type and rounds or boxes.
- POL Petrol, oil, lubricants. Consumable petrochemicals.
- Facilities Training facilities required.
- Vehicles Trucks, trailers, specialist equipment.
- **Communications** Radios, computer equipment.

# 6.9.1.1 Courses

Figure 89 is a partial screen shot of the data capture for a course. There is a header which describes some general information about the course; the number of students per course is the most important for A-SMART. Following the header a section describes the instructors required to run the course. These are specified by Corps, Rank, and ECN. The next section describes the support staff, which are specified in a similar manner. Then follows a number of other sections, each describing various resources required to run the course; only *Rations* and *Ammunition* were able to fit on this screen-shot, although more follow in the file.

	SUBJ 1 CPL,xls				[	
	В	С	D	E	F	
1	COURSES CONDUC	CTED BY LWC				
2	C					
3		ALL CORPS				
4	Tradestream					
5	EUN Saharal Daamamaikia faratka Causaa	SUBJ 1 CPL ALL CORPS				
6	School Responsible for the Course					
4	Location of School	LVVC Detachments				
8	Course Name	SUB 11 CDI				
10	Length of course (days)	42				
11	Target Rank	PTE (P)				
12	Course Prerequisites	Nil				
13	TIR Preferred	1				
14	No. of courses per 1 Y	72				
16	Student/Instructor Ratio 1 to	8				
17			Quantity	Unit of Issue	Total	
18	Instructors	Ranks and preferred ECN				
19	Senior Instructor	MAJ_RAINF RAAC RAA	1	Module	1	
20	3510	SGT ECN 123 386 426 074-2 076-0 084-		Module		
21	Instructors	5 099-4 229-2 256-4 315-2 357-3	1	Section	9	
22	Support Staff	Ranks and preferred ECN				
23	- Dedicated to the course		4	kda aluda	4	-
24	Auministrative Clerk Stores Clerk	CPL RAAOC ECN 074-1	1	Module	1	
26	Enemy Party or Drill Squad	RAINF PTE ECN 343-3	9	Module	9	
27	Medical Assistant	RAAMC CPL ECN 031	1	Module	1	
28	No. of Instructor man/days	966		Total	23	
29	Additional Support Staff - Only as Require	(Not Included in total mandays)				
30	VL Padra		1	Module	1	
32	Band		1	Module	1	
33	TDO		1	Module	1	
34						
35	Major Resources Required			· · ·	0000	
36	Rations	Fresh rations	31	per trainee	2232	
38		Hot box meal	0	Module	0	
39		Hot box meal	0	per staff	Ō	
40		Combat Rat 5 man	0	per trainee	0	
41		Combat Rat 5 man	0	per staff	0	
42		Combat Rat 1 man	11	per trainee	792	
43		IMCR	0	per stan	255	
45		IMCR	0	per staff	0	
46						
47	Ammunition (No. and type)	Cart .30 Cal 4B/1T Linked	0	per trainee	0	
48		Cart .30 Cal Blank Linked	0	per trainee	0	
49		Cart 50 Cal 4B/TT Linked	0	nertrainee	0	
51		Cart .50 Cal Blank Linked	0	per trainee	0	
52		Cart .50 Cal 4B/1T Linked	0	Course	0	
53		5.56mm Ball (TRG)	3824	Course	3824	
54		5.56mm Blank	21244	Course	21244	
56		5 56mm 4B/1T Linked	0 	ner trainee	1700 Ω	
57		7.62mm Ball	0	per trainee	0	
58		7.62mm Blank	0	per trainee	0	
59		7.62mm 4B/IT Linked	0	per trainee	0	
60		7.62mm Ball Linked	0	per trainee	0	
61		7.oznim blank brass Linked Grenade Smoke Blue	12	Course	12	
63		Grenade Smoke Green	12	Course	12	
64		Grenade Smoke Red	12	Course	12	
65		Grenade Smoke Yellow	12	Course	12	
66		Grenade Smoke 76mm	0	Course	0	
67		Sig illuminating Ked Star	U 8		8	
69		Sim grenade	20	Course	20	
70		Sim PROJ GND Burst	60	Course	60	
71		Grenade Hand F1	0	per trainee	0	
72		Grenade Hand Prac	0	per trainee	0	
73		Inp hare	16	Course per traisco	16	
75		21mm Sub cal	0	per trainee	0	
76		25mm APDS-T	0	pertrainee	0	
77		25mm APDS-T	0	Course	0	
78		25mm MP-T SD Mk II	0	per trainee	0	
79		25mm TP-1 40mm HEDR	0	per trainee	0	
00			0	por trainee	0	<b>- v</b>
H	♦ N \SUBJ 2 CPL ALL CORPS /	<		1111		<b>&gt;</b> ].:

Figure 89: Example Course Spreadsheet

# 6.9.1.2 Streams

Figure 90 is a screenshot of a specific captured stream. This shows the career profile of a Rifleman, which has been converted into a set of linear steps. Starting from a new recruit, this table specifies the training courses and rank progression of the trade. The ECN numbers are used to tie the various streams together, and the streams to the personnel entitlement data, using the Jobcode Map.

<b>1</b>	Career Profile and Co	urses Rifleman.>	ds				•	
	С	D	E	F	G	Н	1	
1		RAINF RIF	LEMA	N CAREER PROP	ILE AND COUR	SES		
2								
3	Min Time in Rank	Career Year		CAREER FL	OW CHART		File Reference	
4								
5	~	16.5		W01 E	CN 350			
6								
⊬				RSIVICO	JURSE		ALL CORPS COMMON/RSM COURSE	$\vdash$
G G	6	10.5		W02 E	CN 387			$\vdash$
10	0	10.5		102 1				
11				SUBJ 2 SPVR I	NF OPS - COY		RAINE/4 SIO-C	
12								
13				SUBJ	1 WO		ALL CORPS COMMON/SUBJ 1 WO	
14								
15	4	6.5		SGT E	CN 386			
16								
17				SUBJ 2 SPVR	INF OPS - PL		RAINF/3_SIO-P	
18				0.00				
19				SOBJ	ISGI		ALL CORPS COMMON/SUBJ 1 SGT	— E
20	2	25		CPL EC	N 242 2			-
22	J	3.5			1 343-3			
23				SUBJ 2 SPVR IN	VF OPS - SECT		RAINE/2 SIO-S	
24								
25	1	2.5		LCPL EC	N 343-2			
26								
27				SUBJ	1 CPL		ALL CORPS COMMON/SUBJ 1 CPL	
28								
29	1	1.5		PTE(P) E	CN 343-1			
30				0.17.42.1				$\vdash$
31				UJI 12 IV	IUNTHS			$\vdash$
32	1	0.5		ECN (	3/3-1			
34		0.0		LON	545-1			
35				IET EC	N 343		RAINF/1_IET COURSE	
36								
37				ECN	510			
38								
39				ECN	500		ALL CORPS COMMON/Recruit Training Course	
40	0	0		Recruit	Training			
41								~
H 4	🕞 🕨 🔪 Career - Infa	ntry Rifleman /				<		>

Figure 90: Career Profile - Rifleman

# 6.9.2 Data Processing

Since A-SMART cannot use spreadsheet information directly, this data needed to be converted to a usable format. A set of database tables were created to represent as much of the captured information as possible, even if that information was not yet required for modelling purposes. See Figure 91 for a diagram of the database schema used to represent this information. Unfortunately, as the data was entered by hand from many different sources, there were many inconsistencies and errors within it. Some spreadsheets were simply formatted using different columns and rows, or the trainers were not defined in a consistent manner. As the data was updated, new inconsistencies and problems arose. Some of these were handled by special-case algorithms in the data loader, and many others were addressed by hand. Where there was insufficient information to resolve issues (such as unknown major system types), the data was simply lost. As such, the spreadsheet format from the previous

section was not viewed as a practical means to update and maintain the stream information and training requirements.



Figure 91: Structure of Training Data

# 6.9.2.1 Streams

An additional manual step was required to convert the stream profile spreadsheets into a suitable format. They were all placed in a single spreadsheet, and extraneous lines and information removed. On the job training was removed as a specific step, and instead the minimum time in rank was used to translate this information into a format suitable for A-SMART. An example of the result is shown in Table 19. It has a 1:1 correspondence with the Rifleman career profile shown in Figure 90.

Corps	Stream	Substream	Stage	Percent	Min TIR	Rank In	Rank Out	ECN in	ECN Out	Course Step	Notes
RAINF	Rifleman				0	Recruit		500	510	Recruit Training Course.xls	
					0	Recruit		510		1_IET COURSE.xls	
					6	Recruit	PTE	510	343-1		
					12	PTE		343-1	343-2		
					12	PTE		343-2		SUBJ 1 CPL.xls	
					12	PTE	LCPL	343-2			
					0	LCPL		343-2		2_SIO-S COURSE.xls	
					12	LCPL	CPL	343-2	343-3		
					0	CPL		343-3		SUBJ 1 SGT.xls	
					0	CPL		343-3		3_SIO-P COURSE.xls	
					36	CPL	SGT	343-3	386		
					0	SGT		386		SUBJ 1 WO.xls	
					0	SGT		386		4_SIO-C COURSE.xls	
			1		48	SGT	WO2	386	387		
					0	WO2		387		RSM COURSE.xls	
			1		60	WO2	WO1	387	350		

Table 19: Career Stream – Rifleman

## 6.9.3 Recommendation

In hindsight, too much detail was captured in the original spreadsheets, and too much time was wasted trying to import data elements which were never used. The enormity of the data capture task meant that application agility was compromised — one of the key features of the A-SMART application. In future, detailed data requirements analysis should be conducted prior to commencing data capture and import to ensure only relevant data is generated.

# 7. Summary

This document provides a guide to users of A-SMART and an introduction on how to interpret results produced by the tool; it provides clear directions for the setup of force structures (organisation and entitlement data), scenarios, and in loading input data (including details of format requirements). It is not meant to replace training on the system but to compliment it. A second phase of A-SMART development would include adding in modules for Facilities and Collective Training, as well as optimisation and costing capabilities (to compliment the current personnel, major systems and supplies/strategic lift modules).

DEFENCE SCIENCE	AND	TECHNOLO	GY ORG	ANISA							
DOCU	MEN	T CONTROL	DATA			1.	PRIVACY MARK	ING/C	CAVEAT (OF DOCUMENT)		
2. TITLE Army Sustainability Modellin	g Ana	lysis and		3 1 0	3. SECURIT THAT ARE I CLASSIFICA	Y CLA LIMIT ATION	ASSIFICATION (F ED RELEASE US I)	OR UN E (L) N	JCLASSIFIED REPORTS NEXT TO DOCUMENT		
Reporting Tool Phase 1: User	Manua	al and Results Int	erpretation	ı	D	ocum	ent	(T	D		
Guide					Title (U)						
			Abstract (U)								
4. AUTHORS			5. CORPORATE AUTHOR								
Matthew K. Richmond, John S Andrew Roth, Shane Miller ar	n, Michael Zucch Do	i,	I	DSTO Defence Science and Technology Organisation PO Box 1500							
				E	Edinburgh South Australia 5111 Australia						
6a. DSTO NUMBER		6b. AR NUMBER		6	C. TYPE OF	REPO	ORT	7. D	OCUMENT DATE		
DSTO-GD-0595		AR-014-654			General Do	cume	ent	Nov	ember 2009		
8. FILE NUMBER 9.	TASK	NUMBER	10. TASK	K SPONSO	OR	11. N	O. OF PAGES		12. NO. OF REFERENCES		
2009/1049831 AI		/ 103	DGDF-F	A, AIIQ		111			0		
13. URL on the World Wide Web				14. REI	LEASE AUT	HORI	TY				
http://www.dsto.defence.gov GD-0595.pdf	v.au∕c	corporate/reports	s/DSTO-	Chief,	Land Ope	eratio	ns Division				
15. SECONDARY RELEASE STAT	TEMEN	NT OF THIS DOCU	MENT	•							
			Approved	l for put	blic releas	e					
OVERSEAS ENQUIRIES OUTSIDE ST	ATED I	LIMITATIONS SHOU	LD BE REFERI	RED THRO	DUGH DOCU	MENT	EXCHANGE, PO B	OX 1500	), EDINBURGH, SA 5111		
16. DELIBERATE ANNOUNCEM	IENT										
No Limitations											
17. CITATION IN OTHER DOCU 18. DSTO RESEARCH LIBRARY	JMEN' THESA	TS AURUS	Yes								
User Menuel Softwar	Mod	alling Tool Susta	inability								
Oser Manual, Software Modelling 1001, Sustainability											
19. ABSTRACT This report is designed to assist users of the Army Sustainability Modelling Analysis and Reporting Tool (A-SMART) in setting up input parameters and scenarios, running the models and interpreting the model outputs. A-SMART is a software tool under contracted development, sponsored by Army Headquarters and managed by the DSTO, which has reached the end of its first phase of development. A- SMART is a strategic-level force structure liability analysis tool, designed to forecast the dynamics of personnel and equipment populations over time for a particular scenario; it is intended that it will support analysis of the sustainability of planned Army force structures against a range of possible scenarios, as well as the force options testing process. A-SMART Phase 1 has been limited to the development of personnel, major equipment and supplies /strategic lift modules; it is envisioned that a second phase of development would extend the capability to the											

Page classification: UNCLASSIFIED

other Fundamental Inputs to Capability including collective training and facilities, and also include a costing module.